PURCHASE AGREEMENT

Dated as of October 27, 2018 among

CORDILLERA COMMUNICATIONS, LLC

and

SCRIPPS MEDIA, INC.

with respect to the acquisition of

certain subsidiaries of CORDILLERA COMMUNICATIONS, LLC

TABLE OF CONTENTS

		Page
ARTICLE 1 D	efinitions; Interpretation	1
1.1	Definitions	1
1.2	Definitions Cross-Reference Table	
1.3	Interpretation	
	urchase of Equity Interests	
	· ·	
2.1 2.2	Purchase of Equity Interests	
2.2	Working Capital Adjustment	
2.3 2.4	Closing	
2.5	Governmental Consents	
2.6	Withholding	
ARTICLE 3 S	eller Representations and Warranties	
3.1	Organization and Capitalization	16
3.2	Authorization	
3.3	No Conflicts	
3.4	FCC and Programming Distribution Matters	
3.5	Taxes	
3.6	Tangible Personal Property	
3.7	Real Property	
3.8	Contracts	
3.9	Environmental	24
3.10	Intangible Property	24
3.11	Employees; Labor Matters; Employee Benefit Plans	25
3.12	Insurance	27
3.13	Compliance with Law; Permits	28
3.14	Litigation; Orders	
3.15	Financial Statements	
3.16	No Undisclosed Liabilities	
3.17	Absence of Changes	
3.18	No Brokers	
3.19	Related Party Transactions	
3.20	All Assets	
3.21	Bank Accounts	
ARTICLE 4 B	uyer Representations and Warranties	
4.1	Organization	
4.2	Authorization	
4.3	No Conflicts	
4.4	Litigation	
4.5	Qualification	
4.6	Projections and Other Information	
4.7	Sufficient Funds	
4.8	No Brokers	
4.9	Securities Laws	
	ertain Covenants	
5.1	Seller's Covenants	
5.2	Inspection Rights	
5.3	Control and Maintenance of Qualification	
5.4	Debt Financing Cooperation	35

TABLE OF CONTENTS (continued)

			Page
ARTI	CLE 6 J	oint Covenants	36
	6.1	Confidentiality	36
	6.2	Announcements	
	6.3	Consents	
	6.4	Employees; Employee Plans	
	6.5	Access to and Retention of Records	
	6.6	Cooperation; Financial Statements	
	6.7	Interim Reports	
	6.8	Fulfillment of Conditions.	
	6.9	Title Commitments; Surveys	
	6.10	No Negotiation	
	6.11	Officers and Directors Indemnification; Insurance Tail Coverage	
	6.12	Termination of Related Party Transactions	
	6.13	Multi-Station Contracts	
	6.14	Non-Solicitation Contracts	
	6.15	Mail and Other Communications	_
	6.16	Joint Privilege	
	6.17	Environmental Assessments; Phase I Investigations.	
	6.18	Satellite Matter	
	6.19	Tower Matter	
	6.20	Certain Additional Matters	
ADTI			
AKII		eller Closing Conditions	
	7.1	Representations and Covenants	
	7.2	Proceedings	
	7.3	FCC Authorization	
	7.4	Hart Scott Rodino	
	7.5	Deliveries	
ARTI	CLE 8 B	uyer Closing Conditions	
	8.1	Representations and Covenants	46
	8.2	Proceedings	
	8.3	FCC Authorization	46
	8.4	Hart Scott Rodino	46
	8.5	Deliveries	46
	8.6	Consents	46
	8.7	No MAE	
ARTI	CLE 9 C	losing Deliveries	
	9.1	Seller Documents	
	9.2	Buyer Documents	47
ARTI	CLE 10	Survival; Indemnification	48
	10.1	Survival	48
	10.2	Indemnification by Seller	
	10.3	Indemnification by Buyer	
	10.4	Procedure for Indemnification	
	10.5	No Circular Recovery	
ARTI		Tax Matters	
	11.1 11.2	Tax IndemnityStraddle Period Allocation	
		Tax Returns	
	11.3	TAX NEWITS	

TABLE OF CONTENTS (continued)

		Pa	ige
	11.4	Refunds	
		Cooperation	
	11.6	Transfer Taxes	
	11.7	Purchase Price Allocations	_
	11.8	Tax Claims	
		Tax Treatment of Indemnity Payments	
		Wage Reporting	
	11.11	Tax Sharing, Indemnity and Allocation Agreements	55
ARTICL	E 12 Te	rmination And Remedies	55
	12.1	Termination	55
	. —	Cure Period	
	12.3	Effect of Termination	
	12.4	Specific Performance	
ARTICL	E 13 Mi	scellaneous	57
	13.1	Expenses	57
	13.2	Further Assurances	
	13.3	Assignment	57
	13.4	Notices	
	13.5	Amendments; Waivers	58
	13.6	Severability	58
	13.7	No Beneficiaries	58
	13.8	Governing Law; Consent to Jurisdiction; Waiver of Jury Trial	59
	13.9	Neutral Construction	
	13.10	Non-Recourse Against Debt Financing Sources; Waiver of Certain Claims	
	13.11	Counterparts; Delivery by Facsimile/Email	
	13.12	Entire Agreement	60

PURCHASE AGREEMENT

THIS PURCHASE AGREEMENT (this "Agreement") is made as of October 27, 2018, by and between CORDILLERA COMMUNICATIONS, LLC, a South Carolina limited liability company ("Seller"), and SCRIPPS MEDIA, INC., a Delaware corporation ("Buyer").

RECITALS

WHEREAS, Seller, together with the Acquired Companies, owns and operates, directly or indirectly, the television broadcast stations (each, a "*Station*" and collectively, the "*Stations*") set forth on **EXHIBIT A**, pursuant to certain licenses, permits and other authorizations issued by the FCC;

WHEREAS, pursuant to the terms and subject to the conditions set forth in this Agreement, Seller desires to sell and transfer to Buyer, and Buyer desires to purchase from Seller, all of the issued and outstanding equity interests in the Acquired Companies (collectively, the "**Equity Interests**"); and

WHEREAS, in connection with the foregoing, this Agreement contemplates the filing with the FCC of multiple applications to obtain the necessary FCC consents with respect to the transactions described in this Agreement, which applications will include one or more applications seeking FCC consent to the transfer of control of the Acquired Companies to Buyer.

AGREEMENT

Now, Therefore, in consideration of the mutual covenants and agreements set forth herein, the parties, intending to be legally bound, hereby agree as follows:

ARTICLE 1

DEFINITIONS; INTERPRETATION

- **1.1 Definitions**. As used in this Agreement, the following terms shall have the following meanings:
 - "Acquired Companies" shall mean the Subsidiaries set forth on EXHIBIT B.
- "Action" shall mean any legal, administrative or regulatory claim, suit, action, complaint, charge, arbitration, examination, audit, inquiry, investigation, or other proceeding by or before any Governmental Entity, other than random selection for EEO audit or complaints and other matters pending at the FCC as to which no notice has been given to an Acquired Company.
- "Affiliate" shall mean, with respect to a specified Person, any Person or member of a group of Persons acting together that, directly or indirectly, through one or more intermediaries, controls, or is controlled by or is under common control with, the specified Person. As used in this definition, the term "control" (including the terms "controlling," "controlled by" and "under common control with") shall mean the possession, direct or indirect, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, by contract or otherwise.
- "Ancillary Documents" shall mean, collectively, the Seller Ancillary Agreements and the Buyer Ancillary Agreements.
 - "Balance Sheet Date" shall mean August 31, 2018.
- "Business" shall mean the business and operation of the Stations (taken as a collective group and not on an individual basis).

- "Business Day" shall mean any day that is not a Saturday, Sunday or other day on which banks are required or authorized by law to be closed (or are actually closed) in the City of New York.
- "Closing Indebtedness Certificate" shall mean a certificate executed by the Chief Financial Officer of Seller and each of the Acquired Companies certifying on behalf of Seller and each of the Acquired Companies and not in any individual capacity a good faith estimate (on an itemized basis) of all outstanding Indebtedness of the Acquired Companies as of immediately prior to Closing and the Persons to whom such outstanding Indebtedness is owed and an aggregate total of such outstanding Indebtedness.
- "Closing Transaction Costs Certificate" shall mean a certificate executed by the Chief Financial Officer of Seller and each of the Acquired Companies certifying on behalf of Seller and each of the Acquired Companies and not in any individual capacity a good faith estimate of the amount of Company Transaction Costs remaining unpaid as of immediately prior to Closing (including an itemized list of each such unpaid Company Transaction Cost with a description of the nature of such cost and the Person to whom such expense is owed).
- "CMS MSA" means the Master Services Agreement between Ackmann and Dickenson, LLC and Cordillera Digital, dated effective as of June 11, 2018, as amended by Addendum #1, dated effective as of June 11, 2018, between such parties.
 - "Code" shall mean the Internal Revenue Code of 1986, as amended.
- "Communications Laws" shall mean, collectively, the Communications Act of 1934, as amended, and the rules, regulations and written policies promulgated by the FCC thereunder, in each case as in effect from time to time.
- "Company Transaction Costs" shall mean all fees, costs and expenses of any brokers, financial advisors, investment bankers, attorneys, or other advisors engaged by or on behalf of Seller or the Acquired Companies incurred on or prior to the Closing Date in connection with the preparation, negotiation and execution of this Agreement and the Ancillary Documents, and the performance and consummation of the transactions contemplated hereby and thereby plus (i) all retention bonus, change in control and similar payments to directors, officers, employees or service providers of Seller or the Acquired Companies payable in connection with the Closing of the transactions contemplated by this Agreement and (ii) all prepayment penalties with respect to any Indebtedness of any of the Acquired Companies that is paid off in conjunction with the Closing of the transactions contemplated by this Agreement.
- "Contracts" shall mean contracts, agreements, leases, non-governmental licenses, sales and purchase orders and other agreements (including Revenue Leases, Real Property Leases and employment agreements), written or oral (in each case, including any amendments, supplements, renewals or modifications thereto).
- "Copyrights" shall mean all copyrights and works of authorship, whether or not copyrightable, and all copyright applications and registrations therefor (and renewals thereof).
- "Current Assets" shall mean cash and cash equivalents, accounts receivable, inventory and prepaid expenses of the Acquired Companies, as determined in accordance with GAAP, but shall exclude (i) all Tax assets, (ii) any intercompany balances or receivables from any of any Acquired Company's Affiliates, managers, employees, officers or members or any of their respective Affiliates and (iii) the portion of any prepaid expense of which Buyer will not receive the benefit following the Closing.
- "Current Liabilities" shall mean accounts payable, accrued and current Tax liabilities and accrued expenses of the Acquired Companies, as determined in accordance with GAAP, but shall

exclude (i) the current portions of Indebtedness and Company Transaction Costs described in Section 2.2(a) and (b) and (ii) intercompany balances.

- "**Debt Financing**" shall mean any debt financing provided to Buyer or its Affiliates in connection with the transactions contemplated hereby.
- "Debt Financing Provisions" shall mean, collectively, Section 5.4 (Debt Financing Cooperation), Section 13.3 (Assignment), Section 13.5 (Amendment), Section 13.7 (No Beneficiaries), Section 13.8 (Governing Law; Consent to Jurisdiction; Waiver of Jury Trial) and Section 13.10 (Non-Recourse Against Debt Financing Sources; Waiver of Certain Claims).
- "Debt Financing Sources" shall mean the Persons that have committed to provide or arrange or otherwise entered into agreements in connection with the Debt Financing, including the parties to any debt commitment letters, engagement letters, joinder agreements, indentures or credit agreements entered pursuant thereto or relating thereto, together with their respective Affiliates, and their respective Affiliates' officers, directors, employees, agents and representatives and their respective successors and assigns.
 - "DOJ" shall mean the United States Department of Justice.
- "Enforceability Exceptions" shall mean bankruptcy, moratorium, insolvency, reorganization or other similar laws affecting or limiting the enforcement of creditors' rights generally and general principles of equity (regardless of whether enforceability is considered in a proceeding at law or in equity).
- "Environmental Law" shall mean any Law whether local, state, or federal relating to (a) Releases or threatened Releases of Hazardous Materials into the environment; (b) the management, use, treatment, generation, processing, storage, disposal, remediation, handling, discharging or shipment of Hazardous Material; (c) the regulation of storage tanks; or (d) otherwise relating to pollution (or the cleanup thereof) or protection of natural resources, endangered or threatened species, human health, occupational safety or the environment (including ambient air, soil, surface water or groundwater, or subsurface strata).
 - "EPI" shall mean Evening Post Industries, Inc., an Affiliate of Seller.
 - "ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended.
- "*ERISA Affiliate*" shall mean any Person, employer, trade or business, whether or not incorporated, that, together with any of the Acquired Companies or any of their respective Affiliates, would be deemed a "single employer" within the meaning of Section 414 of the Code or Section 4001(b)(i) of ERISA.
 - "Escrow Agent" shall mean U.S. Bank National Association.
- "FCC" shall mean the Federal Communications Commission, including any official bureau or division thereof acting on delegated authority, or any successor agency thereto.
- "FCC Consent" shall mean the consent of the FCC (including any such action duly taken by the FCC's staff pursuant to delegated authority) to the transactions contemplated by this Agreement and described in the FCC Applications, each without any material adverse conditions other than those of general applicability to the television broadcasting industry or those imposed by reason of an action of Buyer or Seller in violation of this Agreement.
- "FCC License" shall mean any FCC license, permit or other authorization, including any renewals or modifications thereof or any transferable pending application therefor, issued by the FCC and granted

or assigned to any Acquired Company and including, but only during any period that it is necessary for the operation of a Station, any temporary waiver or special temporary authorization.

"Flow-Through Tax Returns" shall mean the following Tax Returns filed by Seller for any Pre-Closing Tax Period: (i) IRS Form 1065 and (ii) any state or local Income Tax Return that allocates the income or other items of Seller's Subsidiaries under state or local income Tax Law that conforms to Section 702 of the Code.

"FTC" shall mean the Federal Trade Commission.

"GAAP" shall mean United States generally accepted accounting principles as in effect as of the relevant date, consistently applied.

"Governmental Consents" shall collectively mean the FCC Consent and HSR Clearance.

"Governmental Entity" shall mean and include any court or tribunal or administrative, governmental or regulatory body, agency, commission, board, legislature, instrumentality, division, department, public body or other authority of any nation or government or any political subdivision thereof, whether foreign or domestic and whether national, supranational, federal, state or local, or any self-regulated organization or other non-governmental regulatory authority or quasi-governmental authority (to the extent that the rules, regulations or orders of such organization or authority have the force of Law).

"Governmental Order" shall mean any order, writ, judgment, injunction, decree, stipulation, determination or award entered by or with any Governmental Entity.

"Hazardous Material" shall mean (i) hazardous or toxic wastes, chemicals, substances, constituents, pollutants or related material, whether solids, liquids, or gases, defined or regulated under § 101(14) of CERCLA; the Resource Conservation and Recovery Act, 42 U.S.C. §§ 6901 et seq.; the Toxic Substances Control Act, 15 U.S.C. §§ 2601 et seq.; the Safe Drinking Water Act, 42 U.S.C. §§ 300(f) et seq.; the Clean Air Act, as amended, 42 U.S.C. §§ 7401 et seq.; the Federal Water Pollution Control Act, 33 U.S.C. §§ 1251 et seq.; the Emergency Planning and Community Right-to-Know Act of 1986, 42 U.S.C. §§ 11001 et seq.; the Occupational Safety and Health Act of 1970, 29 U.S.C. §§ 651 et seq. or any similar applicable federal, state or local Environmental Laws and (ii) petroleum or petroleum-derived products, radon, radioactive materials or wastes, asbestos in any form, lead or lead-containing materials, urea formaldehyde foam insulation, and polychlorinated biphenyls.

"HSR Act" shall mean the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended.

"Income Tax" shall mean any federal, state, local or foreign tax measured by or imposed on net income.

"Income Tax Return" shall mean any Tax Return relating to Income Taxes.

"Indebtedness" shall mean, with regard to any Person, any liability or obligation, whether or not contingent, (i) in respect of borrowed money or evidenced by bonds, monies, debentures, or similar instruments or upon which interest payments are normally made, (ii) for the payment of any deferred purchase price of any property, assets or services, but excluding trade payables and Program Rights Obligations, (iii) under or pursuant to any leases that have been capitalized historically or are required to be capitalized in accordance with GAAP, (iv) in respect of guaranties, direct or indirect, in any manner, of all or any part of any Indebtedness of any Person, (v) all obligations under acceptance, standby letters of credit or similar facilities, (vi) all matured obligations to purchase, redeem, retire, defease or otherwise make any payment (including any mandatory dividend or distribution) in respect of any membership interests, shares of capital stock or other ownership or profit interest or any warrants, rights or options to acquire such membership interests, shares or such other ownership or profit interest, (vii) all obligations

under any interest rate, currency swap or other hedging agreement or arrangement, (viii) all accrued interest, prepayment premiums or penalties related to any obligations referred to in (i) - (vii) and (ix) all obligations referred to in (i) - (viii) of a third party secured by any Lien on property or assets. For the avoidance of doubt, Indebtedness shall not include any Taxes.

"Intellectual Property" shall mean all intellectual property rights in or arising from any of the following: call letters, Trademarks, patents and patent applications, inventions, Trade Secrets, know-how, Internet domain names, web addresses, URLs, websites (and all content and data thereon or relating thereto), Copyrights, all other intellectual property and proprietary rights (including computer programs, operating systems, applications, firmware, and other code, including all source code, object code, application programming interfaces, data files, databases, protocols, specifications, and other documentation thereof), and all goodwill, if any, associated with any of the foregoing.

"IRS" shall mean the United States Internal Revenue Service.

"knowledge" shall mean (a) with respect to Seller, the knowledge of any of Terrance P. Hurley, the President of Seller, Sean Franklin, the Vice President, Engineering and Technology, of Seller, Steve Chase, the Vice President, Director of Sales, of Seller, Clinton Ebeling, the Director of Business Operations of Seller, Jay Small, the President of Seller's Affiliate, Cordillera Interactive LLC, Joe Waring, the CFO of Seller's Affiliate, EPI, or the general manager of any Station, in each case after reasonable inquiry and (b) with respect to Buyer, the knowledge of William Appleton, the Executive Vice President and General Counsel of Buyer, or Robin Davis, the Vice President/Strategy and Corporate Development of Buyer, in each case after reasonable inquiry.

"Law" shall mean any United States (federal, state, local) or foreign law, constitution, treaty statute, ordinance, regulation, rule, code, order, judgment, injunction, writ or decree.

"Lien" shall mean, with respect to any property or asset, any mortgage, lien, pledge, charge, easement, right of way, restrictive covenant, encroachment, security interest, equitable interest, right of first refusal, or encumbrance or restriction of any kind whatsoever, including any restriction on use, voting, transfer, receipt of income or exercise of any other attribute of ownership, whether voluntarily incurred or arising by operation of Law or otherwise, in respect of such property or asset.

"Losses" shall mean and include all losses, costs, damages, Taxes, liabilities, deficiencies, Actions, judgments, interest, awards, penalties, fines, and expenses of whatever kind, including reasonable attorneys' fees and expenses and the reasonable costs of enforcing any right to indemnification hereunder and the cost of pursuing any insurance providers.

"Market" shall mean the geographic area delineated and determined by Section 76.55(e) of the Communications Laws, or such other rule or decision of the FCC as may be promulgated from time to time for purposes of its must-carry rules to determine local television markets for commercial broadcast television stations, and as may be amended by applicable market modification decisions of the FCC, for the Stations.

"Marketing Period" shall mean the 15 consecutive Business Day period after the date of this Agreement during which each of the following is true: (a) Buyer shall have received all of the Required Information; (b) the FCC Consent shall have been granted and shall be in full force and effect; (c) any applicable waiting period under the HSR Act relating to transactions contemplated hereby shall have expired or been terminated; and (d) Seller shall be in compliance in all material respects with Section 5.4; provided that such period shall not include any date from November 21, 2018 through and including November 23, 2018 and if such period has not ended on or before December 20, 2018, it shall not commence before January 2, 2019. If Seller in good faith reasonably believes that it has delivered the Required Information, it may deliver to Buyer written notice to that effect (stating when it believes it completed any such delivery), in which case the receipt of the Required Information shall be deemed to have occurred and the 15 consecutive Business Day period described above shall be deemed to have commenced on the second Business Day following the date of receipt of such notice, unless Buyer in

good faith reasonably believes that Seller has not completed delivery of the Required Information and, within two Business Days after receipt of such notice from Seller, Buyer delivers a written notice to Seller to that effect (stating with specificity which information is required to complete the delivery of the Required Information).

"Material Adverse Effect" shall mean any event, state of facts, circumstance, development, change, effect or occurrence (an "Effect") that, individually or in the aggregate with any other Effect, has had or would reasonably be expected to have a materially adverse effect on (a) the business, properties, assets, financial condition or results of operations of the Business taken as whole, or (b) the ability of Seller to perform its obligations on a timely basis under this Agreement, excluding in all respects any Effects resulting from (i) conditions in the economy of the United States generally, including changes in the United States or foreign credit, debt, capital or financial markets in general (including changes in interest or exchange rates), or general conditions in the economy of any town, city or region or country in which the Stations conduct business, (ii) general changes or developments in the broadcast television industry, (iii) the public announcement of this Agreement and the transactions contemplated hereby, the consummation of the transactions contemplated hereby, the compliance with the terms of this Agreement or the taking of any action required by this Agreement or consented to by Buyer in writing, (iv) global, national or regional political conditions, including hostilities, military actions, political instability, acts of terrorism or war or any escalation or material worsening of any such hostilities, military actions, political instability, acts of terrorism or war existing or underway as of the date hereof, (v) earthquakes, hurricanes, tornadoes, or other natural disasters, (vi) any failure, in and of itself, by Seller, any Acquired Company or any Station or group of Stations to meet any internal or published projections, forecasts or revenue or earnings predictions for any period ending on or after the date of this Agreement (provided, however, that the underlying causes of such failure (subject to other provisions of this definition) shall not be excluded). (vii) any breach by Buyer of its obligations under this Agreement or (viii) changes in Law or GAAP or the interpretation thereof; provided however, that any Effect referred to in clauses (i), (ii) or (iv) immediately above shall be taken into account in determining whether a Material Adverse Effect has occurred or would reasonably be expected to occur to the extent that such Effect has a disproportionate effect on Seller, any Acquired Company or any Station compared to other participants in the broadcast television industry.

"Multiemployer Plan" shall mean a multiemployer plan, within the meaning of Sections 3(37) and 4001(a)(3) of ERISA, to which any of the Acquired Companies contributes or is required to contribute to, or under which any of the Acquired Companies has or may have any liability or obligation under, on behalf of current or former employees of any of the Acquired Companies.

"Net Working Capital Amount" shall mean (a) the amount of all Current Assets minus (b) the amount of all Current Liabilities of the Acquired Companies on a combined basis, in each case calculated in a manner consistent with GAAP.

"Ordinary Commercial Agreement" shall mean any contract, such as a loan agreement or a lease, entered into in the ordinary course of business consistent with past practice the primary purpose of which is unrelated to Taxes.

"Organizational Documents" shall mean, with respect to any Person (other than an individual), the articles or certificate of incorporation, bylaws, regulations, articles or certificate of limited partnership, partnership agreement, articles or certificate of formation or organization, limited liability company agreement, operating agreement and any other organizational or governing documents of such Person and any amendment to any of the foregoing.

"Permitted Liens" shall mean, as to any property or asset of any Station, collectively, (a) liens for Taxes, assessments and other charges of Governmental Entities not yet due and delinquent or that are being contested in good faith by appropriate proceedings and, in each case, for which appropriate reserves have been established in accordance with GAAP, (b) applicable zoning Laws and ordinances and similar Laws that are not violated in any material respect by any existing improvement, provided that such Laws and ordinances do not, individually or collectively, interfere with the use of the Real Property

as currently used in the Business, (c) any right reserved to any Governmental Entity to regulate the affected property that is stated in any Permit, recorded document or applicable Law, (d) in the case of any leased asset, (i) the rights of any lessor under the applicable lease agreement or any Lien granted by any lessor or any Lien that the applicable lease is subject to (to the extent any Lien that the applicable lease is subject to is contained within the lease itself), (ii) any Lien for Taxes, assessments and other charges of Governmental Entities that are not yet due and payable or that are being contested in good faith by appropriate proceedings and for which appropriate reserves have been created in accordance with GAAP and (iii) the rights of the grantor of any easement or any Lien granted by such grantor on such easement property, so long as none of the foregoing such matters affecting a leased asset, individually or collectively, materially interferes with the operation of the Business in a manner consistent with the current use by any Acquired Company or Station, (e) statutory, inchoate liens of carriers, warehousemen, mechanics, material men and other like Liens imposed by law arising or incurred in the ordinary course of business consistent with past practice for amounts that are not yet due and payable or that are being contested in good faith by appropriate proceedings and for which appropriate reserves have been created in accordance with GAAP and that are not resulting from any breach, violation or default of any Contract by Seller or any of its Affiliates (including the Acquired Companies), (f) restrictive covenants, easements, rights-of-way, encroachments, restrictions and any state of facts that an accurate survey or physical inspection would show, provided none of such matters, individually or collectively, render title unmarketable, interfere with the present use of the applicable Real Property in the Business or materially and adversely impact the commercial value of the Real Property, (g) for periods prior to the Closing Date, Liens that will be released prior to or as of the Closing Date, including all mortgages and security interests securing indebtedness of Seller or the Acquired Companies, (h) non-exclusive licenses of Intellectual Property granted in the ordinary course of business consistent with past practice that, individually or in the aggregate, do not, and would not reasonably be expected to, materially interfere with the use of the Intellectual Property by any Acquired Company or Station, and (i) Liens designated as Permitted Liens on Schedule 1.1 under the heading "Permitted Liens", if any.

- "*Person*" shall mean any natural person or any corporation, limited liability company, partnership, joint venture, trust or other legal entity, including a Governmental Entity.
- "Pre-Closing Tax Period" shall mean any taxable year or period that ends on or before the Closing Date and, with respect to any Straddle Period, the portion of such Straddle Period ending on and including the Closing Date.
- "*Program Rights*" shall mean all rights of any of the Stations to broadcast television programs or shows as part of the Station's programming, including all rights of any of the Stations under all film and program barter agreements, sports rights agreements, news rights or service agreements, affiliation agreements and syndication agreements.
- "*Program Rights Obligations*" shall mean all obligations in respect of the purchase, use, license or acquisition of programs, programming materials, films and similar assets that relate to the utilization of the Program Rights.
- "Reference Time" shall mean 11:59 p.m. Eastern Time on the date prior to the Closing Date; provided that the Reference Time for purposes of calculating Tax liabilities included in the Net Working Capital Amount shall be 11:59 p.m., local time for each location of the Seller or any of the Acquired Companies, on the Closing Date.
- "Release" shall mean any release, spill, emission, leaking, dumping, injection, pouring, deposit, disposal, discharge, dispersal, leaching or migration into the environment (including ambient air, surface water, groundwater, land surface or subsurface strata) or within any building, structure, facility or fixture.
- "Required Information" shall mean (i) the Financial Statements and (ii) the interim reports described in Section 6.7.

- "Restructuring" shall mean the merger of EPI with and into Merger EPIG Sub, LLC, a South Carolina limited liability company and wholly-owned subsidiary of EPI Group, LLC with EPI as the surviving company.
- "Revenue Leases" shall mean those leases, subleases, licenses or other occupancy agreements (including any and all assignments, amendments and other modifications of such leases, subleases, licenses and other occupancy agreements), pertaining to the use or occupancy of the Real Property under which an Acquired Company is a landlord, licensor, sublandlord or sublicensor.
 - "Straddle Period" shall mean a Tax period that includes (but does not end on) the Closing Date.
- "Sharing Agreement Station" shall mean the full service television broadcast station owned and controlled by a third party and not owned or controlled by any Acquired Company to which an Acquired Company provides services pursuant to a Sharing Agreement as set forth in Exhibit A.
- "Sharing Agreement Station License" shall mean the main station license issued by the FCC for the Sharing Agreement Station.
- "Station Licenses" shall mean the main station license issued by the FCC for each of the Stations set forth in Exhibit A.
- "Subsidiaries" shall mean and include those direct or indirect subsidiaries of Seller that are involved in the ownership or operation of the Stations or any assets relating to the ownership or operation of the Stations.
- "Tax" or "Taxes" shall mean all taxes, charges, fees, levies, imposts or other similar assessments or liabilities in the nature of a tax (including income, receipts, ad valorem, value added, excise, real or personal property, sales, occupation, service, stamp, transfer, land transfer, registration, goods and services, natural resources, severance, premium, windfall or excess profits, environmental, customs, duties, import and export, use, licensing, escheat, withholding, employment, social security, social services, education, unemployment, disability, employer health, health insurance and other Governmental Entity pension plan and workers compensation premiums or contributions, payroll, share, capital, capital gain, surplus, alternative, minimum, add-on minimum, estimated, franchise, or any other taxes, charges, surtaxes, fees, levies or other similar assessments or liabilities in the nature of a tax of any kind whatsoever and denominated by any name whatsoever), whether computed on a separate, consolidated, unitary or combined basis or in any other manner, and includes any interest, fines, penalties, assessments, deficiencies or additions thereto and any installments required in respect thereof.
- "*Tax Returns*" includes all returns, reports, declarations, elections, notices, filings, forms, statements and other documents (whether in tangible or electronic form (including magnetic media)) and including any amendments, schedules, attachments, supplements, appendices and exhibits thereto and any claims for refund, declarations of estimated Tax and information returns, made, prepared, filed or required by a Taxing Authority to be made, prepared or filed by Law in respect of Taxes.
- "*Taxing Authority*" shall mean any Governmental Entity responsible for the administration or imposition of any Tax.
- "Tower Leases" shall mean any agreement to which an Acquired Company is a party pertaining to the use and/or installation of radio masts and/or towers used or useful by any of the Stations for telecommunications and broadcasting in connection with the operation of the Business, where the Acquired Company holds an interest as tenant or subtenant. For the avoidance of doubt, all such Tower Leases shall constitute "Real Property Leases" for purposes of this Agreement.

"*Trade Secrets*" shall mean all trade secrets, discoveries, improvements, technology, business and technical information, databases, data compilations and collections, tools, methods, processes, techniques, and other confidential or proprietary information of any Person.

"*Trademarks*" shall mean all trade names, trademarks, service marks, trade dress, jingles, slogans, logos, brands, other source or business identifiers, trademark and service mark registrations and trademark and service mark applications owned by any Person, and the goodwill appurtenant thereto.

"Tradeout Agreement" shall mean any Contract, other than film and program barter agreements, pursuant to which an Acquired Company has agreed to sell or trade commercial air time or commercial production services of a Station in consideration for any property or service in lieu of or in addition to cash.

"Treasury Regulations" shall mean the Treasury Regulations promulgated under the Code.

1.2 Definitions Cross-Reference Table. The following terms defined in this Agreement in the sections set forth below shall have the respective meanings therein defined:

ACA Section 3.11(e) Acquired Companies Recitals Active Employees Section 6.4(a) Agreement Preamble Assessments Section 6.17(b) Audited Financial Statements Section 6.6(c) Buyer Preamble Buyer Ancillary Agreements Section 6.4(a) Buyer Benefit Plans Section 6.4(e) Buyer Indemnified Party and Buyer Indemnified Parties Section 10.2(a) Claimant Section 10.4(a) Closing Section 2.4(a) Closing Date Section 2.4(a) CMS Assignment and Assumption Section 9.1(k) COBRA Section 3.10(c) Continuing Employees Section 3.10(c) Continuing Employees Section 13.8(a) Cure Period Section 10.2(b) Disagreement Notice Section 10.2(b) Disagreement Notice Section 2.5(d) Divestiture Section 2.5(d) Divestiture Notice Section 3.11(c) Employment Agreement Section 6.4(a) Employment Agreement Section 6.4(a)		
Active Employees Agreement Preamble Assessments Section 6.17(b) Audited Financial Statements Buyer Buyer Ancillary Agreements Buyer Benefit Plans Section 6.4(a) Buyer Indemnified Party and Buyer Indemnified Parties Closing Closing Closing Closing Cobran Section 2.4(a) COBRA Company Systems Continuing Employees Covered Matters Cure Period Disagreement Notice Disagreement Notice Divestiture Divestiture Notice Directions Employment Agreement Employment Commencement Date Employment Colosing Balance Sheet Estimated Net Working Capital Amount EWS EEVION 2.3(a) Section 6.4(a) Section 10.2(b) Disagreement Notice Section 6.4(a) Section 6.4(a) Section 1.1(b) Section 1.1(c) Section 1.1(c) Section 1.1(c) Section 1.2(c) Section 2.3(e) Section 2.5(d) Section 6.4(a) Section 6	ACA	Section 3.11(e)
Agreement Assessments Section 6.17(b) Audited Financial Statements Suyer Preamble Buyer Ancillary Agreements Section 4.1 Buyer Benefit Plans Section 6.4(a) Buyer's 401(k) Plan Section 6.4(e) Buyer Indemnified Party and Buyer Indemnified Parties Claimant Closing Section 10.2(a) Closing Section 2.4(a) Closing Date CMS Assignment and Assumption COBRA Section 3.11(h) Company Systems Section 3.10(c) Continuing Employees Section 10.2(a) Section 10.2(a) Covered Matters Section 3.10(c) Covered Matters Section 3.10(c) Disagreement Notice Disagreement Notice Section 2.5(d) Divestiture Section 2.5(d) Divestiture Section 3.11(c) Employment Agreement Employment Agreement Environmental Work Environmental Work Environmental Work Section 6.4(a) Estimated Closing Balance Sheet Estimated Closing Balance Sheet EWS Section 6.6(c) Exchange Act Section 6.6(c)		
Assessments Audited Financial Statements Section 6.6(c) Buyer Preamble Buyer Ancillary Agreements Section 4.1 Buyer Benefit Plans Section 6.4(a) Buyer Indemnified Party and Buyer Indemnified Parties Section 6.4(e) Buyer Indemnified Party and Buyer Indemnified Parties Section 10.4(a) Claimant Section 2.4(a) Closing Section 2.4(a) Closing Section 2.4(a) CMS Assignment and Assumption COBRA Section 3.11(h) Company Systems Section 3.10(c) Continuing Employees Section 3.8(a) Cure Period Section 10.2(b) Disagreement Notice Divestiture Divestiture Notice Section 2.5(d) Divestiture Notice Section 3.11(c) Employment Agreement Environmental Work Environmental Work Environmental Work Section 6.4(a) Estimated Net Working Capital Amount Section 2.3(a) Estimated Net Working Capital Amount Section 2.3(a) Estimated Net Working Capital Amount Section 2.3(a) EEWS Section 2.3(a) Section 2.3(a) Section 2.3(a) EEWS Section 6.6(c)		
Audited Financial Statements Buyer Buyer Ancillary Agreements Buyer Benefit Plans Section 6.4(a) Buyer Senefit Plans Section 6.4(b) Buyer Indemnified Party and Buyer Indemnified Parties Claimant Closing Section 10.2(a) Closing Section 10.4(a) Closing Section 2.4(a) Closing Date Company Systems Section 3.11(h) Company Systems Section 3.11(c) Covered Matters Section 10.2(b) Disagreement Notice Divestiture Section 2.3(e) Divestiture Section 2.5(d) Diraft Allocation Section 3.11(c) Employment Agreement Employment Agreement Section 6.4(a) Employment Agreement Section 6.4(a) Employment Agreement Section 6.4(a) Employment Commencement Date Employment Agreement Section 6.4(a) Employment Commencement Date Employment Agreement Section 6.4(a) Employment Commencement Date Section 6.4(a) Employment Agreement Section 6.4(a) Employment Agreement Section 6.4(a) Employment Commencement Date Section 6.4(a) Employment Agreement Section		
Buyer Ancillary Agreements Buyer Benefit Plans Section 4.1 Buyer Senefit Plans Section 6.4(e) Buyer's 401(k) Plan Buyer Indemnified Party and Buyer Indemnified Parties Claimant Closing Closing Closing Section 10.4(a) Closing Closing Section 2.4(a) Closing Date Section 2.4(a) COBRA COBRA Section 3.11(h) Company Systems Section 3.10(c) Continuing Employees Section 13.8(a) Cure Period Section 12.2 Deductible Amount Disagreement Notice Divestiture Section 2.5(d) Divestiture Notice Section 11.7 Employee Plan Section 5.11(c) Employment Agreement Employment Agreement Employment Commencement Date Employment Commencement Date Estimated Closing Balance Sheet Estimated Net Working Capital Amount Section 6.6(c) Exchange Act Section 6.6(c)		3.7
Buyer Ancillary Agreements Buyer Benefit Plans Section 6.4(a) Buyer's 401(k) Plan Section 6.4(e) Buyer Indemnified Party and Buyer Indemnified Parties Claimant Closing Section 10.4(a) Closing Section 2.4(a) Closing Date CMS Assignment and Assumption COBRA Section 3.11(h) Company Systems Section 3.11(h) Company Systems Section 6.4(a) Covered Matters Section 13.8(a) Cure Period Section 10.2(b) Disagreement Notice Divestiture Section 2.5(d) Divestiture Section 2.5(d) Divestiture Notice Section 1.7 Employee Plan Section 3.11(c) Employment Agreement Employment Agreement Employment Commencement Date Employment Commencement Date Employment Commencement Date Section 6.4(a) Environmental Work Section 5.3(a) Estimated Closing Balance Sheet Estimated Net Working Capital Amount Section 6.6(c) Exchange Act Section 6.6(c)	Audited Financial Statements	
Buyer Benefit Plans Buyer's 401(k) Plan Buyer Indemnified Party and Buyer Indemnified Parties Claimant Claimant Closing Closing Section 2.4(a) Closing Date CMS Assignment and Assumption Company Systems Section 3.11(h) Company Systems Section 10.2(b) Covered Matters Cure Period Deductible Amount Disagreement Notice Divestiture Divestiture Notice Divestiture Notice Section 2.5(d) Direstiture Notice Section 2.5(d) Direstiture Notice Section 3.11(c) Employment Agreement Section 11.7 Employment Commencement Date Environmental Work Environmental Mork Estimated Closing Balance Sheet Estimated Net Working Capital Amount Section 2.3(a) Estimated Net Working Capital Amount Section 2.3(a) Estimated Net Working Capital Amount Section 6.6(c) Exchange Act Section 6.6(c)	Buyer	Preamble
Buyer's 401(k) Plan Buyer Indemnified Party and Buyer Indemnified Parties Section 10.2(a) Claimant Closing Section 2.4(a) Closing Date Section 2.4(a) COBRA Section 3.11(h) Company Systems Section 3.10(c) Continuing Employees Section 13.8(a) Cure Period Deductible Amount Deductible Amount Disagreement Notice Divestiture Section 2.5(d) Divestiture Notice Section 3.10(c) Draft Allocation Employment Agreement Employment Agreement Section 10.2(b) Employment Commencement Date Environmental Work Environmental Section 6.4(a) Environmental Section 6.11(b) Equity Interests Estimated Closing Balance Sheet Exchange Act Section 6.6(c) Section 6.6(c) Section 6.6(c) Section 6.6(c) Section 6.6(c)		
Buyer Indemnified Party and Buyer Indemnified Parties Claimant Closing Section 10.4(a) Closing Section 2.4(a) Closing Date CMS Assignment and Assumption COBRA Company Systems Section 3.11(h) Company Systems Section 3.10(c) Continuing Employees Section 13.8(a) Cure Period Section 12.2 Deductible Amount Disagreement Notice Section 2.3(e) Divestiture Section 2.5(d) Divestiture Section 2.5(d) Divestiture Notice Section 1.7 Employee Plan Section 3.11(c) Employment Agreement Section 6.4(a) Environmental Work Section 6.17(c) EPI Group Section 1.7(c) Estimated Closing Balance Sheet Estimated Net Working Capital Amount Section 2.3(a) Section 2.3(a) EWS Section 6.6(c) Section 6.6(c) Section 6.6(c)	Buyer Benefit Plans	Section 6.4(a)
Claimant Section 10.4(a) Closing Section 2.4(a) Closing Date Section 2.4(a) CMS Assignment and Assumption Section 9.1(k) COBRA Section 3.11(h) Company Systems Section 3.10(c) Continuing Employees Section 6.4(a) Covered Matters Section 13.8(a) Cure Period Section 12.2 Deductible Amount Section 10.2(b) Disagreement Notice Section 2.3(e) Divestiture Section 2.5(d) Divestiture Notice Section 3.11(c) Employment Agreement Section 6.4(a) Employment Agreement Section 6.4(a) Employment Commencement Date Section 6.4(a) Environmental Work Section 6.17(c) EPI Group Section 6.11(b)		Section 6.4(e)
Closing Date Section 2.4(a) CMS Assignment and Assumption Section 9.1(k) COBRA Section 3.11(h) Company Systems Section 3.10(c) Continuing Employees Section 6.4(a) Covered Matters Section 13.8(a) Cure Period Section 12.2 Deductible Amount Section 10.2(b) Disagreement Notice Section 2.3(e) Divestiture Section 2.5(d) Divestiture Notice Section 2.5(d) Draft Allocation Section 3.11(c) Employee Plan Section 3.11(c) Employment Agreement Section 6.4(a) Employment Commencement Date Section 6.4(a) Environmental Work Section 6.17(c) EPI Group Section 6.11(b) Equity Interests Recitals Estimated Closing Balance Sheet Section 2.3(a) Estimated Net Working Capital Amount Section 6.6(c) Exchange Act Section 6.6(c)	Buyer Indemnified Party and Buyer Indemnified Parties	Section 10.2(a)
Closing Date CMS Assignment and Assumption COBRA Section 9.1(k) COBRA Section 3.11(h) Company Systems Section 3.10(c) Continuing Employees Section 6.4(a) Covered Matters Section 13.8(a) Cure Period Section 12.2 Deductible Amount Section 10.2(b) Disagreement Notice Section 2.3(e) Divestiture Section 2.5(d) Divestiture Notice Section 2.5(d) Divestiture Notice Section 11.7 Employee Plan Section 3.11(c) Employment Agreement Section 6.4(a) Employment Commencement Date Environmental Work Section 6.17(c) EPI Group Section 6.11(b) Equity Interests Estimated Closing Balance Sheet Section 2.3(a) Estimated Net Working Capital Amount Section 6.6(c) Exchange Act Section 6.6(c)		Section 10.4(a)
CMS Assignment and Assumption COBRA Section 9.1(k) COBRA Section 3.11(h) Company Systems Section 3.10(c) Continuing Employees Section 6.4(a) Covered Matters Section 13.8(a) Cure Period Section 12.2 Deductible Amount Disagreement Notice Section 2.3(e) Divestiture Section 2.5(d) Divestiture Notice Section 2.5(d) Direstiture Notice Section 11.7 Employee Plan Section 3.11(c) Employment Agreement Section 6.4(a) Employment Commencement Date Environmental Work Section 6.17(c) EPI Group Section 6.11(b) Equity Interests Estimated Closing Balance Sheet Section 2.3(a) Estimated Net Working Capital Amount Section 6.6(c) Exchange Act Section 6.6(c)	Closing	Section 2.4(a)
COBRA Company Systems Section 3.10(c) Continuing Employees Section 6.4(a) Covered Matters Section 13.8(a) Cure Period Deductible Amount Disagreement Notice Divestiture Divestiture Divestiture Notice Diraft Allocation Employee Plan Employment Agreement Section 2.5(d) Employment Commencement Date Environmental Work EPI Group Equity Interests Estimated Closing Balance Sheet Exchange Act Section 3.11(h) Section 3.10(c) Section 13.8(a) Section 12.2 Section 10.2(b) Section 2.3(e) Section 2.3(e) Section 2.5(d) Divestiture Notice Section 2.5(d) Section 11.7 Employee Plan Section 3.11(c) Employment Agreement Section 6.4(a) Environmental Work Section 6.4(a) Environmental Work Section 6.17(c) EPI Group Section 6.11(b) Equity Interests Estimated Closing Balance Sheet Section 2.3(a) Exchange Act Section 6.6(c)	Closing Date	Section 2.4(a)
Company Systems Continuing Employees Section 3.10(c) Continuing Employees Section 6.4(a) Covered Matters Section 13.8(a) Cure Period Section 12.2 Deductible Amount Section 10.2(b) Disagreement Notice Section 2.3(e) Divestiture Section 2.5(d) Divestiture Notice Section 2.5(d) Draft Allocation Section 11.7 Employee Plan Section 3.11(c) Employment Agreement Section 6.4(a) Employment Commencement Date Section 6.4(a) Environmental Work Section 6.17(c) EPI Group Section 6.11(b) Equity Interests Recitals Estimated Closing Balance Sheet Section 2.3(a) Estimated Net Working Capital Amount Section 6.6(c) Exchange Act Section 6.6(c)	CMS Assignment and Assumption	Section 9.1(k)
Continuing Employees Covered Matters Section 13.8(a) Cure Period Section 12.2 Deductible Amount Disagreement Notice Section 2.3(e) Divestiture Section 2.5(d) Divestiture Notice Section 2.5(d) Diraft Allocation Section 11.7 Employee Plan Section 3.11(c) Employment Agreement Section 6.4(a) Employment Commencement Date Section 6.4(a) Environmental Work Section 6.17(c) EPI Group Section 6.17(c) EQUITY Interests Estimated Closing Balance Sheet Sestion 2.3(a) Estimated Net Working Capital Amount Section 6.6(c) Exchange Act Section 6.6(c)	COBRA	Section 3.11(h)
Covered Matters Cure Period Section 12.2 Deductible Amount Disagreement Notice Divestiture Divestiture Notice Section 2.5(d) Direstiture Notice Section 2.5(d) Direstiture Notice Section 2.5(d) Draft Allocation Section 11.7 Employee Plan Section 3.11(c) Employment Agreement Section 6.4(a) Employment Commencement Date Section 6.17(c) EPI Group Section 6.11(b) Equity Interests Recitals Estimated Closing Balance Sheet Section 2.3(a) Estimated Net Working Capital Amount Section 6.6(c) Exchange Act Section 6.6(c)	Company Systems	Section 3.10(c)
Cure Period Section 12.2 Deductible Amount Section 10.2(b) Disagreement Notice Section 2.3(e) Divestiture Section 2.5(d) Divestiture Notice Section 2.5(d) Draft Allocation Section 11.7 Employee Plan Section 3.11(c) Employment Agreement Section 6.4(a) Employment Commencement Date Section 6.4(a) Environmental Work Section 6.17(c) EPI Group Section 6.11(b) Equity Interests Recitals Estimated Closing Balance Sheet Section 2.3(a) Estimated Net Working Capital Amount Section 6.6(c) Exchange Act Section 6.6(c)	Continuing Employees	Section 6.4(a)
Deductible Amount Disagreement Notice Divestiture Section 2.3(e) Divestiture Notice Section 2.5(d) Divestiture Notice Section 2.5(d) Draft Allocation Section 11.7 Employee Plan Section 3.11(c) Employment Agreement Section 6.4(a) Employment Commencement Date Section 6.4(a) Environmental Work Section 6.17(c) EPI Group Section 6.11(b) Equity Interests Estimated Closing Balance Sheet Section 2.3(a) Estimated Net Working Capital Amount Section 6.6(c) Exchange Act Section 6.6(c)	Covered Matters	Section 13.8(a)
Disagreement Notice Divestiture Section 2.3(e) Section 2.5(d) Divestiture Notice Section 2.5(d) Draft Allocation Section 11.7 Employee Plan Section 3.11(c) Employment Agreement Section 6.4(a) Employment Commencement Date Section 6.4(a) Environmental Work Section 6.17(c) EPI Group Section 6.11(b) Equity Interests Recitals Estimated Closing Balance Sheet Section 2.3(a) Estimated Net Working Capital Amount Section 6.6(c) Exchange Act Section 6.6(c)	Cure Period	Section 12.2
Divestiture Notice Section 2.5(d) Draft Allocation Section 11.7 Employee Plan Section 3.11(c) Employment Agreement Section 6.4(a) Employment Commencement Date Section 6.4(a) Environmental Work Section 6.17(c) EPI Group Section 6.11(b) Equity Interests Recitals Estimated Closing Balance Sheet Section 2.3(a) Estimated Net Working Capital Amount Section 6.6(c) Exchange Act Section 6.6(c)	Deductible Amount	Section 10.2(b)
Divestiture Notice Draft Allocation Section 2.5(d) Section 11.7 Employee Plan Section 3.11(c) Employment Agreement Section 6.4(a) Employment Commencement Date Section 6.4(a) Environmental Work Section 6.17(c) EPI Group Section 6.11(b) Equity Interests Recitals Estimated Closing Balance Sheet Section 2.3(a) Estimated Net Working Capital Amount Section 6.6(c) Exchange Act Section 6.6(c)	Disagreement Notice	Section 2.3(e)
Draft Allocation Section 11.7 Employee Plan Section 3.11(c) Employment Agreement Section 6.4(a) Employment Commencement Date Section 6.4(a) Environmental Work Section 6.17(c) EPI Group Section 6.11(b) Equity Interests Recitals Estimated Closing Balance Sheet Section 2.3(a) Estimated Net Working Capital Amount Section 2.3(a) EWS Section 6.6(c) Exchange Act Section 6.6(c)	Divestiture	Section 2.5(d)
Employee Plan Employment Agreement Employment Commencement Date Environmental Work Environmental Work EPI Group Section 6.17(c) Equity Interests Estimated Closing Balance Sheet Estimated Net Working Capital Amount EWS Exchange Act Section 3.11(c) Section 6.4(a) Section 6.4(a) Section 6.17(c) Section 6.11(b) Recitals Section 2.3(a) Section 2.3(a) Section 6.6(c)	Divestiture Notice	Section 2.5(d)
Employment Agreement Section 6.4(a) Employment Commencement Date Section 6.4(a) Environmental Work Section 6.17(c) EPI Group Section 6.11(b) Equity Interests Recitals Estimated Closing Balance Sheet Section 2.3(a) Estimated Net Working Capital Amount Section 2.3(a) EWS Section 6.6(c) Exchange Act Section 6.6(c)	Draft Allocation	Section 11.7
Employment Commencement Date Environmental Work EPI Group Equity Interests Estimated Closing Balance Sheet Estimated Net Working Capital Amount EWS Exchange Act Section 6.4(a) Section 6.17(c) Section 6.11(b) Section 6.11(b) Section 2.3(a) Section 2.3(a) Section 6.6(c)	Employee Plan	Section 3.11(c)
Environmental Work EPI Group Section 6.17(c) Section 6.17(b) Equity Interests Estimated Closing Balance Sheet Section 2.3(a) Estimated Net Working Capital Amount EWS Section 6.6(c) Exchange Act Section 6.6(c)	Employment Agreement	Section 6.4(a)
EPI Group Equity Interests Estimated Closing Balance Sheet Estimated Net Working Capital Amount EWS Exchange Act Section 6.11(b) Recitals Section 2.3(a) Section 2.3(a) Section 6.6(c)	Employment Commencement Date	Section 6.4(a)
Equity InterestsRecitalsEstimated Closing Balance SheetSection 2.3(a)Estimated Net Working Capital AmountSection 2.3(a)EWSSection 6.6(c)Exchange ActSection 6.6(c)	Environmental Work	Section 6.17(c)
Equity InterestsRecitalsEstimated Closing Balance SheetSection 2.3(a)Estimated Net Working Capital AmountSection 2.3(a)EWSSection 6.6(c)Exchange ActSection 6.6(c)	EPI Group	Section 6.11(b)
Estimated Net Working Capital Amount EWS Section 2.3(a) Section 6.6(c) Exchange Act Section 6.6(c)	Equity Interests	Recitals
Estimated Net Working Capital Amount EWS Section 2.3(a) Section 6.6(c) Exchange Act Section 6.6(c)	Estimated Closing Balance Sheet	Section 2.3(a)
EWS Section 6.6(c) Exchange Act Section 6.6(c)	Estimated Net Working Capital Amount	Section 2.3(a)
Exchange Act Section 6.6(c)	EWS	
	Exchange Act	
	FCC Application	

Final Allocation	Final Adjustment Amount	Section 2.3(i)
Final Closing Balance Sheet		
Final Net Working Capital Amount		
Financial Statements		
HCERA		
Health Plan		
Health Plan		
HSR Clearance		\ /
Inactive Employees		\ /
Indemnifying Party		
Indemnite		\ /
Indemnity Escrow Agreement	, , ,	
Indemnity Escrow Deposit		
Indemnity Escrow Funds		
Information		
Information		
Intangible Property		
Interim Balance Sheet		Section 6.1(b)
P Agreements	Intangible Property	
Leased Real Property Section 3.7(b) Licensed IP Section 3.10(a) Material Contracts Section 3.8(a) Multi-Station Contract Section 6.13 MVPDs Section 3.4(e) NDA Section 6.1(a) Other Seller Station Section 6.17(a) Outside Date Section 3.10(a) Owned Intangible Property Section 3.10(a) Owned Real Property Section 3.7(a) Permits Section 3.13(b) Phase I Environmental Assessment Section 6.17(a) Phase II Consent Section 6.17(a) Phase II Consent Section 6.17(b) Phase II Election Notice Section 6.17(b) Phase II Report Section 6.17(b) Phase II Report Section 6.17(b) Phase II Request Section 6.17(b) Phase II Time Period Section 6.17(b) Post-Closing Tax Period Section 6.17(b) Post-Closing Tax Refund Section 11.1(b) Post-Closing Tax Refund Section 11.2 Purchase Price Section 3.7(b) Real Property	Interim Balance Sheet	Section 3.15
Licensed IP	IP Agreements	Section 3.8(a)
Material Contracts Section 3.8(a) Multi-Station Contract Section 6.13 MVPDs Section 3.4(e) NDA Section 6.14 Other Seller Station Section 6.13 Outside Date Section 12.1(d) Owned Intangible Property Section 3.10(a) Owned Real Property Section 3.7(a) Permits Section 3.13(b) Phase I Environmental Assessment Section 6.17(a) Phase I Time Period Section 6.17(b) Phase II Consent Section 6.17(b) Phase II Bection Notice Section 6.17(b) Phase II Inspection Section 6.17(b) Phase II Report Section 6.17(c) Phase II Report Section 6.17(b) Phase II Time Period Section 6.17(b) Post-Closing Tax Period Section 11.1(b) Post-Closing Tax Refund Section 11.1(b) Post-Closing Tax Refund Section 11.2 Purchase Price Section 3.7(b) Real Property Section 3.7(b) Real Property Leases Section 3.7(b) Real Property Leases	Leased Real Property	Section 3.7(b)
Multi-Station Contract Section 6.13 MVPDs Section 3.4(e) NDA Section 6.1(a) Other Seller Station Section 6.13 Outside Date Section 12.1(d) Owned Intangible Property Section 3.10(a) Owned Real Property Section 3.7(a) Permits Section 3.13(b) Phase I Environmental Assessment Section 6.17(a) Phase I Time Period Section 6.17(a) Phase II Consent Section 6.17(b) Phase II Election Notice Section 6.17(b) Phase II I Environmental Assessment Section 6.17(b) Phase II Report Section 6.17(b) Phase II Report Section 6.17(b) Phase II Request Section 6.17(b) Phase II Time Period Section 6.17(b) Post-Closing Tax Period Section 11.1(b) Post-Closing Tax Refund Section 11.4 Pre-Closing Tax Return Section 11.2 Purchase Price Section 3.7(b) Real Property Section 3.7(b) Real Property Leases Section 3.7(b) Real Proper	Licensed IP	Section 3.10(a)
Multi-Station Contract Section 6.13 MVPDs Section 3.4(e) NDA Section 6.1(a) Other Seller Station Section 6.13 Outside Date Section 12.1(d) Owned Intangible Property Section 3.10(a) Owned Real Property Section 3.7(a) Permits Section 3.13(b) Phase I Environmental Assessment Section 6.17(a) Phase I Time Period Section 6.17(a) Phase II Consent Section 6.17(b) Phase II Election Notice Section 6.17(b) Phase II I Environmental Assessment Section 6.17(b) Phase II Report Section 6.17(b) Phase II Report Section 6.17(b) Phase II Request Section 6.17(b) Phase II Time Period Section 6.17(b) Post-Closing Tax Period Section 11.1(b) Post-Closing Tax Refund Section 11.4 Pre-Closing Tax Return Section 11.2 Purchase Price Section 3.7(b) Real Property Section 3.7(b) Real Property Leases Section 3.7(b) Real Proper	Material Contracts	Section 3.8(a)
NDA	Multi-Station Contract	
NDA	MVPDs	
Other Seller Station Section 6.13 Outside Date Section 12.1(d) Owned Intangible Property Section 3.10(a) Owned Real Property Section 3.7(a) Permits Section 3.13(b) Phase I Environmental Assessment Section 6.17(a) Phase I Time Period Section 6.17(a) Phase II Consent Section 6.17(b) Phase II Election Notice Section 6.17(b) Phase II Environmental Assessment Section 6.17(b) Phase II Report Section 6.17(b) Phase II Request Section 6.17(c) Phase II Time Period Section 6.17(b) Post-Closing Tax Period Section 11.1(b) Post-Closing Tax Refund Section 11.4 Pre-Closing Tax Return Section 11.2 Purchase Price Section 2.2 Real Property Section 3.7(b) Real Property Leases Section 3.7(b) Recognized Environmental Condition Section 3.10(a) Registered IP Section 3.10(a) Required Consents Section 6.3 Reviewed Closing Balance Sheet Section 2.3(d) <td></td> <td></td>		
Outside Date Section 12.1(d) Owned Intangible Property Section 3.10(a) Owned Real Property Section 3.7(a) Permits Section 3.13(b) Phase I Environmental Assessment Section 6.17(a) Phase I Time Period Section 6.17(b) Phase II Consent Section 6.17(b) Phase II Election Notice Section 6.17(b) Phase II Environmental Assessment Section 6.17(b) Phase II Inspection Section 6.17(b) Phase II Report Section 6.17(b) Phase II Time Period Section 6.17(b) Phase II Time Period Section 6.17(b) Post-Closing Tax Period Section 11.1(b) Post-Closing Tax Refund Section 11.4 Pre-Closing Tax Return Section 11.2 Purchase Price Section 2.2 Real Property Section 3.7(b) Real Property Leases Section 3.7(b) Recognized Environmental Condition Section 6.17(b) Registered IP Section 3.10(a) Related Party Transactions Section 3.19 Required Consents Section 2.3(d)<		
Owned Intangible Property Owned Real Property Section 3.10(a) Permits Section 3.7(a) Permits Section 3.13(b) Phase I Environmental Assessment Phase I Time Period Section 6.17(a) Phase II Consent Section 6.17(b) Phase II Election Notice Section 6.17(b) Phase II I Environmental Assessment Section 6.17(b) Phase II I Environmental Assessment Section 6.17(b) Phase II Report Section 6.17(b) Phase II Report Section 6.17(c) Phase II Request Section 6.17(b) Phase II Time Period Section 6.17(b) Post-Closing Tax Period Section 11.1(b) Post-Closing Tax Refund Section 11.4 Pre-Closing Tax Refund Section 11.4 Pre-Closing Tax Return Section 11.2 Purchase Price Section 2.2 Real Property Section 3.7(b) Real Property Leases Section 3.7(b) Reognized Environmental Condition Section 3.10(a) Related Party Transactions Section 3.19 Required Consents Section 2.3(d)		
Owned Real PropertySection 3.7(a)PermitsSection 3.13(b)Phase I Environmental AssessmentSection 6.17(a)Phase I Time PeriodSection 6.17(b)Phase II ConsentSection 6.17(b)Phase II Election NoticeSection 6.17(b)Phase II Environmental AssessmentSection 6.17(b)Phase II InspectionSection 6.17(b)Phase II ReportSection 6.17(c)Phase II RequestSection 6.17(b)Phase II Time PeriodSection 6.17(b)Post-Closing Tax PeriodSection 11.1(b)Post-Closing Tax RefundSection 11.4Pre-Closing Tax ReturnSection 11.2Purchase PriceSection 2.2Real PropertySection 3.7(b)Real Property LeasesSection 3.7(b)Recognized Environmental ConditionSection 3.10(a)Registered IPSection 3.10(a)Related Party TransactionsSection 3.19Required ConsentsSection 2.3(d)		
Permits Section 3.13(b) Phase I Environmental Assessment Section 6.17(a) Phase I Time Period Section 6.17(a) Phase II Consent Section 6.17(b) Phase II Election Notice Section 6.17(b) Phase II Environmental Assessment Section 6.17(b) Phase II Inspection Section 6.17(b) Phase II Report Section 6.17(c) Phase II Request Section 6.17(b) Phase II Time Period Section 6.17(b) Post-Closing Tax Period Section 11.1(b) Post-Closing Tax Refund Section 11.1(a) Pre-Closing Tax Return Section 11.2 Purchase Price Section 3.7(b) Real Property Leases Section 3.7(b) Recognized Environmental Condition Section 3.10(a) Related Party Transactions Section 6.3 Reviewed Closing Balance Sheet Section 2.3(d)		\ /
Phase I Environmental Assessment Phase I Time Period Section 6.17(a) Phase II Consent Section 6.17(b) Phase II Election Notice Section 6.17(b) Phase II Environmental Assessment Section 6.17(b) Phase II Inspection Section 6.17(b) Phase II Report Section 6.17(c) Phase II Request Section 6.17(b) Phase II Time Period Section 6.17(b) Post-Closing Tax Period Section 11.1(b) Post-Closing Tax Refund Section 11.4 Pre-Closing Tax Return Section 11.2 Purchase Price Section 3.7(b) Real Property Leases Section 3.7(b) Recognized Environmental Condition Required Consents Section 3.19 Required Consents Section 2.3(d)		
Phase I Time Period Section 6.17(a) Phase II Consent Section 6.17(b) Phase II Election Notice Section 6.17(b) Phase II Environmental Assessment Section 6.17(b) Phase II Inspection Section 6.17(b) Phase II Report Section 6.17(c) Phase II Request Section 6.17(b) Phase II Time Period Section 6.17(b) Post-Closing Tax Period Section 11.1(b) Post-Closing Tax Refund Section 11.1(a) Pre-Closing Tax Return Section 11.2 Purchase Price Section 3.7(b) Real Property Leases Section 3.7(b) Recognized Environmental Condition Section 3.10(a) Related Party Transactions Section 6.3 Reviewed Closing Balance Sheet Section 2.2(d)		\ /
Phase II Consent Phase II Election Notice Section 6.17(b) Phase II Election Notice Section 6.17(b) Phase II Environmental Assessment Section 6.17(b) Phase II Inspection Section 6.17(b) Phase II Report Section 6.17(c) Phase II Request Section 6.17(b) Phase II Time Period Section 6.17(b) Post-Closing Tax Period Section 11.1(b) Post-Closing Tax Refund Section 11.4 Pre-Closing Tax Period Section 11.1(a) Pre-Closing Tax Return Section 11.2 Purchase Price Section 2.2 Real Property Section 3.7(b) Real Property Leases Section 3.7(b) Recognized Environmental Condition Section 3.10(a) Related Party Transactions Section 6.3 Reviewed Closing Balance Sheet Section 2.3(d)		
Phase II Election Notice Phase II Environmental Assessment Section 6.17(b) Phase II Inspection Section 6.17(b) Phase II Report Section 6.17(c) Phase II Request Section 6.17(b) Phase II Time Period Post-Closing Tax Period Section 11.1(b) Post-Closing Tax Refund Section 11.4 Pre-Closing Tax Period Section 11.1(a) Pre-Closing Tax Return Section 11.2 Purchase Price Section 2.2 Real Property Section 3.7(b) Recognized Environmental Condition Section 3.10(a) Related Party Transactions Required Consents Section 2.3(d)		()
Phase II Environmental Assessment Phase II Inspection Phase II Report Section 6.17(b) Phase II Report Section 6.17(c) Phase II Request Section 6.17(b) Phase II Time Period Post-Closing Tax Period Section 11.1(b) Post-Closing Tax Refund Pre-Closing Tax Refund Section 11.4 Pre-Closing Tax Return Section 11.1(a) Pre-Closing Tax Return Section 11.2 Purchase Price Section 2.2 Real Property Section 3.7(b) Real Property Leases Section 3.7(b) Recognized Environmental Condition Section 3.17(b) Registered IP Section 3.10(a) Related Party Transactions Section 6.3 Reviewed Closing Balance Sheet Section 2.3(d)		\ /
Phase II Inspection Phase II Report Section 6.17(b) Phase II Request Section 6.17(c) Phase II Request Section 6.17(b) Phase II Time Period Section 6.17(b) Post-Closing Tax Period Section 11.1(b) Post-Closing Tax Refund Section 11.4 Pre-Closing Tax Period Section 11.1(a) Pre-Closing Tax Return Section 11.2 Purchase Price Section 2.2 Real Property Section 3.7(b) Real Property Leases Section 3.7(b) Recognized Environmental Condition Section 6.17(b) Registered IP Section 3.10(a) Related Party Transactions Section 6.3 Reviewed Closing Balance Sheet Section 2.3(d)		\ /
Phase II Report Phase II Request Section 6.17(b) Phase II Time Period Post-Closing Tax Period Post-Closing Tax Refund Pre-Closing Tax Refund Section 11.4 Pre-Closing Tax Return Section 11.1(a) Pre-Closing Tax Return Section 11.2 Purchase Price Section 2.2 Real Property Section 3.7(b) Real Property Leases Section 3.7(b) Recognized Environmental Condition Section 3.10(a) Related Party Transactions Section 3.19 Required Consents Section 2.3(d)		\ /
Phase II Request Phase II Time Period Section 6.17(b) Post-Closing Tax Period Section 11.1(b) Post-Closing Tax Refund Pre-Closing Tax Period Section 11.4 Pre-Closing Tax Period Section 11.1(a) Pre-Closing Tax Return Section 11.2 Purchase Price Section 2.2 Real Property Section 3.7(b) Real Property Leases Section 3.7(b) Recognized Environmental Condition Section 6.17(b) Registered IP Section 3.10(a) Related Party Transactions Section 6.3 Reviewed Closing Balance Sheet Section 2.3(d)	•	(/
Phase II Time Period Section 6.17(b) Post-Closing Tax Period Section 11.1(b) Post-Closing Tax Refund Section 11.4 Pre-Closing Tax Period Section 11.1(a) Pre-Closing Tax Return Section 11.2 Purchase Price Section 2.2 Real Property Section 3.7(b) Real Property Leases Section 3.7(b) Recognized Environmental Condition Section 6.17(b) Registered IP Section 3.10(a) Related Party Transactions Section 6.3 Required Consents Section 2.3(d)	-	
Post-Closing Tax Period Post-Closing Tax Refund Pre-Closing Tax Period Pre-Closing Tax Period Pre-Closing Tax Return Pre-Closing Tax Return Purchase Price Real Property Real Property Recognized Environmental Condition Registered IP Related Party Transactions Required Consents Reviewed Closing Balance Sheet Section 11.1 Section 1		. ,
Post-Closing Tax Refund Pre-Closing Tax Period Pre-Closing Tax Return Section 11.1(a) Pre-Closing Tax Return Section 11.2 Purchase Price Section 2.2 Real Property Section 3.7(b) Real Property Leases Section 3.7(b) Recognized Environmental Condition Section 6.17(b) Registered IP Section 3.10(a) Related Party Transactions Section 3.19 Required Consents Section 2.3(d)		. ,
Pre-Closing Tax Period Section 11.1(a) Pre-Closing Tax Return Section 11.2 Purchase Price Section 2.2 Real Property Section 3.7(b) Real Property Leases Section 3.7(b) Recognized Environmental Condition Section 6.17(b) Registered IP Section 3.10(a) Related Party Transactions Section 3.19 Required Consents Section 6.3 Reviewed Closing Balance Sheet Section 2.3(d)		. ,
Pre-Closing Tax Return Purchase Price Real Property Section 2.2 Real Property Section 3.7(b) Recognized Environmental Condition Registered IP Related Party Transactions Required Consents Reviewed Closing Balance Sheet Section 11.2 Section 2.2 Section 3.7(b) Section 3.7(b) Section 6.17(b) Section 3.10(a) Section 3.19 Section 6.3 Section 6.3		
Purchase Price Section 2.2 Real Property Section 3.7(b) Real Property Leases Section 3.7(b) Recognized Environmental Condition Section 6.17(b) Registered IP Section 3.10(a) Related Party Transactions Section 3.19 Required Consents Section 6.3 Reviewed Closing Balance Sheet Section 2.3(d)		· /
Real PropertySection 3.7(b)Real Property LeasesSection 3.7(b)Recognized Environmental ConditionSection 6.17(b)Registered IPSection 3.10(a)Related Party TransactionsSection 3.19Required ConsentsSection 6.3Reviewed Closing Balance SheetSection 2.3(d)		
Real Property LeasesSection 3.7(b)Recognized Environmental ConditionSection 6.17(b)Registered IPSection 3.10(a)Related Party TransactionsSection 3.19Required ConsentsSection 6.3Reviewed Closing Balance SheetSection 2.3(d)		
Recognized Environmental ConditionSection 6.17(b)Registered IPSection 3.10(a)Related Party TransactionsSection 3.19Required ConsentsSection 6.3Reviewed Closing Balance SheetSection 2.3(d)		
Registered IPSection 3.10(a)Related Party TransactionsSection 3.19Required ConsentsSection 6.3Reviewed Closing Balance SheetSection 2.3(d)	. ,	
Related Party TransactionsSection 3.19Required ConsentsSection 6.3Reviewed Closing Balance SheetSection 2.3(d)		· /
Required Consents Section 6.3 Reviewed Closing Balance Sheet Section 2.3(d)		` ` ` `
Reviewed Closing Balance Sheet Section 2.3(d)		
Reviewed Net Working Capital Amount Section 2.3(d)		
	Reviewed Net Working Capital Amount	Section 2.3(d)

Reviewed Net Working Capital Overage	Section 2.3(h)
Reviewed Net Working Capital Shortfall	Section 2.3(h)
SEC	Section 6.6(c)
Securities Act	Section 4.9
Seller	Preamble
Seller Ancillary Agreements	Section 3.1(a)
Seller's 401(k) Plan	Section 6.4(e)
Seller Indemnified Party and Seller Indemnified Parties	Section 10.3(a)
Seller Taxes	Section 11.1(a)
Sharing Agreement	Section 3.4(d)
Station and Stations	Recitals
Station Employees	Section 3.11(b)
Straddle Period	Section 11.2
Straddle Tax Return	Section 11.3(c)
Surveys	Section 6.9
Tangible Personal Property	Section 3.6(a)
Tax Claim	Section 11.8(a)
Third Party Action	Section 10.4(a)
Title Commitments	Section 6.9
Transfer Taxes	Section 11.6
Transition Services Agreement	Section 9.1(f)

1.3 Interpretation. Article titles and section headings herein are for convenience of reference only and are not intended to affect the meaning or interpretation of this Agreement. The Schedules hereto shall be construed with and as an integral part of this Agreement to the same extent as if set forth verbatim herein. The information set forth in each Schedule to this Agreement shall qualify the corresponding section or subsection of this Agreement and any other section or subsection of this Agreement, if and only to the extent that it is readily apparent, based on the face of such disclosure, that it applies to such other section or subsection of this Agreement. Where the context so requires or permits, the use of the singular form includes the plural, and the use of the plural form includes the singular. When used in this Agreement, unless the context clearly requires otherwise, (a) words such as "herein", "hereof", "hereto", "hereby" and "hereafter" shall refer to this Agreement as a whole, (b) the terms "include", 'includes" and "including" shall not be limiting, but shall be deemed to be followed by the phrase "without limitation", (c) the word "or" shall not be exclusive, (d) references to Articles, Sections, Schedules and Exhibits mean the Articles and Sections of, and Schedules and Exhibits to, this Agreement, and (e) the terms "Dollars", "dollars" and "\$" each mean lawful money of the United States of America.

ARTICLE 2

PURCHASE OF EQUITY INTERESTS

- **2.1 Purchase of Equity Interests**. On the terms and subject to the conditions hereof, at the Closing, Seller shall sell and transfer to Buyer, and Buyer shall purchase from Seller, the Equity Interests, free and clear of all Liens, in exchange for the Purchase Price.
- **2.2 Purchase Price**. At the Closing and subject to Section 2.5, Buyer shall pay or cause to be paid an aggregate amount equal to Five Hundred Twenty One Million Dollars (\$521,000,000) (the "*Purchase Price*"), subject to adjustment as set forth in Section 2.3, by wire transfer of immediately available funds, as follows:
- (a) Buyer shall pay or cause to be paid all Indebtedness of the Acquired Companies outstanding as of immediately prior to the Closing, which shall be set forth on the Closing Indebtedness Certificate that Seller shall deliver to Buyer at least three Business Days before Closing. In order to facilitate such repayment, prior to the Closing, Seller shall obtain and deliver to Buyer payoff letters for all

such Indebtedness, which payoff letters shall indicate that the lenders of such Indebtedness have agreed to release all Liens in respect of such Indebtedness relating to the Equity Interests or the assets and properties of the Acquired Companies upon receipt of the amounts indicated in such payoff letters. Buyer shall be entitled to rely conclusively on the amounts and other information set forth in the Closing Indebtedness Certificate and such payoff letters.

- **(b)** Buyer shall pay or cause to be paid, on behalf of the Acquired Companies, all Company Transaction Costs payable by any of the Acquired Companies that remain outstanding as of the Closing Date, which shall be set forth on the Closing Transaction Costs Certificate that Seller shall deliver to Buyer at least three Business Days before Closing, to such account or accounts as are set forth on the Closing Transaction Costs Certificate. Buyer shall be entitled to rely conclusively on the amounts and other information set forth in the Closing Transaction Costs Certificate.
- **(c)** Buyer shall pay or cause to be paid to an account designated by the Escrow Agent, the Indemnity Escrow Deposit.
- (d) Buyer shall pay or cause to be paid to an account designated by Seller, the Purchase Price less the amounts described in Sections 2.2(a), 2.2(b) and 2.2(c) above, subject to adjustment as provided in Section 2.3.

2.3 Working Capital Adjustment.

- (a) No later than five (5) Business Days prior to the Closing Date, Seller shall deliver to Buyer an unaudited pro forma estimated balance sheet for the Acquired Companies, as of the Reference Time (the "Estimated Closing Balance Sheet"), together with Seller's good faith estimated, itemized calculation, based on the Estimated Closing Balance Sheet, of the Net Working Capital Amount determined as of the Reference Time (the "Estimated Net Working Capital Amount"). The Estimated Closing Balance Sheet shall be prepared in accordance with GAAP. A sample Estimated Closing Balance Sheet prepared as if the Closing Date was September 30, 2018 is attached hereto as Exhibit F.
- (b) At any time prior to the Closing Date, Buyer shall have the right to object to the Estimated Closing Balance Sheet or the Estimated Net Working Capital Amount by delivering written notice of such objection to Seller. If Buyer does not provide such written notice of objection to Seller, then the Estimated Closing Balance Sheet and the Estimated Net Working Capital Amount shall, subject to and without limiting the remaining provisions of this Section 2.3, be deemed accepted by Buyer. If, on the other hand, Buyer provides such written notice of objection to Seller, then representatives of Buyer and Seller shall meet as promptly as practicable to discuss in good faith the proper Estimated Closing Balance Sheet and the proper Estimated Net Working Capital Amount; provided, however, that if such representatives of Buyer and Seller are unable to agree in good faith prior to the Closing Date on the proper Estimated Closing Balance Sheet and the proper Estimated Net Working Capital Amount, then the Estimated Net Working Capital Amount shall be deemed to be equal to the Estimated Net Working Capital Amount as presented by Seller, subject to and without limiting the remaining provisions of this Section 2.3.
- **(c)** At the Closing, the Purchase Price shall be (i) increased by the Estimated Net Working Capital Amount if the Estimated Net Working Capital Amount is a positive number, and (ii) decreased by the absolute value of the Estimated Net Working Capital Amount, if the Estimated Net Working Capital Amount is a negative number.
- (d) Within one hundred twenty (120) days following the Closing Date, Buyer shall prepare and deliver to Seller an unaudited balance sheet for the Acquired Companies determined as of the Reference Time (the "Reviewed Closing Balance Sheet"), together with an itemized calculation, based on the Reviewed Closing Balance Sheet, of the Net Working Capital Amount determined as of the Reference Time (the "Reviewed Net Working Capital Amount"). The Reviewed Closing Balance Sheet shall be prepared in accordance with GAAP using the same accounting principles, policies and methods

as were used by Seller in preparing the Estimated Closing Balance Sheet in connection with the calculation of each of the line items reflected thereon.

- (e) If Seller disagrees with the Reviewed Closing Balance Sheet or Buyer's calculation of the Reviewed Net Working Capital Amount, then during the thirty (30) days following the date of Seller's receipt of the Reviewed Closing Balance Sheet, Buyer and the Acquired Companies shall each provide Seller with access to the working papers of Buyer and the Acquired Companies relating to the Reviewed Closing Balance Sheet and the resulting calculation of the Reviewed Net Working Capital Amount, as well as any other information used in preparing the Reviewed Closing Balance Sheet as is reasonably requested by Seller; provided, that such access shall be in a manner that does not interfere with the normal business operations of Buyer or any of the Acquired Companies. The Reviewed Closing Balance Sheet and the resulting calculation of the Reviewed Net Working Capital Amount shall become final and binding and deemed accepted by Seller at the end of such thirty (30) day period unless, prior to the end of such period, Seller has delivered to Buyer written notice of its disagreement with such Reviewed Closing Balance Sheet and the resulting calculation of the Reviewed Net Working Capital Amount (a "Disagreement Notice"), specifying the nature and amount of any disputed item.
- In the event that any Disagreement Notice is timely provided to Buyer pursuant to Section 2.3(e), Buyer and Seller shall cooperate for a period of thirty (30) days following Buyer's receipt of the Disagreement Notice (or such longer period as Buyer and Seller may mutually agree) to resolve any disagreements set forth in the Disagreement Notice with respect to the Reviewed Closing Balance Sheet and the Reviewed Net Working Capital Amount. If, at the end of such thirty (30) day period, Buyer and Seller are unable to resolve all such disagreements with respect to the Reviewed Closing Balance Sheet or the Reviewed Net Working Capital Amount, then a mutually acceptable, nationally recognized independent accounting firm that does not then have a relationship with Seller or Buyer (the "Independent Accountant") shall be engaged to resolve such remaining disagreements with respect to the Reviewed Closing Balance Sheet and/or the Reviewed Net Working Capital Amount; provided, however, that if Buyer and Seller are unable to agree on the choice of such firm within 10 Business Days following the end of such 30 day period, then such firm will be a nationally recognized accounting firm selected by lot (after Seller and Buyer shall have each submitted two proposed firms and then excluded one firm designated by the other party). Buyer and Seller shall instruct the Independent Accountant to determine as promptly as practicable, but in any event within thirty (30) days of the date on which such dispute is referred to the Independent Accountant, whether and to what extent the Reviewed Closing Balance Sheet and/or the Reviewed Net Working Capital Amount requires adjustment; provided, however, that the Independent Accountant shall be authorized to resolve only those items remaining in dispute between Buyer and Seller as specified in the Disagreement Notice, in each case within the range of the difference between Buyer's position with respect thereto and Seller's position with respect thereto. The fees and expenses of the Independent Accountant (including any indemnity obligations to the Independent Accountant) shall be allocated between Buyer and Seller based on their relative success with respect to the disputed items (as finally determined by the Independent Accountant). The fees and expenses (if any) of Buyer's independent auditors and attorneys incurred in connection with the review of the Disagreement Notice shall be borne by Buyer, and the fees and expenses (if any) of Seller's independent auditors and attorneys incurred in connection with their review of the Reviewed Net Working Capital Amount shall be borne by Seller.
- Working Capital Amount, in the event that Seller does not provide a Disagreement Notice to Buyer within the thirty (30) day period provided for in Section 2.3(e) or (ii) the as-adjusted Reviewed Net Working Capital Amount as determined by the parties or the Independent Accountant pursuant to Section 2.3(f), in the event that Seller provides a Disagreement Notice to Buyer within the thirty (30) day period provided for in Section 2.3(e). The Final Net Working Capital Amount shall be set forth on the "Final Closing Balance Sheet", which shall be (x) the Reviewed Closing Balance Sheet in the event that Seller does not provide a Disagreement Notice to Buyer within the thirty (30) day period provided for in Section 2.3(e) or (y) the as-adjusted Reviewed Closing Balance Sheet as determined by the parties or the Independent Accountant pursuant to Section 2.3(f), in the event that Seller provides a Disagreement Notice to Buyer within the thirty (30) day period provided for in Section 2.3(e).

- (h) If the Estimated Net Working Capital Amount is greater than the Final Net Working Capital Amount, then the difference between the Estimated Net Working Capital Amount, *minus* the Final Net Working Capital Amount shall be referred to as a "Reviewed Net Working Capital Shortfall". If the Final Net Working Capital Amount is greater than the Estimated Net Working Capital Amount, then the difference between the Final Net Working Capital Amount, *minus* the Estimated Net Working Capital Amount shall be referred to as a "Reviewed Net Working Capital Overage".
- (i) The "Final Adjustment Amount" shall be an amount equal to either the Reviewed Net Working Capital Shortfall or the Reviewed Net Working Capital Overage, as applicable. In the event that the Final Adjustment Amount is a Reviewed Net Working Capital Shortfall, the Purchase Price shall be reduced by such amount and Seller shall pay such amount to Buyer by wire transfer in immediately available funds to an account designated by Buyer. In the event that the Final Adjustment Amount is a Reviewed Net Working Capital Overage, the Purchase Price shall be increased by such amount and Buyer shall pay such amount to Seller by wire transfer in immediately available funds to an account(s) designated by Seller.
- (j) Any payments to be made pursuant to Section 2.3(i) shall be (i) made within five (5) days of the final determination of the Final Adjustment Amount pursuant to Section 2.3(i) and (ii) treated as an adjustment to the Purchase Price by the parties for Tax purposes, unless otherwise required by Law.

2.4 Closing.

- (a) Subject to any prior termination of this Agreement pursuant to Section 12.1, the consummation of the sale and purchase of the Equity Interests pursuant to this Agreement (the "Closing") shall take place at the offices of Cooley LLP located at 1299 Pennsylvania Avenue, N.W., Washington D.C. 20004 on (i) the fifth (5th) Business Day after the date that is the later of (x) the date on which the FCC Consent shall have been granted and shall be in full force and effect, and (y) the date on which the HSR Clearance has been obtained, or (ii) such other date or at such other location as is mutually agreed to by Buyer and Seller in writing (as applicable, the "Closing Date"), subject to the satisfaction or waiver of the conditions to Closing set forth herein (other than those conditions that by their nature are to be satisfied at the Closing but subject to the satisfaction or waiver of such conditions at the Closing); provided, that if the Marketing Period has not ended on or prior to the time the Closing would have otherwise been required to occur pursuant to the foregoing, the Closing shall not occur until the third Business Day following the final day of the Marketing Period, subject to the satisfaction or waiver of the conditions to Closing set forth herein (other than those conditions that by their nature are to be satisfied at the Closing but subject to the satisfaction or waiver of such conditions at the Closing but subject to the satisfaction or waiver of such conditions at the Closing).
- **(b)** A breach by a party of its obligations to effect the Closing pursuant to the terms and subject to the conditions of this Agreement, including this Section 2.4, shall be subject to Section 12.1(b) or Section 12.1(c), as applicable (and shall not be subject to the Cure Period under Section 12.2).
- **(c)** For purposes of this Agreement, the Closing shall be effective at 12:01 a.m. Eastern Time on the Closing Date.

2.5 Governmental Consents.

(a) Within ten (10) Business Days of the date of this Agreement, Buyer and Seller shall, and Seller shall cause the Acquired Companies to, jointly file an application or applications with the FCC requesting the grant of its consent to the transfer of control of the Acquired Companies from Seller to Buyer (collectively, the "FCC Application"). Any FCC filing fees relating to the transactions contemplated hereby shall be borne one-half (1/2) by Buyer and one-half (1/2) by Seller, irrespective of whether the transactions contemplated hereby are consummated. To the extent applicable, Buyer shall include in the FCC Application an exhibit requesting that the FCC grant continued satellite status for the Stations currently operating as satellite stations under FCC rules. Buyer and Seller shall, and Seller shall cause the Acquired Companies to, diligently prosecute the FCC Application and otherwise use their

reasonable best efforts to obtain the FCC Consent as soon as practicable. Seller and Buyer each shall oppose any petitions to deny or other objections filed with respect to the FCC Application to the extent such petition or objection relates to such party. Neither Buyer nor Seller shall, and Seller shall cause the Acquired Companies not to, take any intentional action which would, or intentionally fail to take such action the failure of which to take would, reasonably be expected to materially delay, materially impede or prevent receipt of the grant of the FCC Consent.

- (b) If applicable, within ten (10) Business Days after the date of this Agreement, Buyer and Seller shall make any required filings with the FTC and the DOJ pursuant to the HSR Act with respect to the transactions contemplated hereby (including a request for early termination of the waiting period thereunder), and shall thereafter promptly respond to all requests received from such agencies for additional information or documentation and otherwise use their reasonable best efforts to obtain the expiration or termination of any applicable waiting period under the HSR Act (the "HSR Clearance"). Any filing fees payable under the HSR Act relating to the transactions contemplated hereby shall be borne one-half (1/2) by Buyer and one-half (1/2) by Seller, irrespective of whether the transactions contemplated hereby are consummated.
- **(c)** If the Closing shall not have occurred for any reason within the original effective period of the FCC Consent, and neither party shall have terminated under Section 12.1, Buyer and Seller shall jointly request an extension period of the FCC Consent. No extension of the FCC Consent shall limit the rights of either party under Section 12.1.
- Notwithstanding anything in this Agreement to the contrary, and in addition to the other covenants set forth in this Agreement. Buyer agrees to take promptly any and all steps necessary to eliminate each and every impediment and obtain all consents under any antitrust or competition Law (including the HSR Act) or any Communications Law that may be required by the FCC, the FTC, the DOJ, any state Attorney General or any other U.S. federal, state or local governmental authority, or any applicable non-U.S. antitrust or competition governmental authority, in each case having competent jurisdiction, so as to enable the parties to close the transactions contemplated by this Agreement as promptly as practicable, including committing to or effecting, by consent decree, pocket consent decree, hold separate orders, trust or otherwise, such divestitures of the Stations (each, a "Divestiture") as are required in order to obtain the FCC Consent or the HSR Clearance and to avoid the entry of (or to effect the dissolution of or vacate or lift) any order that would otherwise have the effect of preventing or materially delaying the consummation of the transactions contemplated by this Agreement. Notwithstanding anything to the contrary in this Section 2.5(d), if any of the consents or approvals (or elimination of impediments) contemplated by the preceding sentence have not been obtained (or eliminated), in each case as of the date that is six (6) months following the date hereof, and if Seller, after consultation with Buyer, reasonably determines in good faith, or, if at any time after the date hereof, the FCC, the FTC, the DOJ, any state Attorney General or any other U.S. federal, state or local governmental authority, or any applicable non-U.S. antitrust or competition governmental authority, has indicated, that a Divestiture is required to obtain the FCC Consent or the HSR Clearance, or otherwise to remove any impediment or to obtain any required consents under any antitrust or competition Law or under the Communications Laws in connection with the consummation of the transactions contemplated hereby, then Seller shall have the right to provide written notice of such determination or indication to Buyer (a "Divestiture Notice"). Upon receipt of a Divestiture Notice, Buyer shall promptly (and in all respects prior to the Outside Date) implement or cause to be implemented such Divestiture. Further, and for the avoidance of doubt, Buyer shall take any and all actions necessary in order to ensure that (x) no requirement for any non-action, consent or approval of the FCC, the FTC, the DOJ, any state Attorney General or any other U.S. federal, state or local governmental authority, or any applicable non-U.S. antitrust or competition governmental authority, (y) no decree, judgment, injunction, temporary restraining order or any other order in any suit or proceeding, and (z) no other matter relating to any antitrust or competition Law or any Communications Law would preclude consummation of the transactions contemplated by this Agreement on or before the Outside Date.

- In connection with their obligations pursuant to this Section 2.5 with respect to pursuing the FCC Consent and the HSR Clearance, Buyer and Seller shall (i) keep each other informed in all material respects and on a reasonably timely basis of any material communication received by such party from, or given by such party to, any Governmental Entity and of any material communication received or given in connection with any claim, suit, action, complaint, charge, or arbitration proceeding brought by a private party before a Governmental Entity, in each case with respect to this Agreement, the Stations or the transactions contemplated hereby, (ii) notify each other of all documents filed with, submitted to or received from any Governmental Entity with respect to this Agreement, the Stations or the transactions contemplated hereby, (iii) furnish each other with such information and assistance as the other may reasonably request in connection with their preparation of any governmental filing or submission hereunder and (iv) reasonably cooperate with each other in connection with and in advance of any filing or submission with a Governmental Entity in connection with the transactions contemplated by this Agreement and in connection with any investigation or other inquiry by or before any Governmental Entity relating to this Agreement, the Stations or the transactions contemplated hereby. including any claim, suit, action, complaint, charge, or arbitration proceeding initiated by a private party. Subject to applicable Laws relating to the exchange of information, Buyer, on the one hand, and Seller, on the other hand, (x) shall have the right to review in advance, and to the extent practicable each will consult with each other on, all information that appears in any filing made with, or written materials submitted to, any Governmental Entity with respect to this Agreement, the Stations or the transactions contemplated hereby, and (v) shall give the other a reasonable opportunity to attend and participate in meetings and telephone conferences with any such Governmental Entity relating to the foregoing. Neither Buyer nor Seller shall file any amendment to the FCC Application or, after grant of the FCC Consent, request any modification of the FCC Consent without the consent of the other party, such consent not to be unreasonably withheld or delayed.
- **2.6 Withholding**. Buyer or its designee and each Acquired Company shall be entitled to withhold and deduct from the consideration otherwise payable pursuant to this Agreement such amounts as such Person is required to deduct and withhold with respect to the making of such payment under the Code or any applicable provision of state, local or foreign Tax law. The applicable withholding agent shall use commercially reasonable efforts to reduce or eliminate any such withholding including by requesting any appropriate Tax forms, including IRS Form W-9 or any similar information from Seller. To the extent that amounts are so withheld and paid over to the appropriate Taxing Authority in accordance with applicable Law, such amounts shall be treated for all purposes of this Agreement as having been paid to the Person in respect of which such deduction and withholding were made.

ARTICLE 3

SELLER REPRESENTATIONS AND WARRANTIES

Seller hereby makes the following representations and warranties to Buyer as of the date hereof and as of the Closing:

3.1 Organization and Capitalization.

(a) Each Acquired Company is a limited liability company duly organized, validly existing and in good standing under the laws of the jurisdiction of its organization and has full limited liability company power and authority to own, operate or lease the properties and assets now owned, operated or leased by it and to carry on its business as it has been and is currently conducted. Schedule 3.1(a) sets forth each jurisdiction in which any Acquired Company is licensed or qualified to do business, and each Acquired Company is duly licensed or qualified to do business and is in good standing in each jurisdiction in which the assets or properties owned by it or the operation of its business as currently conducted makes such licensing or qualification necessary. Seller is a limited liability company duly organized, validly existing and in good standing under the laws of South Carolina and has the requisite limited liability company power and authority to execute, deliver and perform this Agreement and all of the

other agreements and instruments to be made by Seller pursuant hereto (collectively, the "Seller Ancillary Agreements") and to consummate the transactions contemplated hereby and thereby.

- The Equity Interests (i) constitute all of the outstanding equity interests of the Acquired Companies, (ii) are owned directly, beneficially and of record by Seller as set forth on Schedule 3.1(b), in each case free and clear of all Liens other than restrictions on transfer under the Securities Act or any applicable state securities law and (iii) are duly authorized, validly issued, fully paid and nonassessable. Upon consummation of the Closing, Buyer shall own all of the Equity Interests, free and clear of all Liens other than restrictions on transfer under the Securities Act or any applicable state securities law and Liens resulting from actions of Buyer. Other than the Equity Interests, there are no issued, reserved for issuance or outstanding (w) equity interests in, other securities of or other ownership interests in any Acquired Company, (x) securities of any Acquired Company convertible into or exchangeable for equity interests in, other securities of or other ownership interests in any Acquired Company, (y) warrants, calls, options or other rights to acquire from any Acquired Company, or other obligations of any Acquired Company to issue, any equity interests in, other securities of or other ownership interests in such Acquired Company or securities directly or indirectly convertible into or exercisable or exchangeable for equity interests in, other securities of or other ownership interests in any Acquired Company or (z) restricted shares, stock appreciation rights, performance units, contingent value rights, "phantom" stock or similar securities or rights that are derivative of, or provide economic benefits based, directly or indirectly, on the value or price of any equity interests in, other securities of or other ownership interests in any Acquired Company. There are no voting trusts, proxies or other agreements or understandings in effect with respect to the voting or transfer of any of the Equity Interests. The Equity Interests were issued in compliance with applicable Laws. The Equity Interests were not issued in violation of the Organizational Documents of any Acquired Company or any other Contract to which Seller or any Acquired Company is a party and are not subject to or in violation of any preemptive or similar rights of any Person.
- **(c)** None of the Acquired Companies owns, or has any interest in, any equity interests in, other securities of or other ownership interests in any other Person.
- **3.2 Authorization**. The execution, delivery and performance of this Agreement and the Seller Ancillary Agreements by Seller and, as applicable, the Acquired Companies, and the consummation by Seller of the transactions contemplated hereby and thereby, have been duly authorized and approved by all necessary limited liability company action of Seller and the Acquired Companies and their respective directors, officers, managers and members and do not require any further authorization or consent of Seller or any of the Acquired Companies or their respective directors, officers, managers or members. This Agreement is, and each Seller Ancillary Agreement when executed and delivered by Seller and the other parties thereto will be, a legal, valid and binding agreement of Seller, enforceable against Seller in accordance with its terms, except in each case as such enforceability may be limited by the Enforceability Exceptions.
- 3.3 No Conflicts. The execution, delivery and performance by Seller of this Agreement and the Seller Ancillary Agreements and the consummation by Seller of the transactions contemplated hereby or thereby does not and will not, with or without notice or the passage of time (a) violate the Organizational Documents of Seller or any of the Acquired Companies, (b) violate, or result in the creation of any Lien under, any Law, Permit, or Governmental Order to which Seller or any Acquired Company or Station is subject, (c) except as set forth on Schedule 3.3(c), result in a breach of, event of default or the creation of any Lien under, result in the acceleration of or create in any party the right to accelerate, terminate, modify or cancel, or require any consent or notice under, any Material Contract, (d) require the consent or approval of, material notice to, or a material filing by Seller or any Acquired Company with, any Governmental Entity except for and in connection with the Governmental Consents or (e) except as set forth on Schedule 3.3(e), require the consent or approval of, or notice to, any third Person (other than a Governmental Entity or under a Material Contract), except, in the case of the foregoing clause (e), as would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect.

3.4 FCC and Programming Distribution Matters.

- (a) Schedule 3.4(a) sets forth a true and complete list of the FCC Licenses and the holders thereof and any Antenna Structure Registration associated with an FCC License for a Station, digital Class A television station, digital low power television station and digital television translator station.
- (i) The FCC Licenses include all of the material FCC authorizations that the Acquired Companies are required by the FCC to hold for the operation of the Stations as currently operated.
- (ii) The FCC Licenses are in effect in accordance with their respective terms and have not been revoked, suspended, canceled, rescinded or terminated, and have not expired. Except as set forth on Schedule 3.4(a)(ii), the FCC Licenses (A) have been issued for the full terms customarily issued by the FCC for the respective type and class of each such FCC License and (B) are not subject to any adverse condition outside the ordinary course, except for those conditions appearing on the face of the FCC Licenses and conditions generally applicable to the type and class of each of such FCC Licenses.
- (iii) An Antenna Structure Registration is on file with the FCC for each FCC License for a Station, Class A television station, digital low power television station and digital television translator station for which such registration is required under the Communications Laws, and the coordinates listed on each such Antenna Structure Registration are consistent with its associated FCC License in accordance with the Communications Laws.
- Except as set forth in Schedule 3.4(b), (i) the Acquired Companies have operated each of the Stations in compliance in all material respects with the Communications Laws and the FCC Licenses and have paid or caused to be paid all FCC regulatory fees due in respect to each of the FCC Licenses and (ii) all material registrations and reports required to have been filed with the FCC relating to the FCC Licenses have been filed. Except as set forth on Schedule 3.4(b) and, except for random selection for EEO audit or complaints and other matters pending at the FCC as to which no notice has been given to an Acquired Company, there is not pending, nor, to Seller's knowledge, threatened, any Action by or before the FCC to revoke, suspend, cancel, rescind or adversely modify any of the Station Licenses (other than proceedings to amend FCC rules of general applicability to commercial full-power, low power television, television translator, and Class A television broadcast stations), nor is there issued or outstanding, by or before the FCC, any order to show cause, notice of violation, notice of apparent liability, or order of forfeiture against the Station Licenses or the Acquired Companies with respect to the Station Licenses that would reasonably be expected to result in any such action. Except as set forth on Schedule 3.4(b) and except for proceedings affecting, respectively, commercial full-power, low power television, television translator, or Class A television broadcast stations generally and random selection for EEO audit or complaints and other matters pending at the FCC as to which no notice has been given to an Acquired Company, there are no material applications, petitions, proceedings or other material Actions or complaints pending or, to Seller's knowledge, threatened before the FCC against the Station Licenses.
- **(c)** Neither Seller nor any Acquired Company is required to obtain a waiver of or exemption from any provision of the Communications Laws, with respect to Seller or any Acquired Company for the FCC Consent to be obtained.
- (d) Except as set forth on Schedule 3.4(d), none of the Acquired Companies is a party to any outstanding local marketing agreement, time brokerage agreement, joint sales agreement or other similar agreement (collectively, a "Sharing Agreement"). Schedule 3.4(a) separately identifies each Sharing Agreement Station License. Except as may be set forth on Schedule 3.4(a) or 3.4(d), (i) the list of Sharing Agreement Station Licenses in Schedule 3.4(a) includes all of the material licenses and permits required by the FCC for operation of the Sharing Agreement Station as currently operated, (ii) the representations in Section 3.4(a)(ii) in respect of the FCC Licenses are likewise true and correct in

respect of the Sharing Agreement Station Licenses and (iii) except for random selection for EEO audit or complaints and other matters pending at the FCC as to which no notice has been given to an Acquired Company, there is not pending, nor, to Seller's knowledge, threatened, any Action by or before the FCC to revoke, suspend, cancel, rescind or adversely modify the Sharing Agreement Station Licenses (other than proceedings to amend FCC rules of general applicability to commercial full-power stations and proceedings affecting those stations generally).

Schedule 3.4(e) contains (i) a list of all retransmission consent agreements or any other carriage agreement with multi-channel video programming distributors, including cable systems, telephone companies and direct broadcast satellite systems (together, "MVPDs"), with more than 10,000 subscribers within the Market of any Station, and (ii) a list of the MVPDs that, to Seller's knowledge, carry any Station and have more than 10,000 subscribers with respect to each such Station outside such Station's Market. Seller or an Acquired Company (i) has entered into a retransmission consent agreement or other carriage agreement with respect to each MVPD with more than 10,000 subscribers within the Market for any such Station and (ii) has made a timely election for retransmission consent, pursuant to FCC rules, for the current must-carry/retransmission consent election cycle for each Station with respect to each MVPD with more than 10,000 subscribers in such Station's Market. To Seller's knowledge, no market modification proceeding is pending at the FCC with respect to any Station. Since January 1, 2018, except as set forth on Schedule 3.4(e), (x) no headend with more than 10,000 subscribers covered by an MVPD in any Market of any Station has provided written notice to Seller or to any Acquired Company of any material signal quality issue or has failed to respond to a request for carriage or, to Seller's knowledge, has sought any form of relief from carriage of a Station from the FCC and (y) neither Seller nor any Acquired Company has received any written notice (1) from any MVPD with more than 10,000 subscribers in the Market of a Station of such MVPD's intention to delete such Station from carriage or to change such Station's channel position or (2) of a petition seeking FCC modification of any Station's Market.

3.5 Taxes. Except as set forth on Schedule 3.5:

- (a) All Income Tax and other material Tax Returns required to be filed by (i) the Acquired Companies and (ii) by Seller with respect to the Business, have been filed with the appropriate Taxing Authority, and all such Tax Returns are true, correct and complete in all material respects. Except for Taxes contested in good faith by appropriate proceedings and adequately reserved for on the Interim Balance Sheet, all material Taxes due and payable by (A) the Acquired Companies and (B) by Seller with respect to the Business, have been paid and will, to the extent due on or before the Closing Date, continue to be paid in full, including all installments or prepayments of Taxes due on or before the Closing Date. There are no Liens for Taxes upon any of the assets of the Acquired Companies, except for Permitted Liens. None of Seller (with respect to the Business) or the Acquired Companies currently is the beneficiary of any extension of time to file any Tax Return to a date later than the Closing Date.
- (b) There is no Action pending against (i) the Acquired Companies or (ii) Seller with respect to the Business, in each case, relating to Taxes or Tax Returns in respect of the Business or the assets of the Acquired Companies, and, to Seller's knowledge, no such proceeding has been threatened against (A) the Acquired Companies or (B) Seller with respect to the Business, by any Taxing Authority relating to Taxes or Tax Returns in respect of the Business or the assets of the Acquired Companies. Except for Taxes contested in good faith by appropriate proceedings and adequately reserved for on the Interim Balance Sheet, there is no Tax assessment or deficiency asserted against Seller with respect to the Business or any of the Acquired Companies that has not been paid. Neither any of the Acquired Companies nor Seller has waived any statute of limitations in respect of Taxes nor agreed to any extension of time with respect to a Tax assessment or deficiency, in each case (1) with respect to the assets of any of the Acquired Companies and (2) which has not expired.
- **(c)** Neither Seller nor any of the Acquired Companies is a party to or bound by any Tax sharing agreement or similar arrangement with any third party (other than this Agreement and any Ordinary Commercial Agreement).

- (d) Neither Seller with respect to the Business nor any Acquired Company has received any written claim that (i) has not been finally settled or otherwise finally resolved and (ii) was made by a Taxing Authority in a jurisdiction in which any of Seller or any of the Acquired Companies do not file Tax Returns, that Seller or any such Acquired Company is or may be subject to taxation by, or required to file any Tax Return in, that jurisdiction.
- **(e)** Each Acquired Company is and, at all times since the consummation of the Restructuring has been, disregarded as an entity separate from Seller for U.S. federal income tax purposes. Seller is and, since the consummation of the Restructuring has been, treated as a partnership for U.S. federal income tax purposes.
- (f) All material Taxes required to be withheld by (i) any of the Acquired Companies or (ii) Seller with respect to the Business, in connection with any amounts paid or owing to any employee, independent contractor, creditor, equity holder or other third party, have been withheld and either paid to the respective Taxing Authority or set aside in accounts for such purpose. For employment Tax purposes, Seller, with respect to the Business, and the Acquired Companies are in compliance in all material respects with applicable Laws with respect to classifying individuals performing services for the Business as employees or independent contractors.
 - (g) Seller is not a "foreign person" within the meaning of Section 1445 of the Code.
- **(h)** No Acquired Company has any liability for the Taxes of any Person (i) under Treasury regulation Section 1.1502-6 (or any analogous provision of state, local or foreign Tax Law), (ii) as a transferee or successor, or (iii) by Contract (other than any Ordinary Commercial Agreement).
- (i) Neither Seller nor any Acquired Company is or has been a party to any "listed transaction" as defined in Code Section 6707A(c)(1) and Treasury Regulation Section 1.6011-4(b)(2).
- (j) Neither Seller (with respect to the Business) nor any of the Acquired Companies has engaged in a trade or business, had a permanent establishment (within the meaning of an applicable Tax treaty or convention between the United States and a foreign country), or otherwise been subject to taxation in any country other than the country of its formation.
- **(k)** Neither Seller nor any of the Acquired Companies uses the cash receipts and disbursements method of accounting for federal or applicable state, local or foreign Tax purposes.
- (I) None of the Acquired Companies owns an interest, directly or indirectly, in any joint venture, partnership, limited liability company, association, or other entity that is treated as a partnership for U.S. federal, state or local Income Tax purposes
- (m) No Employee Plan provides for any payment by any of the Acquired Companies that would result in the payment of any compensation or other payments that would not be deductible under the terms of Section 280G of the Code after giving effect to the transactions contemplated hereby.
- (n) Each Employee Plan that is a "non-qualified deferred compensation plan" (as such term is defined in Section 409A(d)(1) of the Code), either has been maintained, in form and operation, in compliance in all material respects with the requirements of Section 409A of the Code and applicable guidance issued thereunder or is eligible to be excluded from the requirements of Section 409A of the Code by reason of the short term deferral rule of such provisions. Neither Seller nor any Acquired Company has any obligation to gross-up or indemnify any individual with respect to any Tax imposed under Section 409A(a)(1)(B) of the Code.

3.6 Tangible Personal Property.

- (a) Schedule 3.6(a) contains a list of all items of equipment, transmitters, antennas, cables, towers, vehicles, furniture, fixtures, spare parts and other tangible personal property of every kind and description with an original value in excess of \$50,000 owned, used or held for use by any of the Acquired Companies (together with all other personal property and other assets reflected in the Financial Statements or acquired after the Balance Sheet Date other than such other properties and assets sold or otherwise disposed of in the ordinary course of business consistent with past practice since the Balance Sheet Date and otherwise in compliance with this Agreement, the "Tangible Personal Property").
- **(b)** Except as set forth on Schedule 3.6(b), immediately prior to the Closing, the Acquired Companies will have good and valid title to, or a valid and enforceable leasehold interest in or license to, all of the Tangible Personal Property, in each case free and clear of all Liens (other than Permitted Liens).
- (c) Except as set forth on Schedule 3.6(c), all material items of Tangible Personal Property are in operating condition, ordinary wear and tear excepted, and have been maintained in accordance with standard industry practice. None of such material items of Tangible Personal Property is in need of maintenance or repairs except for ordinary, routine maintenance and repairs that are not material in nature or cost.
- (d) No Person other than an Acquired Company has any rights in any of the Tangible Personal Property, whether by ownership, lease, sublease, license or other otherwise, other than set forth on Schedule 3.6(d).

3.7 Real Property.

- (a) Schedule 3.7(a) contains a list of all real property (including any appurtenant easements, buildings, structures, fixtures and other improvements thereon) that is owned in fee simple by any of the Acquired Companies, indicating the owner and location thereof (collectively, the "Owned Real Property").
- **(b)** Schedule 3.7(b) contains a list of all Contracts (collectively, "*Real Property Leases*") pursuant to which any Acquired Company leases, licenses or sublicenses real property (including any appurtenant easements, buildings, structures, fixtures and other improvements thereon) (collectively, the "*Leased Real Property*" and, together with the Owned Real Property, the "*Real Property*") as lessee, sublessee, licensee or sublicensee, as applicable. The Acquired Company identified on Schedule 3.7(b) has a valid leasehold interest in the Leased Real Property indicated on Schedule 3.7(b) as being leased by such Acquired Company. Except as set forth on Schedule 3.7(b), neither Seller nor any Acquired Company has pledged, mortgaged or otherwise granted any Lien on its leasehold interest in any Leased Real Property. There are no oral agreements regarding the use or occupancy of any Real Property.
- (c) The Acquired Company identified on Schedule 3.7(a) has good and marketable fee simple title to the Owned Real Property indicated on Schedule 3.7(a) as being owned by such Acquired Company, in each case free and clear of Liens, other than Permitted Liens. Except as set forth on Schedule 3.7(c), no Acquired Company is obligated under, nor is a party to, any option, right of first refusal or other contractual right to purchase, acquire, sell, assign or dispose of any of the Owned Real Property or any portion thereof or interest therein.
- (d) With respect to the Real Property, there is no (i) pending or, to Seller's knowledge, threatened condemnation, eminent domain or taking proceeding or (ii) to Seller's knowledge, private restrictive covenant or governmental use restriction (including zoning) on all or any portion of the Real Property that prohibits or materially interferes with the current use of the Real Property in the Business.

- **(e)** The use and operation of the Real Property in the conduct of the Business do not violate in any material respect any Law or Material Contract. Except as set forth on Schedule 3.7(e), neither Seller nor any Acquired Company, within the past two (2) years, has received any written notice of any violation of any Law affecting any Real Property or any Acquired Company's use thereof.
- **(f)** Within the past two (2) years, neither Seller nor any Acquired Company has received any written notice of any existing plan or study by any Governmental Entity or by any other Person that challenges or otherwise adversely affects (or would be reasonably expected to adversely affect) the continuation of the use or operation of any Real Property and Seller has no knowledge of any such plan or study with respect to which it has not received written notice.
- (g) Except as set forth in the Revenue Leases listed on Schedule 3.7(g), (i) there is no Person in possession of any Owned Real Property other than an Acquired Company and (ii) no Acquired Company is a sublessor or grantor under any sublease or other instrument granting to any other Person any right to the possession, lease, occupancy, or enjoyment of any Leased Real Property.
- **(h)** Except as identified in Schedule 3.7(h), no Person has any right to acquire any interests in any of the Owned Real Property.
- (i) No material improvements constituting a part of the Real Property encroach on real property owned or leased by a Person other than any Acquired Company.
- (j) The Real Property is sufficient for the continued conduct of the Business after the Closing in substantially the same manner as conducted prior to the Closing and constitutes all of the real property necessary to conduct the Business as currently conducted.

3.8 Contracts.

- (a) Schedule 3.8(a) sets forth a true and complete list of the following Contracts to which any Acquired Company is a party, beneficiary or bound or that is otherwise applicable to the Business and to which Seller or any of its other Affiliates is a party or bound:
- (i) any Contract under which the aggregate payments or receipts (A) for the past twelve (12) months exceeded, or for the following twelve (12) months is reasonably expected to exceed, \$100,000 or (B) exceed \$200,000 over the remaining term of such Contract;
- (ii) any Contract under which payments by or obligations of any Acquired Company will be increased, accelerated or vested by the occurrence (whether alone or in conjunction with any other event) of any of the transactions contemplated by this Agreement, or under which the value of the payments by or obligations of any Acquired Company will be calculated on the basis of any of the transactions contemplated by this Agreement, whether pursuant to a change in control or otherwise;
- (iii) any Contract for Program Rights that involves aggregate payments or receipts in excess of \$100,000 over the remaining term of such Contract;
 - (iv) any network affiliation agreement;
- (v) any retransmission consent agreement with any MVPD with more than 10,000 subscribers in any Market of any Station;
- **(vi)** any Contract that relates to an ownership interest in any corporation, partnership, joint venture or other business enterprise or other entity or that otherwise involves a sharing of profits, losses, costs or liabilities with any other Person;
 - (vii) any Real Property Lease;

- **(viii)** any Contract that relates to the guarantee (whether absolute or contingent) by any Acquired Company of (x) the performance of any other Person or (y) the whole or any part of the Indebtedness or liabilities of any other Person;
- (ix) any Contract that contains any power of attorney authorizing the incurrence of an obligation on the part of any Acquired Company;
- (x) any Contract that creates any partnership or joint venture or relates to the acquisition, issuance or transfer of any securities;
- (xi) any Contract that limits or restricts (x) where or when any Acquired Company may conduct business, (y) the type or line of business (current or future) in which any Acquired Company may engage or (z) any acquisition of assets or stock or other equity interests (tangible or intangible) by any Acquired Company;
- (xii) any Contract that relates to (A) Indebtedness of any of the Acquired Companies or (B) the provision of any security, offset or other similar arrangement by or on behalf of any Acquired Company;
- (xiii) any Contract that grants any Person an option or a right of first refusal, right of first offer or similar preferential right to purchase or acquire any equity interest in, or assets of, any Acquired Company;
- (xiv) any Contract involving the purchase or sale of Real Property that has not closed as of the date hereof;
- (xv) any Contract that contains outstanding earn-out, deferred or contingent purchase price or similar contingent payment obligations;
- (xvi) any Contract entered into in the last three years (A) relating to the acquisition or disposition of any material portion of the Business (whether by merger, sale of stock or other equity interests, sale of assets or otherwise) or (B) that relates to the acquisition or disposition of any other business or the equity or assets of any other Person (whether by merger, sale of stock or other equity interests, sale of assets or otherwise);
- (xvii) any executory Contract involving construction, architecture, engineering or other agreements relating to construction projects, in each case that involve payments in excess of \$100,000, and any other executory Contract for capital expenditures in excess of \$100,000 for any single item and \$200,000 for any project consisting of multiple items;
- (xviii) any Contract involving compensation to any Station Employee, or independent contractor or consultant engaged to perform services for the Business, in each case in excess of \$150,000 per year; provided, however, that for purposes of this Section 3.8(a)(xviii), the term Contract shall not include at-will Contracts or Contracts that are terminable upon 30 days' notice, in either case, without penalty or additional payment;
 - (xix) any Contract with a Governmental Entity;
- (xx) any Contract relating to the use of a Station's digital bit stream other than in connection with broadcast television services;
- (xxi) any Contract that requires any Acquired Company to purchase its total requirements of any product or service from a third party or that contains "take or pay" provisions;

(xxii) other than this Agreement and any Ordinary Commercial Agreement, any Contract that provides for the assumption of any Tax or environmental liability by any Acquired Company of any Person;

(xxiii) any Contract between or among any Acquired Company, on the one hand, and Seller or any Affiliate of Seller (other than any Acquired Company), on the other hand;

(**xxiv**) any Contract that involves any resolution or settlement of any actual or threatened material Action since January 1, 2016;

(xxv) any collective bargaining agreement or Contract with any union, works council or labor organization; and

(xxvi) excluding Contracts for Program Rights, any license, sublicense, consent to use agreement, settlement, coexistence agreement, covenants not to sue, waiver, release, permission and other Contracts relating to Intellectual Property that is licensed to or by any of the Acquired Companies and material to the Business as currently conducted ("*IP Agreements*").

The Contracts required to be disclosed pursuant to this Section 3.8(a) are, collectively with the Revenue Leases, referred to herein as the "*Material Contracts*".

- (b) Each of the Material Contracts is in full force and effect and is binding and enforceable in accordance with its terms upon the Acquired Companies and, to Seller's knowledge, the other parties thereto, subject in each case to the Enforceability Exceptions. The Acquired Companies have performed their respective obligations under each of the Material Contracts in all material respects and are not in breach or default thereunder in any material respect, and to Seller's knowledge, no other party to any of the Material Contracts is in breach or default thereunder in any material respect. None of Seller or any of the Acquired Companies has provided or received any written notice of any intention to terminate any Material Contract. Complete and correct copies of each Material Contract (including all modifications, amendments, and supplements thereto and waivers thereunder) have been made available to Buyer.
- Environmental. Except as set forth on Schedule 3.9, (a) each Acquired Company is and has been in compliance in all material respects with all Environmental Laws applicable to the Stations, the Business or the Real Property, which compliance includes obtaining, maintaining and complying in all material respects with all Permits, licenses or other authorizations required by Environmental Law, (b) no Actions are pending or, to Seller's knowledge, threatened against Seller, any Acquired Company, the Stations or the Real Property alleging a violation of or liability under Environmental Laws or arising from or related to electromagnetic spectrum pollution or emissions generated by or originating from any of the Stations or otherwise related to the operation of the Business, (c) no conditions exist at the Stations or any Real Property that would reasonably be expected to result in the owner or operator of the Stations or the Real Property incurring material liability under Environmental Laws, (d) there has been no Release of Hazardous Materials at, from, to, on or under any Real Property in contravention of Environmental Law, and (e) during the last five (5) years from the date hereof, none of Seller or any of the Acquired Companies has received any written request for information, written notice of violation or other written communication from any Governmental Entity alleging a violation of or liability under any Environmental Law. There are no underground storage tanks at the Owned Real Property or, to Seller's knowledge, at the Leased Real Property. Seller has made available to Buyer copies of all environmental assessments, audits, investigations or other similar environmental reports relating to the Real Property or the operation of the Business that are in the possession or control of Seller or any of its Affiliates.

3.10 Intangible Property.

- (a) Schedule 3.10(a) contains a true and complete list or description of (i) all Intellectual Property that is owned by any of the Acquired Companies and registered or the subject of an application for registration with any Governmental Entity or authorized private registrar (collectively, the "Registered IP") and (ii) all other Intellectual Property that is owned by any of the Acquired Companies and material to the Business as currently conducted (collectively with the Registered IP, the "Owned Intangible Property"). All required filings and fees related to the Registered IP have been timely filed with and paid to the relevant Governmental Entities and authorized registrars, and all Registered IP is otherwise in good standing. For purposes of this Section 3.10, "Licensed IP" shall mean Intellectual Property licensed to any of the Acquired Companies and used in the operation of the Business; and "Intangible Property" shall mean the Owned Intangible Property and the Licensed IP.
- (b) Except as set forth on Schedule 3.10(b), (i) no Acquired Company's use of the Intangible Property infringes upon, misappropriates or otherwise violates any third party's Intellectual Property in any respect, (ii) to Seller's knowledge, none of the Owned Intangible Property is being infringed or misappropriated by any third party, (iii) no Owned Intangible Property or Acquired Company is the subject of any pending or, to Seller's knowledge, threatened Action claiming infringement, misappropriation or other violation of any third party's Intellectual Property, (iv) in the past two (2) years, neither Seller nor any Acquired Company has received any written claim asserting that its use of any Intangible Property is unauthorized or violates, misappropriates or infringes upon the Intellectual Property of any third party or challenging the ownership, use, validity or enforceability of any Owned Intangible Property and (v) the Acquired Companies are the sole and exclusive legal and beneficial owners of the Owned Intangible Property and have the valid and enforceable right to use the Licensed IP in the manner in which such Licensed IP is currently being used or as is necessary for the conduct of the Business as currently conducted, in each case free and clear of Liens other than Permitted Liens.
- (c) The computer hardware, servers, networks, platforms, peripherals, data communication lines, and other information technology equipment and related systems that are owned, used or held for use by any of the Acquired Companies ("Company Systems") are reasonably sufficient for the immediate needs of the Business. Except as set forth on Schedule 3.10(c), in the past eighteen (18) months, there has been no unauthorized access, use, intrusion, or breach of security, or material failure, breakdown, performance reduction or other adverse event affecting any Company Systems that has caused or would reasonably be expected to cause any: (i) substantial disruption of or interruption in or to the use of such Company Systems or the conduct of the Business; (ii) material loss, destruction, damage, or harm of or to any Acquired Company or its Stations, operations, personnel, property, or other assets; or (iii) material liability of any kind to any Acquired Company.

3.11 Employees; Labor Matters; Employee Benefit Plans.

Except as set forth on Schedule 3.11(a), each Acquired Company has complied in all material respects with all applicable labor and employment Laws, including those which relate to wages, hours, independent contractors, employee classifications, overtime compensation, collective bargaining, pay equity, retaliation, reasonable accommodation, disability rights or benefits, immigration, child labor, hiring, promotion and termination of employees, working conditions, meal and break periods, privacy, vacation, leaves of absence, rights of recall, paid sick leave and unemployment insurance, terms and conditions of employment, discrimination in employment and collective bargaining, equal opportunity. harassment, immigration, workers' compensation, unemployment compensation, occupational health and safety and the collection and payment of withholding. Except as set forth on Schedule 3.11(a), no Action is pending or, to Seller's knowledge threatened against Seller (with respect to the Business or any Station Employee) or any Acquired Company alleging any violation of or liability under any labor or employment Laws. Except as set forth on Schedule 3.11(a), since January 1, 2018, there has been no unfair labor practice complaint or charge against any Acquired Company pending or, to Seller's knowledge, threatened before the National Labor Relations Board, any state labor relations board or any court or tribunal, nor has any written complaint pertaining to any such charge or potential charge been delivered to Seller or any Acquired Company. Except as set forth on Schedule 3.11(a), there is no strike, dispute. request for representation, slowdown, picketing or stoppage pending or, to Seller's knowledge, threatened in respect of any Acquired Company nor has any such event occurred within the last 12

months. Except as set forth on Schedule 3.11(a), no Acquired Company is a party to any collective bargaining, union or similar agreement with respect to its respective Station Employees, and to Seller's knowledge, no union represents or claims to represent such Station Employees.

- (b) Schedule 3.11(b) sets forth a list, as of the date hereof, of all employees of the Acquired Companies (the "Station Employees"), including the current rate of compensation, employment status (i.e., active, disabled, on authorized leave), department, title, and whether full-time or part-time. Two weeks prior to the Closing Date, Seller shall deliver to Buyer an updated Schedule 3.11(b) containing a true and correct list of the foregoing information for the Station Employees hired following the date hereof. Other than corporate personnel of Seller providing tax, human resources, accounting, finance, IT, risk management and legal compliance services, no individuals employed by Seller or any of their Affiliates (other than the Acquired Companies) provide services to or on behalf of the Stations or the Business. For purposes of applicable labor and employment Laws, Seller, with respect to the Business, and the Acquired Companies are in compliance in all material respects with such Laws with respect to classifying individuals performing services as employees or independent contractors.
- (c) Schedule 3.11(c) contains a list setting forth each employee benefit plan, program or arrangement currently sponsored, maintained or contributed to by any Acquired Company or any ERISA Affiliate, or with respect to which an Acquired Company or any ERISA Affiliate has or may have any actual or contingent liability or obligation (including any such obligations under any terminated plan or arrangement), including employee benefit plans, as defined in Section 3(3) of ERISA, Multiemployer Plans, deferred compensation plans, stock option or other equity compensation plans, stock purchase plans, phantom stock plans, bonus plans, fringe benefit plans, life, health, dental, vision, hospitalization, disability and other insurance plans, employee assistance program, severance or termination pay plans and policies, and sick pay and vacation plans or arrangements, whether or not described in Section 3(3) of ERISA. Each and every such plan, program, agreement or arrangement shall constitute an "*Employee Plan*" hereunder. Except as set forth on <u>Schedule 3.11(c)</u>, Seller is the sponsor (within the meaning of Section 3(16)(B) of ERISA) of every such Employee Plan and no Acquired Company is a plan sponsor of any Employee Plan.
- (d) Except as set forth on Schedule 3.11(d), with respect to each Employee Plan: (i) each has been established and is being operated in compliance in all material respects with its terms and all applicable Laws, including ERISA and the Code; (ii) no Actions or disputes are pending, or to Seller's knowledge, threatened (other than routine claims for benefits); (iii) no audits, inquiries, reviews, proceedings, claims, or demands are pending with any Governmental Entity; (iv) to Seller's knowledge, there are no facts which could give rise to any liability in the event of any such investigation, claim, Action, audit, review, or other proceeding; (v) all premiums, contributions, or other payments required to have been made by Law or under the terms of any Employee Plan or any Contract or agreement relating thereto have been made; (vi) all material reports, returns and similar documents required to be filed with any Governmental Entity or distributed to any plan participant have been duly and timely filed or distributed; and (vii) to Seller's knowledge, no "prohibited transaction" has occurred within the meaning of the applicable provisions of ERISA or the Code.
- (e) Each Acquired Company, each ERISA Affiliate and each Employee Plan that is a "group health plan" as defined in Section 733(a)(1) of ERISA (each, a "Health Plan") (i) is currently in compliance in all material respects with the Patient Protection Affordable Care Act, Pub. L. No. 111-148 ("ACA"), the Health Care and Education Reconciliation Act of 2010, Pub. L. No. 111-152 ("HCERA"), and all regulations and guidance issued thereunder (collectively, with ACA and HCERA, the "Health Care Reform Laws") and (ii) has been in compliance in all material respects with all Health Care Reform Laws since March 23, 2010, in the case of each of clauses (i) and (ii), to the extent the Health Care Reform Laws are applicable thereto. No Acquired Company, no ERISA Affiliate and no Health Plan has incurred (and, to Seller's knowledge, nothing has occurred, and no condition or circumstance exists, that could reasonably be expected to subject any Acquired Company, any ERISA Affiliate or any Health Plan to) any penalty or excise Tax under Code Section 4980D, 4980H or 4980I or any other provision of the Health Care Reform Laws.

- (f) Seller and its ERISA Affiliates do not maintain any Employee Plan which is a "group health plan," as such term is defined in Section 5000(b)(1) of the Code, that has not been administered and operated in compliance in all material respects with the applicable requirements of Section 601 of ERISA, Section 4980B(b) of the Code and the applicable provisions of the Health Insurance Portability and Accountability Act of 1986. Seller and all Acquired Companies are not subject to any liability, including additional contributions, fines, penalties or loss of tax deduction as a result of such administration and operation.
- (g) Except as set forth on Schedule 3.11(g), neither the execution and delivery of this Agreement nor the consummation of the transactions contemplated hereby shall: (i) result in the acceleration of the time of payment or vesting or creation of any rights of any current or former employee, manager, director or consultant to compensation or benefits under any Employee Plan or otherwise that would be payable by any Acquired Company or Buyer; (ii) result in any payment becoming due (for severance or termination pay or otherwise), or increase the amount of any compensation due, in each case, from any Acquired Company or Buyer to any current or former employee, manager, director or consultant of the Acquired Companies; or (iii) increase any benefits otherwise payable under any Employee Plan.
- Except as set forth on Schedule 3.11(h), (i) none of the Acquired Companies (h) contributes to, has ever been required to contribute to, or has any liability or obligation to, any Multiemployer Plan and (ii) no Employee Plan is (w) subject to Section 412 of the Code or Title IV of ERISA, (x) is a "multiple employer plan" within the meaning of Section 210 of ERISA or Section 413(c) of the Code, (y) is a "multiple employer welfare arrangement" as such term is defined in Section 3(40) of ERISA, or (z) provides group health or death benefits following termination of employment, other than to the extent required by Part 6 of Subtitle B of Title I of ERISA or Section 4980B of the Code or by a comparable state Law ("COBRA"). With respect to any Multiemployer Plans set forth on Schedule 3.11(h): (A) all contributions required to be made with respect to employees of the Acquired Companies have been timely paid; (B) neither any of the Acquired Companies nor any ERISA Affiliate has withdrawn, partially withdrawn, or received any notice of any claim or demand for withdrawal liability or partial withdrawal liability against any of them; (C) no such plan is (or is reasonably expected to be) insolvent or in reorganization and no accumulated funding deficiency (as defined in Section 302 of ERISA and Section 412 of the Code), whether or not waived, exists or is reasonably expected to exist with respect to any such plan nor any such plan is or reasonably expected to be "at-risk" under Section 430 of the Code; and (D) neither any of the Acquired Companies nor any ERISA Affiliate has any actual or contingent liability under Section 4204 of ERISA.
- (i) Except as set forth in Schedule 3.11(i), neither Seller nor any Acquired Company has any current obligation under any Employee Plan to provide (whether now or in the future) postemployment (i) health, life or other welfare benefits other than required under Section 4980B of the Code or any similar applicable Law or (ii) severance, separation, termination, change in control, or non-qualified deferred compensation benefits or notice or pay in lieu thereof.
- (j) Each Health Plan that is required to be terminated effective as of Closing, as specified in Section 6.4(g), may be terminated as described in Section 6.4(g) without penalty or the incurrence of damages or claim for breach of contract.
- **3.12** Insurance. Schedule 3.12 lists (a) all insurance policies maintained by Seller or any of its Affiliates (including the Acquired Companies) covering any of the Acquired Companies (or any of their respective employees, officers, directors or managers), the Stations or the Business and (b) the coverages of such policies as of the date of this Agreement. All such policies are (and will remain until the Closing) in full force and effect. Except as listed on Schedule 3.12, there is no material claim pending under any such insurance policy relating to any Acquired Company or the Business, nor is there any claim thereunder as to which coverage has been questioned, denied or disputed by the underwriters of such insurance policy or in respect of which there is an outstanding reservation of rights. Neither Seller nor any of its Affiliates (including the Acquired Companies) has received any written notice of termination or cancellation of, premium increase with respect to, or alteration of coverage under, any of such

insurance policies. None of Seller or any of its Affiliates (including the Acquired Companies) is in default under, or has otherwise failed to comply with, in any material respect, any provision contained in any such insurance policy. Such insurance policies are sufficient for the Acquired Company's compliance in all material respects with all applicable Laws and Material Contracts that require an Acquired Company to maintain insurance. True and complete copies of such insurance policies, together with copies of all policy endorsements, have been made available to Buyer. Seller has made available to Buyer the claim loss history for the past three full calendar years for claims occurring under each such insurance policy.

- **3.13 Compliance with Law; Permits**. Apart from those matters of compliance with the Communications Laws, program distribution, and FCC Licenses that are addressed in Section 3.4,
- (a) except as set forth on Schedule 3.13(a), (i) each of the Acquired Companies has complied, and is complying, in all material respects with all applicable Laws and Governmental Orders and (ii) there are no Actions pending or, to Seller's knowledge, threatened against any Acquired Company with respect to any material violation of such Laws or Governmental Orders, except for those affecting the television broadcast industry as a whole; and
- **(b)** except as set forth on Schedule 3.13(b), (i) each of the Acquired Companies holds all material licenses, franchises, permits, certificates, approvals and authorizations from Governmental Entities necessary for the conduct of the Business as currently conducted (collectively but excluding the FCC Licenses, the "**Permits**"), (ii) all such Permits are valid and in effect in accordance with their respective terms, (iii) each of the Acquired Companies is in compliance in all material respects with the terms of all Permits, (iv) there is no Action pending or, to Seller's knowledge, threatened regarding the suspension, revocation, or cancellation of any Permits, and (v) all fees and charges with respect to such Permits have been paid in full.

3.14 Litigation; Orders.

- (a) Apart from those matters of compliance with the Communications Laws, program distribution, and FCC Licenses that are addressed in Section 3.4 and except as set forth on Schedule 3.14(a), there is no Action pending or, to Seller's knowledge, threatened (i) against or by any of the Acquired Companies affecting or relating to any of the Stations or any of its other properties or assets (or by or against Seller or any Affiliate thereof and relating to any Acquired Company, any Station or the Business) which would reasonably be expected to result in aggregate damages to all or any of the Acquired Companies in excess of \$100,000 or impede, prevent or materially delay Seller's ability to perform its obligations under this Agreement; or (ii) against or by any Acquired Company, Seller or any Affiliate of Seller that challenges or seeks to prevent, enjoin or otherwise delay the transactions contemplated by this Agreement. All Actions required to be set forth on Schedule 3.14(a) against any Acquired Company are subject to insurance coverage under the insurance policies of Seller or its Affiliates and, to Seller's knowledge, any Losses arising out of or resulting from such Actions are being, or will be, funded by insurance under such policies.
- **(b)** Apart from those matters of compliance with the Communications Laws, program distribution, and FCC Licenses that are addressed in Section 3.4 and except as set forth on Schedule 3.14(b), there is no outstanding Governmental Order against or affecting any Acquired Company or any of its properties or assets except for those affecting the television broadcast industry generally.
- **3.15 Financial Statements.** Schedule 3.15 sets forth copies of the following financial statements from Seller's internal reporting system relating solely to the Acquired Companies, the Stations and the Business (such financial statements, collectively, the "*Financial Statements*"): (a) the unaudited balance sheets and statements of operations as of and for the fiscal years ended September 30, 2017 and September 30, 2016 and (b) the unaudited balance sheet (the "*Interim Balance Sheet*") and statement of operations as of and for the eleven (11) months ended the Balance Sheet Date. The Financial Statements have been derived from the books and records of Seller and the Acquired Companies and fairly present, in all material respects, the financial position and results of operations of the Acquired Companies, the Stations and the Business as of the dates thereof and for the periods

indicated therein in conformity with GAAP (except insofar as such unaudited Financial Statements may omit footnotes and may be subject to normal and recurring year-end adjustments that are not, either individually or in the aggregate, material) and reflect the costs and expenses of conducting the Business, including Seller's good faith allocations for costs and expenses of services performed for the Acquired Companies by Seller and its Affiliates (other than the Acquired Companies). Each Acquired Company maintains a standard system of accounting established and administered in accordance with GAAP.

- **3.16 No Undisclosed Liabilities**. Except as set forth on Schedule 3.16, the Acquired Companies have no liabilities or obligations of any kind or nature, whether known or unknown, absolute or contingent, accrued or unaccrued which would be required to be disclosed on a balance sheet or in footnotes thereto, in each case prepared in accordance with GAAP, except for (a) liabilities which are reflected or reserved for in the Interim Balance Sheet, (b) current liabilities incurred in the ordinary course of business consistent with past practice since the Balance Sheet Date that are not, individually or in the aggregate, material in amount, or (c) liabilities contemplated by this Agreement.
- **3.17 Absence of Changes**. Since October 1, 2017, (a) there has not been any Effect that, individually or in the aggregate with any other Effect, has had or would reasonably be expected to have, a Material Adverse Effect, and (b) the Acquired Companies, the Stations and the Business have been operated in all material respects in the ordinary course of business consistent with past practice.
- **3.18 No Brokers**. Except for the services of Methuselah Advisors to Seller, for which all of the applicable fees, commissions and expenses shall be paid solely by Seller, no broker, investment banker, finder, financial advisor or other third party has been employed or retained by or on behalf of Seller or any Acquired Company in connection with the transactions contemplated by this Agreement or is or may be entitled to any broker's, finder's, financial advisor's or other similar fee or commission, or the reimbursement of expenses, in connection with the transactions contemplated by this Agreement based upon arrangements made by or on behalf of Seller or any of the Acquired Companies.
- **3.19 Related Party Transactions.** Except as set forth on Schedule 3.19, none of the Acquired Companies is currently a party to any Contract with Seller, any Affiliate of Seller or any member, manager, director, officer or employee of any Acquired Company (other than employment agreements with Station Employees and Employee Plans, in each case listed on the applicable Schedules hereto), Seller or any Affiliate of Seller (collectively, "**Related Party Transactions**"). Neither Seller nor any Affiliate of Seller (excluding the Acquired Companies), or any current or former member, manager, director, officer or employee of any Acquired Company, Seller or any Affiliate of Seller, (i) owns or leases any material asset or property used in or otherwise relating to the Business or (ii) has borrowed any monies from or has outstanding any Indebtedness or other similar obligation to any Acquired Company.
- **3.20** All Assets. Except as set forth on Schedule 3.20, Buyer, upon the Closing, will acquire, indirectly through its acquisition of the Acquired Companies at the Closing, all right, title and interest in and to all assets and properties (including all Real Property) used or held for use in the Business or necessary to conduct the Business as conducted prior to Closing, and such assets and properties will constitute all the assets and properties owned, leased or licensed by Seller, any of the Acquired Companies or any Affiliate of Seller and used or held for use in the Business or necessary to conduct the Business in all material respects as conducted prior to Closing.
- **3.21 Bank Accounts**. Schedule 3.21 sets forth the account number and names of authorized signatories with respect to each account maintained by or for the benefit of any Acquired Company at any bank or other financial institution.

ARTICLE 4

BUYER REPRESENTATIONS AND WARRANTIES

Buyer hereby makes the following representations and warranties to Seller as of the date hereof and as of the Closing:

- **4.1 Organization**. Buyer is duly organized, validly existing and in good standing under the laws of the jurisdiction of its organization. Buyer has the requisite corporate power and authority to execute, deliver and perform this Agreement and all of the other agreements and instruments to be executed and delivered by Buyer pursuant hereto (collectively, the "*Buyer Ancillary Agreements*") and to consummate the transactions contemplated hereby and thereby.
- **4.2 Authorization**. The execution, delivery and performance of this Agreement and the Buyer Ancillary Agreements by Buyer have been duly authorized and approved by all necessary corporate action of Buyer and its directors and stockholder and do not require any further authorization or consent of Buyer or its directors or stockholder. This Agreement is, and each Buyer Ancillary Agreement when executed and delivered by Buyer and the other parties thereto will be, valid and binding agreements of Buyer, enforceable in accordance with their terms, except in each case as such enforceability may be limited by the Enforceability Exceptions.
- **4.3 No Conflicts**. Except for the Governmental Consents, the execution, delivery and performance by Buyer of this Agreement and the Buyer Ancillary Agreements and the consummation by Buyer of the transactions contemplated hereby or thereby do not and will not, with or without notice or the passage of time (a) violate the Organizational Documents of Buyer, (b) violate in any material respect, or result in the creation of any Lien (other than any Permitted Lien) under, any Law or Governmental Order to which Buyer or its assets are subject, (c) result in a material breach of, or event of default or the creation of any Lien (other than any Permitted Lien) under, any material Contract to which Buyer is a party or to which its assets are subject or (d) require the consent or approval of, or a filing by Buyer with, any Governmental Entity.
- **4.4 Litigation**. There is no Action pending or, to Buyer's knowledge, threatened against Buyer which would reasonably be expected to adversely affect Buyer's ability to perform its obligations under this Agreement or the Buyer Ancillary Agreements or otherwise impede, prevent or materially delay the consummation of the transactions contemplated hereby or thereby.

4.5 Qualification.

- (a) With the exception of obtaining continuation/replacement of the existing Station KBZK (formerly KCTZ) satellite waiver, a copy of which is included in Schedule 3.4(a), Buyer is legally, financially and otherwise qualified to acquire the Equity Interests and to own the Acquired Companies and to control and operate the Stations under the Communications Laws, including the provisions relating to media ownership and attribution, foreign ownership and control and character qualifications, and there are no facts or circumstances relating to Buyer that would, under the Communications Laws and the existing procedures of the FCC, disqualify Buyer as the owner and operator of the Stations or as the transferee of control of the Acquired Companies and the FCC Licenses.
- (b) No waiver of or exemption from any provision of the Communications Laws and policies of the FCC is necessary in respect of Buyer for the FCC Consent to be obtained; and there are no facts or circumstances relating to Buyer that would reasonably be expected to (i) result in the FCC's refusal to grant the FCC Consent or otherwise disqualify Buyer, (ii) materially delay or impede obtaining the FCC Consent or (iii) cause the FCC to impose a material condition or conditions on its granting of the FCC Consent.
- **4.6 Projections and Other Information**. Buyer acknowledges that, with respect to any estimates, projections, forecasts, business plans, budget information and similar documentation or information relating to the Acquired Companies, the Stations, the Business and the transactions contemplated hereby that Buyer has received from Seller, any Acquired Company or any of their respective Affiliates or advisors and that is not documentation or information with respect to which Seller makes any representation, warranty, covenant or other agreement in this Agreement, (a) Buyer is not relying on such documentation in making its determination with respect to signing this Agreement or completing the transactions contemplated hereby, (b) there are uncertainties inherent in attempting to make such estimates, projections, forecasts, plans and budgets, (c) Buyer is taking full responsibility for

making its own evaluation of the adequacy and accuracy of all such estimates, projections, forecasts, plans and budgets so furnished to it and (d) except for a claim of fraud. Buyer will not assert any claim against Seller, the Acquired Companies, their respective Affiliates or any of its or any of their respective directors, officers, members, managers, employees, Affiliates or representatives, or hold Seller, the Acquired Companies or any such Persons liable, with respect thereto. Buyer acknowledges that none of Seller, the Acquired Companies nor any of their respective Affiliates, nor any other Person has made any representation or warranty, express or implied, as to the accuracy or completeness of any information regarding the Acquired Companies, the Stations, the Business or the transactions contemplated by this Agreement except for the representations and warranties of Seller set forth in this Agreement. Except in the case of fraud or a breach of any of the representations, warranties, covenants or other agreements of Seller set forth herein with respect to any such information, none of Seller, the Acquired Companies, any of their respective Affiliates nor any other Person will have or be subject to any liability to Buyer resulting from the distribution to Buyer or its representatives or Buyer's use of, any such information, including any confidential memoranda distributed on behalf of Seller or the Acquired Companies relating to the Acquired Companies or other publications or data room information provided to Buyer or its representatives, or any other document or information in any form provided to Buyer or its representatives in connection with the transactions contemplated hereby. Notwithstanding anything herein to the contrary, nothing in this Section 4.6 will in any way limit Buyer's rights (including under Section 8.1(a) and ARTICLE 10) with respect to the representations and warranties of Seller in ARTICLE 3.

- **4.7 Sufficient Funds**. Buyer currently has or has the ability to obtain, and will have as of the Closing Date, sufficient funds available to pay the Purchase Price in full.
- **4.8 No Brokers.** No broker, investment banker, financial advisor or other third party has been employed or retained by Buyer in connection with the transactions contemplated by this Agreement or is or may be entitled to any broker's, finder's, financial advisor's or other similar fee or commission, or the reimbursement of expenses, in connection with the transactions contemplated by this Agreement based upon arrangements made by or on behalf of Buyer.
- **4.9 Securities Laws.** Buyer is an "accredited investor" within the meaning of regulation D of the Securities Act of 1933, as amended (the "**Securities Act**"), with such knowledge and experience in financial and business matters as are necessary in order to evaluate the merits and risks in an investment in the Equity Interests. The Equity Interests to be acquired by Buyer pursuant to this Agreement shall be acquired for Buyer's own account and not with a view to, or intention of, distribution thereof in violation of the Securities Act or any applicable state securities Laws. Buyer is able to bear the economic risk of its investment in the Equity Interests for an indefinite period of time and acknowledges that the Equity Interests have not been registered under the Securities Act and, therefore, cannot be sold unless subsequently registered under the Securities Act or an exemption from such registration is available.

ARTICLE 5

CERTAIN COVENANTS

5.1 Seller's Covenants. Between the date hereof and the Closing, except (a) as contemplated in this Agreement, (b) as contemplated by Schedule 5.1, or (c) as required by applicable Law, unless Buyer otherwise consents in writing, which consent shall not be unreasonably withheld, conditioned or delayed, Seller shall cause the Acquired Companies to (i) operate the Business in the ordinary course of business consistent with past practice and in accordance with the Communications Laws, the FCC Licenses and all other applicable Laws, (ii) use commercially reasonable efforts to maintain and preserve intact the current organization, business and franchise of each Acquired Company and to preserve the rights, franchises, goodwill and relationships of its employees, customers, lenders, suppliers, regulators and others having business relationships with any Acquired Company, in each case, in the ordinary course of business consistent with past practice, and (iii) without limiting the generality of the foregoing:

- (a) not cause or permit, or agree or commit to cause or permit, by act or failure to act, any of the FCC Licenses to expire or to be revoked, suspended or adversely modified, or take or fail to take any action that would cause the FCC or any other Governmental Entity to institute proceedings for the suspension, revocation or adverse modification of any of the FCC Licenses other than the modification, replacement, or termination in the ordinary course of business of FCC Licenses that have become obsolete or unfit for use or which are no longer useful in the Business;
- (b) except for (i) the sale of advertising time on the Stations in the ordinary course of business consistent with past practice or (ii) for the purpose of disposing of assets that are immaterial in value (individually and in the aggregate) in the ordinary course of business consistent with past practice so long as such assets (A) have become obsolete or unfit for use or (B) are no longer useful in the Business, not sell, lease, license or dispose of or agree to sell, lease, license or dispose of any assets or properties (including the Real Property) of any of the Acquired Companies, the Stations or the Business;
- (c) not create, assume or permit to exist any Liens upon any of the assets or properties of any of the Acquired Companies, the Stations or the Business, except for Permitted Liens;
- (d) not adopt a plan of liquidation, dissolution, merger, consolidation, restructuring, recapitalization or reorganization;
- **(e)** maintain, repair and replace the Tangible Personal Property, including any Tangible Personal Property which has been damaged, destroyed or lost prior to Closing, and maintain, repair and replace the Real Property, including any Real Property which has been damaged, destroyed or lost prior to Closing, in each case ordinary wear and tear excepted; *provided, however* that Seller and the Acquired Companies shall have no obligation to maintain, repair or replace any assets that have become obsolete and which are no longer useful in the Business or necessary for the operation of the Business as currently operated;
- (f) not cause, permit or propose any amendments to the Organizational Documents of any of the Acquired Companies;
- (g) not hire or terminate the employment of any Station general manager (excluding any terminations for "cause" as determined by Seller); provided that Seller may, or may cause an Acquired Company to, replace any such general manager whose employment is so terminated on terms and conditions substantially comparable to such former general manager;
- (h) except as otherwise required by Law, not enter into or become bound by any new, or amend or terminate any existing, employment agreement with a Station Employee other than entry into employment agreements entered into in the ordinary course of business consistent with past practice that provide an annual base salary of less than \$150,000 and that do not provide for any severance, retention, change in control or similar payment that differs materially in amount, conditions or terms than the amounts, conditions and terms for such payments contained in existing employment agreements entered into in the ordinary course of business of any Acquired Companies;
- (i) except for any Company Transaction Costs for which neither Buyer nor any Acquired Company will have any liability or obligation at or after the Closing, not enter into or amend any retention, change in control or similar Contract with any Station Employee, independent contractor or consultant;
- (j) not (i) increase the compensation (including wages, salaries, bonuses or incentives) or benefits of any Station Employee (except for (A) increases in compensation in the ordinary course of business consistent with past practice that do not exceed more than 4% per annum and (B) performance and stay bonuses and other compensation to be paid pursuant to Section 2.2(b) as Company Transaction Costs in connection with the consummation of the transactions contemplated by this Agreement and for which neither Buyer nor any of the Acquired Companies shall have any liability or

obligation at or following the Closing), or (ii) modify any severance policy applicable to any Station Employee that would result in any increase in the amount of severance payable to any such Station Employee (or would expand the circumstances in which such severance is payable);

- **(k)** not voluntarily recognize, or collectively bargain without the participation by and approval of Buyer with, any labor unions as the collective bargaining representative of any Station Employee;
- (I) use commercially reasonable efforts to maintain each Station's MVPD carriage existing as of the date of this Agreement; provided, that, nothing in Section 5.1 shall be deemed to limit any Station's ability to withhold its consent to carriage by an MVPD as part of a retransmission consent negotiation if such Station deems, in its sole discretion, that it is commercially reasonable to withhold such consent:
- (m) not (i) enter into any Contract that would have been a Material Contract on the date of this Agreement unless such Contract (x) is entered into in the ordinary course of business consistent with past practice and (y) does not involve payments by any Acquired Company of greater than \$150,000 during any twelve (12) month period or \$200,000 over the term of such Contract, (ii) amend in any material respect or renew any Material Contract unless such amendment or renewal (x) is effected in the ordinary course of business consistent with past practice and (y) does not increase the amount of payments to be made by any Acquired Company during any twelve (12) month period by \$150,000 or more, (iii) waive or accelerate any material right under any Material Contract or (iv) terminate any Material Contract (excluding the expiration of any Material Contract in accordance with its terms);
- (n) not (i) change any accounting practices, period, policies, procedures or methods (except for any change required under GAAP or applicable law) or (ii) maintain its books and records in a manner other than in the ordinary course of business consistent with past practice;
- (o) not make any acquisition (including by merger or consolidation) of the capital stock or other equity of, or of a material portion of the assets of, any Person or business or division thereof;
- (p) maintain its qualifications to maintain the Station Licenses with respect to each Station and not take any action that will impair such Station Licenses or such qualifications;
- (q) except for any Company Transaction Costs for which neither Buyer nor any of the Acquired Companies shall have any liability or obligation at or following the Closing, not adopt, enter into or become bound by any new Employee Plan or Multiemployer Plan or amend, modify or terminate any existing Employee Plan or Multiemployer Plan;
- **(r)** keep in full force and effect the material insurance policies required to be set forth on Schedule 3.12 (or other insurance policies comparable in cost, amount and scope);
- (s) not (i) issue, sell, pledge, dispose of, grant, encumber, or authorize the issuance, sale, pledge, disposition, or grant, of any Equity Interests or other equity interests in any of the Acquired Companies or any securities convertible into or exchangeable for or entitling the holder thereof to purchase or receive any Equity Interests or other equity interests in any of the Acquired Companies, (ii) split, combine or reclassify any Equity Interests or other equity interests in the Acquired Companies, or (iii) issue or sell any additional interests of, or securities convertible into or exchangeable for, or options, warrants, calls, commitments or rights of any kind to acquire, any Equity Interests or other equity interests in the Acquired Companies; provided that nothing herein shall prohibit the Acquired Companies from making cash distributions or dividends to its equity holders in the ordinary course of business consistent with past practice and otherwise in compliance with this Agreement;

- (t) not make, change or rescind any Tax election, adopt or change any method of Tax accounting, adopt or change any period of Tax accounting, make a request for a Tax ruling from any Taxing Authority, amend any Tax Return, enter into any closing agreement, settle any Tax claim or assessment, surrender any right to claim a Tax refund, consent to any extension or waiver of the limitation period applicable to any Tax claim or assessment (other than pursuant to timely and valid extensions of time to file a Tax Return obtained in the ordinary course of business and consistent with applicable Law); provided, that Seller also shall not undertake any such actions with respect to the Business or the Acquired Companies, except (i) as contemplated in this Agreement, (ii) as contemplated by Schedule 5.1, or (iii) as required by applicable Law;
- (u) not (i) enter into or agree or commit to enter into any new Tradeout Agreement relating to a specific Station with a value in excess of \$100,000 individually, or \$200,000 in the aggregate, that will not be fully performed prior to the Closing or (ii) make any guarantee of commercial ratings other than in the ordinary course of business consistent with past practice;
- (v) not (i) incur, assume or guarantee any Indebtedness for borrowed money that will not be paid off pursuant to Section 2.2(a) or for which Buyer or any of the Acquired Companies shall have any liability or obligation at or following the Closing, (ii) assume, guarantee, endorse or otherwise as an accommodation become responsible for the obligations of any other Person or (iii) make any loan or material advance (excluding travel expenses advanced in the ordinary course of business consistent with past practice) to any Person;
- (w) not cancel or release any material Indebtedness owed to it by any Person or any material claims held by it with respect to any Person, except as required by Contracts in force on the date hereof and set forth in the Schedules;
- (x) not initiate any Action or settle or compromise any Action whether now pending or hereafter made or brought, except for any Action which is settled for cash payments of less than \$100,000, or waive, release or assign any material rights or claims;
- (y) not request or require the acceleration of the payment of any amounts owed to any Acquired Company, defer the payment of any material accounts payable, enter into any sale-leaseback contract with respect to any asset or revalue any assets of any Acquired Company (whether tangible or intangible), including writing off notes or accounts receivable, accelerate, settle, discount or compromise any accounts receivable or reverse any reserves with respect thereto;
- (z) not make or agree to make any capital expenditure or expenditures in excess of either \$50,000 individually or \$75,000 in the aggregate that is not contemplated by the budget made available to Buyer prior to the date hereof;
 - (aa) not enter into any contract with Seller or any Affiliate of Seller; and
- **(bb)** not agree, commit, authorize or resolve to take any actions inconsistent with the foregoing.
- **5.2 Inspection Rights**. From the date hereof until the Closing, Seller shall, and shall cause the Acquired Companies to, (a) upon reasonable advance notice, give Buyer and its representatives reasonable access to and the right to inspect, at reasonable times during normal business hours, the Stations and the properties, assets, premises, books and records, Contracts and other documents and data related to the Acquired Companies, the Stations or the Business and (b) furnish Buyer and its representatives with such financial, operating and other data and information relating to the Acquired Companies, the Stations or the Business that Buyer may reasonably request, provided, however, that such access rights shall not be exercised in a manner that interferes unreasonably with the normal business operations of the Acquired Companies or the Business; provided, further, that Seller and the Acquired Companies shall not be required to (i) violate any obligation of confidentiality or other obligation

under applicable Laws to which it is subject in discharging its obligations pursuant to this Section 5.2, (ii) take any action which, based on the advice of Seller's outside counsel, would constitute a waiver of attorney-client or other privilege or (iii) supply Buyer with any information which, in the reasonable judgment of Seller, any Acquired Company is under a contractual or legal obligation not to supply (provided that Seller shall, and shall cause the Acquired Companies to, use commercially reasonable efforts to seek a waiver from the applicable counterparty or beneficiary to permit the disclosure of such information). Any information disclosed to Buyer by Seller or Acquired Company under this Section 5.2 shall be held in accordance with the NDA (as defined below).

5.3 Control and Maintenance of Qualification.

- (a) Notwithstanding any other provision set forth in this Agreement, including any provision of this ARTICLE 5, Buyer shall not, directly or indirectly, control, supervise or direct the business or operations of the Acquired Companies or the Stations prior to the Closing. Consistent with the Communications Laws, prior to the Closing, control, supervision and direction of the Acquired Companies and the Stations prior to the Closing shall remain the responsibility of Seller as the corporate parent of the Acquired Companies holding the respective FCC Licenses.
- **(b)** Buyer shall remain legally, financially and otherwise qualified under the Communications Laws (including compliance with those Communications Laws relating to media ownership and attribution, foreign ownership and control requirements and character qualifications requirements) to be the transferee of control of the Acquired Companies and to become the licensee of the Stations as contemplated upon the Closing.
- Debt Financing Cooperation. Prior to the Closing, Seller shall use its commercially reasonable efforts to provide, and shall cause its Subsidiaries (including the Acquired Companies) and its and their respective representatives to use commercially reasonable efforts to provide, such cooperation as is reasonably requested by Buyer in connection with the arrangement of the Debt Financing (provided that such requested cooperation does not (1) require Seller or any of its Subsidiaries (including the Acquired Companies) to take any action that would violate any Laws or would result in a violation or breach by Seller or any of its Subsidiaries (including the Acquired Companies), or default under, any contract or agreement to which such Person is a party as of the date hereof or (2) result in any officer, manager or director of Seller or any of its Subsidiaries incurring any personal liability), including by using commercially reasonable efforts to: (i) to the extent reasonable, participate in a reasonable number of meetings (including meetings with prospective lenders), presentations, road shows, drafting sessions and due diligence sessions, including using commercially reasonable efforts to coordinate direct contact between senior management and the independent auditors of Seller and the Acquired Companies on the one hand, and the actual and potential lenders, on the other hand and sessions with rating agencies, at reasonable times and with reasonable advance notice, (ii) furnish to Buyer as promptly as reasonably practicable the Required Information, (iii) provide upon the reasonable request of Buyer such information reasonably deemed necessary to prepare a confidential information memorandum (including a version that does not include material non-public information) and other customary materials reasonably required to complete the syndication, (iv) assist Buyer in the preparation of (A) customary materials for rating agency and investor presentations (including "roadshow" or investor meeting slides), registration statements, offering memoranda, prospectuses, private placement memoranda, and other customary marketing materials and (B) definitive documentation for the Debt Financing, (v) cooperate to facilitate the due diligence efforts of the Debt Financing Sources relating to the Business, to the extent customary and reasonable and not unreasonably interfering with the business of Seller and its Subsidiaries. (vi) facilitate the release of any Liens on the Equity Interests and the termination of all guarantees (if any) in connection therewith subject to the occurrence of the Closing and (vii) provide at least five (5) Business Days prior to the Closing Date, all documentation and other information as is required by applicable "know your customer" and anti-money laundering rules and regulations including the USA PATRIOT Act and the beneficial ownership regulations. Notwithstanding anything in this Agreement to the contrary, (i) none of Seller or any of its Affiliates (other than the Acquired Companies at and following the Closing) shall be required to pay any commitment or other fee or incur any other liability or obligation in connection with the Debt Financing, (ii) no obligation of any Acquired Company under any document, certificate or

instrument executed pursuant to this Section 5.4 shall be effective until the Closing, and (iii) none of Seller or any of its Affiliates shall be required to execute or deliver or have any liability or obligation under any loan agreement or any related document or any other agreement or document (including any certificates, legal opinions or pledge or security documents) related to the Debt Financing. Buyer shall promptly, upon request by Seller, reimburse Seller for all reasonable and invoiced out-of-pocket costs (including reasonable attorneys' fees) incurred by Seller or any of its Affiliates in connection with the cooperation of Seller and its Affiliates contemplated by this Section 5.4. All non-public or other confidential information provided by Seller or its representatives pursuant to this Agreement will be kept confidential in accordance with the NDA, except that Buyer will be permitted to disclose such information to any Debt Financing Sources or prospective Debt Financing Sources and other financial institutions and investors that may become parties to the Debt Financing and to any underwriters, initial purchasers or placement agents in connection with the Debt Financing (and, in each case, to their respective counsel and auditors) so long as such Persons (i) agree to be bound by the NDA as if parties thereto or other confidentiality obligations that are substantially similar to those contained in the NDA or (ii) are subject to customary confidentiality arrangements, or other confidentiality undertakings reasonably satisfactory to Buyer and of which Buyer is a beneficiary.

ARTICLE 6

JOINT COVENANTS

Buyer and Seller hereby covenant and agree as follows:

6.1 Confidentiality.

- (a) Seller (or an Affiliate of Seller) and Buyer (or an Affiliate of Buyer) are parties to a nondisclosure agreement, dated July 30, 2018 (the "*NDA*"), with respect to Seller and the Acquired Companies. To the extent not already a direct party thereto, each of Seller and Buyer hereby assumes the NDA and agrees to be bound by the provisions thereof.
- From and after the Closing, Seller shall, and shall cause its representatives and Affiliates to: (i) treat and hold as confidential (and not disclose or provide access to any Person to) all information, whether written or oral, relating to the Acquired Companies, the Stations, the Business or Buyer and its Affiliates, including information in respect of regulatory compliance, programming, distribution, trade secrets, processes, product development, price, customer and supplier lists, pricing and marketing plans, policies and strategies, details of customer and consultant contracts, operations methods, product development techniques, business acquisition plans, and/or new personnel acquisition plans with respect to the Business (collectively, the "Information"), (ii) in the event that Seller or any such representative or Affiliate becomes legally compelled to disclose any such Information, provide Buyer with prompt written notice of such requirement so that Buyer or any Acquired Company may seek a protective order or other remedy or waive compliance with this Section 6.1(b), and (iii) in the event that such protective order or other remedy is not obtained, or Buyer waives compliance with this Section 6.1(b). furnish only that portion of such Information which is legally required to be provided and exercise its commercially reasonable efforts to obtain assurances that confidential treatment will be accorded such Information: provided, however, that this sentence shall not apply to any Information that, at the time of disclosure, is available publicly and was not disclosed in breach of this Agreement by Seller, its representatives or Affiliates. The term "Information" shall not include such portions of the Information which (A) are or become generally available to the public through no fault of Seller, any of its Affiliates or their respective advisors, representatives and agents, (B) become available to Seller after the Closing on a non-confidential basis from a source not prohibited from disclosing such information to Seller by a legal, contractual or fiduciary obligation to any Acquired Company or Affiliate thereof, (C) are independently acquired or developed after the Closing by Seller or one of its Affiliates without violating any obligations under this Agreement or (D) are used in the operation of any Other Seller Station or any other line of business other than the Business that is directly or indirectly owned by Seller that is not owned by an Acquired Company and that used such Information prior to the Closing for a legitimate business purpose and continues to use such Information after the Closing for a legitimate business purposes.

- (c) Prior to the Closing, (i) Seller shall assign to the Acquired Companies all confidentiality agreements entered into in connection with the contemplated sale of any of the Acquired Companies to which Seller or any of its Affiliates (other than the Acquired Companies) is a party that solely and exclusively involves a third party who sought Information with regard to the Acquired Companies or the Stations and (ii) with respect to those confidentiality agreements entered into in connection with the contemplated sale of any of the Acquired Companies to which Seller or any of its Affiliates (other than the Acquired Companies) is a party that involve a third party who sought Information with regard to the Acquired Companies or any of the Stations and any Other Seller Station, Seller shall assign to the Acquired Companies all rights thereunder to the extent relating to the Acquired Companies or the Stations.
 - (d) The NDA shall be deemed terminated effective upon the Closing.
- **6.2 Announcements**. No party shall (nor shall such party permit any Affiliate thereof to), without the prior written consent of the other (which consent shall not be unreasonably withheld, conditioned or delayed), issue any press release or make any other public announcement concerning the transactions contemplated by this Agreement, except to the extent that such party or any Affiliate thereof is so obligated by Law or any rule or regulation of any securities exchange upon which the securities of such party or any Affiliate thereof are listed or traded, in which case such party shall give advance notice to the other.
- **6.3 Consents.** Seller shall, and shall cause the Acquired Companies to, use commercially reasonable efforts to obtain (a) any third party consents required under any Material Contracts in connection with the consummation of the transactions contemplated by this Agreement (which shall not require any payment to any such third party other than in respect of normal and usual processing fees or other similar costs imposed by a third party in connection with the granting of a consent) and (b) customary estoppel certificates reasonably acceptable to Buyer under the Real Property Leases listed on Schedule 6.3(b), which shall include each Real Property Lease for a Station's studio or Tower Lease for a Station's primary tower or other material tower; *provided, however*, that the parties acknowledge and agree that such third party consents or estoppel certificates are not conditions to Closing, except for those certain third party consents set forth on Schedule 6.3(a) (the "*Required Consents*").

6.4 Employees; Employee Plans.

As of the Closing, Buyer or any of its Affiliates shall, or Buyer shall cause the applicable Acquired Company to continue to, employ each Station Employee who (i) is not then on authorized leave of absence, sick leave, short or long term disability leave, military leave or layoff with recall rights ("Active Employees") or (ii) is then on authorized leave of absence, sick leave, short-term disability leave, military leave or layoff with recall rights and who returns to active employment immediately following such absence and within six (6) months of the Closing Date, or such later date as required under applicable Laws ("Inactive Employees"). For the purposes hereof, all Active Employees and Inactive Employees who continue to be employed on the applicable Employment Commencement Date are hereinafter referred to collectively as the "Continuing Employees," and the "Employment Commencement Date" as referred to herein shall mean (x) as to those Continuing Employees who are Active Employees, the Closing Date, and (y) those Continuing Employees who are Inactive Employees, the date on which the Continuing Employee returns to active employment with the applicable Acquired Company. Following the Closing, the employment of the Continuing Employees who do not have any written Contract with any Acquired Company pursuant to which the Acquired Company has an actual or contingent liability to provide compensation and/or benefits in consideration for services ("Employment Agreements") shall be on at-will basis and Buyer shall cause to be provided initially to such Continuing Employees monetary compensation (consisting of base salary, and, as applicable, commission rate and annual bonus opportunity), benefits and severance benefits substantially the same as similarly situated employees of Buyer. The initial terms and conditions of employment for those Continuing Employees who have Employment Agreements shall be as set forth in such Employment Agreements. To the extent permitted by Law and notwithstanding anything herein to the contrary, Buyer shall give Continuing Employees full credit for all service of the Continuing Employee with the Acquired Companies for

purposes of eligibility waiting periods, vesting and benefit accrual (other than benefit accrual under a defined benefit pension plan) under all employee benefit plans maintained by Buyer or any of its Affiliates ("Buyer Benefit Plans") in which such Continuing Employees participate.

- **(b)** With respect to the Continuing Employees, Buyer and its Affiliates further agree to waive deductible and out-of-pocket requirements under the Buyer Benefit Plans that provide group health benefits and otherwise to give credit under the applicable Buyer Benefit Plans for amounts paid under a corresponding Employee Plan that provides group health benefits, as though such amounts had been paid in accordance with the terms and conditions of the applicable Buyer Benefit Plans.
- (c) Notwithstanding anything to the contrary in this Section 6.4, the parties expressly acknowledge and agree that (i) this Agreement is not intended to create a contract between Buyer, Seller or any of their respective Affiliates on the one hand and any Station Employee on the other hand, and no Station Employee may rely on this Agreement as the basis for any breach of contract claim against Buyer, Seller or any Acquired Company, (ii) nothing in this Agreement shall be deemed or construed to require Buyer or any Acquired Company to employ any particular Station Employee for any period after the Closing, and (iii) except for any applicable restrictions set forth in Section 5.1 for the period prior to the Closing Date, nothing in this Agreement shall be deemed or construed to limit any Acquired Company's right to terminate the employment of any Station Employee during any period prior to the Closing Date or the right of Buyer or any Acquired Company to terminate the employment of any Continuing Employee during any period after the Closing Date.
- (d) In the event that Buyer terminates the employment of any Continuing Employee on or after the Closing, Buyer shall be solely responsible to pay such Continuing Employee any severance or other compensation due, whether by Contract or law, on account of such termination; provided, however, that this Section 6.4(d) shall not limit in any respect any of the representations, warranties, covenants or indemnities of Seller contained in this Agreement.
- Effective as of the Closing Date or the payroll period ending immediately after the Closing Date, Seller or the Acquired Companies shall have contributed to the existing tax-qualified defined contribution plan established or designated by Seller ("Seller's 401(k) Plan") all matching or other employer contributions, if any, with respect to the Continuing Employees' employment service rendered prior to the Closing Date (irrespective of any end-of-year service requirements otherwise applicable to such contributions) and, to the extent required by Law, cause the matching and other employer contribution amounts of all Continuing Employees under the Seller's 401(k) Plan to become fully vested as of the Closing Date. Prior to and following the Closing Date, Seller shall take all actions necessary or appropriate to ensure that under the terms of the Seller's 401(k) Plan, each Continuing Employee with an account balance is eligible to receive a distribution as a result of the Closing. From and after the Closing, Buyer shall permit the Continuing Employees who participate in Seller's 401(k) Plan to elect to make direct rollovers of their account balances (including any outstanding loan balances) into a tax-qualified defined contribution plan established or designated by Buyer or any of its Affiliates ("Buyer's 401(k) Plan") as of the Closing (or as soon as practicable thereafter when Buyer's 401(k) Plan is capable of accepting such rollovers), subject to compliance with applicable Law and subject to the reasonable requirements of Buyer's 401(k) Plan. Such rollovers will be permitted by Buyer only if they consist of the total account balance of the Continuing Employee, Further, only rollovers of cash and outstanding loans may be transferred to Buyer's 401(k) Plan; and provided, further, that with respect to any loan balance transferred, subject to applicable Law and any required consent of individuals, any such loans may be reamortized to reflect any differences in payroll periods.
- (f) Seller shall retain responsibility for and continue to pay all medical, life insurance, disability and other welfare plan expenses and benefits for each Continuing Employee with respect to claims incurred under the terms of the Employee Plans by such Continuing Employees and their covered dependents prior to the Employment Commencement Date. Expenses and benefits with respect to claims incurred by Continuing Employees and their covered dependents on or after the Employment Commencement Date shall be the responsibility of Buyer and its Affiliates, including any claims subject to reimbursement under any Health Reimbursement Accounts. With respect to any welfare benefit plans

maintained by Buyer or any of its Affiliates in which the Continuing Employees are eligible to participate on or after the Employment Commencement Date, to the extent permitted by Laws, Buyer shall, and shall cause its Affiliates to (i) provide Continuing Employees with coverage immediately on the Employment Commencement Date, (ii) cause there to be waived any eligibility requirements or pre-existing condition limitations and (iii) give effect, in determining any deductible, co-insurance and maximum out-of-pocket limitations, to amounts paid by such Continuing Employees (and their covered dependents) under the Buyer Benefit Plans.

- **(g)** Effective as of Closing, Seller shall cause WLEX Communications, LLC to terminate its Health Plan, effective as of the Closing, and Seller shall cause sufficient advance notice to be given to the insurance carrier(s) such that termination may occur without penalty or claim of breach.
- (h) Buyer shall, or shall cause the applicable Acquired Company to, continue to recognize as of the Closing all liabilities for unpaid, accrued vacation or PTO of each Continuing Employee as of the Employment Commencement Date, giving service credit under the PTO policy of Buyer for service with the Acquired Companies, and shall permit Continuing Employees to use their accrued, but unused vacation/PTO entitlement as of Closing in accordance with Buyer's PTO policy. Three (3) days prior to the Closing Date, Seller shall deliver Schedule 6.4(h) to Buyer, which shall set forth the liabilities of the Acquired Companies for unpaid, accrued vacation or PTO relating to the Continuing Employees projected as of the Closing Date. Buyer shall receive a credit in the calculation of the Net Working Capital Amount pursuant to Section 2.3 for the amount of the liabilities for unpaid, accrued vacation or PTO relating to the Continuing Employees as of the Closing Date.
- 4.5 Access to and Retention of Records. Subject to Section 11.5, for a period of five (5) years following the Closing Date, (a) Buyer shall, or shall cause the Acquired Companies to, preserve, in accordance with Buyer's normal document retention procedures and practices, all books and records of the Business in the possession of the Acquired Companies that relate to periods prior to the Closing and (b) Buyer shall afford to Seller, and its counsel, accountants, and other authorized agents and representatives, at Seller's sole cost and expense, at reasonable times during normal business hours and upon reasonable advance notice, reasonable access to such books and records, and the right to make copies and extracts therefrom, to the extent that such access may be reasonably required by Seller (i) to facilitate the investigation, litigation and final disposition of any claims which have been made in writing against Seller by any third-party (not including Buyer or any of its Affiliates) or (ii) for the preparation of Tax Returns and audits, provided in each case that such access does not unreasonably disrupt or interfere with the Business or the business and operations of Buyer or the Acquired Companies. Seller agrees that any information or material accessed, copied, extracted or possessed by Seller under this Section 6.5 shall be Information covered by Section 6.1(b).

6.6 Cooperation; Financial Statements.

- (a) Buyer shall use commercially reasonable efforts to cooperate with Seller to release any Liens applicable to the Acquired Companies, *provided* that Buyer shall not be required to incur any cost, expense or other liability in connection therewith. Buyer acknowledges that Seller may use a portion of the proceeds from the Purchase Price for the repayment of Indebtedness associated with any such Liens pursuant to Section 2.2(a).
- **(b)** Without limiting Section 5.4, Seller shall, and shall cause the Acquired Companies to, provide, on a timely basis and at Buyer's sole cost and expense, such cooperation and assistance as Buyer may reasonably request in connection with (i) the arrangement of any financing for the transactions contemplated by this Agreement and (ii) the preparation of financial statements for the Acquired Companies or the Business for any fiscal period.
- **(c)** Prior to the Closing, but after January 15, 2019, Seller shall have delivered to Buyer copies of the audited consolidated balance sheets and statements of operations for EPI Preferred, LLC, Seller and the Subsidiaries as of and for the fiscal years ended September 30, 2018, September 30, 2017 and September 30, 2016 (the "*Audited Financial Statements*"). Seller hereby consents to the

inclusion of the Audited Financial Statements and related information, as applicable, following the Closing in (i) the Current Report on Form 8-K required to be filed by Buyer's parent, The E.W. Scripps Company ("EWS"), with the Securities and Exchange Commission (the "SEC") under the Securities Exchange Act of 1934, as amended (the "Exchange Act"), in connection with the transactions contemplated by this Agreement, (ii) any registration statement filed by EWS in connection with an offering or exchange of securities on Form S-1, Form S-4 or Form S-8 (or any successor forms) under the Securities Act in compliance with the requirements of Regulation S-X and Regulation S-K, as applicable, and (iii) any prospectuses, private placement memoranda, lender and investor presentations, offering documents, bank information memoranda, rating agency presentations and similar documents customarily used in connection with any debt or equity financing of Buyer or EWS. Seller shall use commercially reasonable efforts to cause its independent auditors, at such time or times as Buyer shall reasonably request, to provide signed consents to the inclusion of unqualified auditors' reports in the aforesaid registration statements and to references therein to such auditors as experts, which such consents may be filed with the SEC in connection with such registration statements.

- (d) Following the Closing, Seller shall reasonably cooperate with Buyer to permit Buyer and its Affiliates to prepare such unaudited pro forma financial statements for the Business for such time periods as required by the Exchange Act, the rules and regulations of the SEC or any rule or regulation of any securities exchange upon which the securities of EWS are listed or traded. Without limiting the generality of the foregoing, Seller shall, at Buyer's sole cost and expense, (i) provide Buyer and its accountants with reasonable access during normal business hours to financial and other information reasonably requested by Buyer in connection with the preparation of such financial statements, including access to work papers of Seller and its accountants reasonably requested by Buyer in connection therewith and (ii) provide reasonable assistance to Buyer and its accountants in the preparation of such financial statements.
- (e) Buyer shall, within 10 days of request by Seller, reimburse Seller for all reasonable and documented out-of-pocket costs and expenses incurred by Seller for any action taken by it or its accountants pursuant to Section 6.6(c) or Section 6.6(d); provided, however, that Buyer shall have no obligation to reimburse Seller for any costs or expenses incurred by Seller in connection with the audit for Seller's fiscal year ended September 30, 2018 or the preparation and audit of financial statements for the fiscal year ended September 30, 2018 except to the extent that Seller's independent accountant performing such audit charges Seller additional costs or fees to perform additional audit procedures solely in connection with such audited financial statements being included any document referred in Section 6.6(c)(i), (ii) or (iii).
- 6.7 Interim Reports. Within forty-five (45) days after the end of each calendar month during the period from the Balance Sheet Date through the Closing, Seller shall provide to Buyer, with respect to the Acquired Companies and the Stations, the unaudited balance sheet as of the end of such month and the related combined unaudited statement of operations for such month ended of the Acquired Companies and the Stations. Such reports shall be prepared on the same basis as the Financial Statements. Seller shall also provide to Buyer weekly pacing reports for each of the Stations promptly following the end of each week during the period from the date hereof through the Closing.
- **6.8** Fulfillment of Conditions. Without limiting any other obligation of a party expressly set forth herein, Seller shall use its commercially reasonable efforts to satisfy each of the conditions to the Closing of Buyer set forth in ARTICLE 8 (Buyer Closing Conditions), and Buyer shall use its commercially reasonable efforts to satisfy each of the conditions to the Closing of Seller set forth in ARTICLE 7 (Seller Closing Conditions), and each of the parties shall use its commercially reasonable efforts to take or cause to be taken all action necessary or desirable in order to consummate the transactions contemplated by this Agreement as promptly as practicable.
- **6.9 Title Commitments; Surveys**. Buyer shall have the responsibility to obtain, if it so elects at its sole option and expense, (a) commitments for owner's and lender's title insurance policies on the Owned Real Property and commitments for lessee's and lender's title insurance policies for all Real Property that is leased pursuant to a Real Property Lease (collectively, the "*Title Commitments*"), and

- (b) an ALTA survey on each parcel of Real Property (the "Surveys"); provided, however, that Seller shall provide Buyer with any existing Title Commitments, title policies and Surveys in its possession or control. The Title Commitments will evidence a commitment to issue an ALTA title insurance policy insuring good, marketable and indefeasible fee simple (or leasehold, if applicable) title to each parcel of the Real Property contemplated above for such amount as Buyer directs. Seller shall, and shall cause the Acquired Companies to, reasonably cooperate with Buyer in obtaining such Title Commitments and Surveys, provided that neither Seller nor any Acquired Company shall be required to incur any cost, expense or other liability in connection therewith. If the Title Commitments or Surveys reveal any Lien on the title or Real Property other than Permitted Liens, Buyer shall notify Seller in writing of such objectionable matter promptly after Buyer becomes aware that such matter is not a Permitted Lien, and Seller agrees to, and shall cause the Acquired Companies to, use commercially reasonable efforts to remove such objectionable matter.
- No Negotiation. Until the earlier of the Closing or such time as this Agreement shall be terminated pursuant to Section 12.1, Seller shall not, and shall cause its Affiliates (including the Acquired Companies) and its and their respective managers, directors, officers, investment bankers and agents not to, directly or indirectly, (a) solicit, initiate, encourage, entertain, consider or accept any inquiries, proposals or offers from any Person (other than Buyer) or (b) discuss, negotiate or communicate with, provide any information to, or enter into any agreements or other instruments with, any Person (other than Buyer) in each case relating to (i) any acquisition or purchase of the Equity Interests (or other securities or ownership interests of any Acquired Company), (ii) any merger, consolidation, business combination, amalgamation, recapitalization, reorganization or similar transaction involving any Acquired Company, (iii) any sale, lease, license, exchange or other disposition of all or a signification portion of the assets of any Acquired Company, the Business or any of the Stations, or (iv) other similar transaction involving any Acquired Company, the Business or any Station. Seller immediately shall cease and cause to be terminated all existing discussions, conversations, negotiations and other communications with any Person conducted heretofore with respect to any of the foregoing. Seller shall notify Buyer promptly if any such proposal or offer, or any inquiry or other contact with any Person with respect thereto, is made and shall, in any such notice to Buyer, indicate in reasonable detail the identity of the Person making such proposal, offer, inquiry or contact and the terms and conditions of such proposal, offer, inquiry or other contact. Seller agrees not to, and to cause the Acquired Companies not to, without the prior written consent of Buyer, release any Person from, or waive any provision of, any confidentiality or standstill agreement to which Seller (solely to the extent related to the Business) or any Acquired Company is a party.

6.11 Officers and Directors Indemnification; Insurance Tail Coverage.

- (a) Except as otherwise required by applicable Law, the indemnification agreements set forth on Schedule 6.11(a) shall remain in effect from and after the Closing for a period of six (6) years but only to the extent related to actions or omissions occurring prior to the Closing Date. Buyer shall cause, for a period of at least six (6) years from Closing, the Organizational Documents of the Acquired Companies as of the Closing to contain provisions no less favorable with respect to exculpation and indemnification of managers, directors and officers from and against liabilities arising prior to the Closing than are set forth in the Organizational Documents of the Companies as of the date hereof, except as required by any applicable Law.
- (b) EPI Group, LLC, a South Carolina limited liability company and parent of Seller ("EPI Group"), shall maintain its existing directors and officers, employment practices and fiduciary liability insurance policies for six (6) years from the Closing Date, and its existing cybersecurity, privacy and media liability insurance policies for three (3) years from the Closing Date, in each case, that name the Acquired Companies as named insureds thereunder with respect to the periods prior to the Closing, in each case without reducing the scope thereof, including retention and limits of coverage. For the sake of clarification, the intent of this Section 6.11(b) is to keep in effect the aforesaid policies, covering acts or omissions occurring prior to the Closing Date, for the benefit of the Acquired Companies and those directors, officers and managers of the Acquired Companies who are currently covered by such policies, as applicable.

41.

- The provisions of Section 6.11(a) and Section 6.11(b) are (i) intended to be for the benefit of, and shall be enforceable by, each individual who on or prior to the Closing Date was a director, manager or officer of the Acquired Companies (each, an "Indemnitee"), his or her heirs and his or her representatives, it being expressly agreed that such Persons shall be third party beneficiaries of this Section 6.11 and (ii) in addition to, not in substitution for, any other right to indemnification or contribution that any such Indemnitee may have under contract or otherwise. Following the Closing, in the event that Buyer or any of its Affiliates (including the Acquired Companies) shall enter into any merger, consolidation or similar transaction, Buyer shall make proper provision so that the surviving or resulting entity will assume the obligations imposed by Section 6.11(a). Following the Closing, in the event that EPI Group shall enter into any merger, consolidation, sale of substantially all of the assets then owned by it or any similar transaction, EPI Group shall purchase a "tail" insurance policy with coverage for (i) its existing directors and officers, employment practices and fiduciary liability insurance policies for the period remaining from the closing of any such transaction and six (6) years from the Closing Date and (ii) its existing cybersecurity, privacy and media liability insurance policies for the period remaining from the closing of any such transaction and three (3) years from the Closing Date, in each case that name the Acquired Companies as named insureds thereunder with respect to the periods prior to the Closing.
- (d) The obligations of Buyer and EPI Group under this Section 6.11 shall not be terminated or modified in such a manner as to adversely affect any Indemnitee to whom this Section 6.11 applies without the consent of the affected Indemnitee (it being expressly agreed that the Indemnitees to whom this Section 6.11 applies shall be third party beneficiaries of this Section 6.11, and each Indemnitee may specifically enforce the terms of this Section 6.11).
- **(e)** EPI Group's insurance policies written on an "occurrence basis" and insuring the Acquired Companies shall provide insurance coverage to the extent the events giving rise to any claim occurred prior to 11:59 p.m. on the Closing Date. EPI Group shall cooperate with Buyer in asserting claims under such policies in connection with insurable events and by executing appropriate assignments as permitted by Law. EPI Group shall remit any recoveries thereunder promptly to Buyer; provided, that any such remitted recoveries shall reduce Buyer's Losses with respect to any claim related to such occurrence.
- 6.12 Termination of Related Party Transactions. Seller shall, and shall cause the Acquired Companies to, terminate, without liability to the Acquired Companies or Buyer, all Related Party Transactions prior to the Closing and all such Related Party Transactions shall be deemed to be terminated in all respects (notwithstanding anything therein relating to the survival of any provisions thereof) effective as of the Closing without any further action by the parties thereto or the parties hereto and without any adverse Tax consequences to the Acquired Companies. To the extent that any Acquired Company is a party to any guarantee, letter of credit, surety bond or similar instrument, each such arrangement shall be released prior to Closing, in form and substance reasonably satisfactory to Buyer.
- 6.13 Multi-Station Contracts. Schedule 6.13 contains a list as of the date hereof of Contracts to which one or more television stations of Seller, Seller or an Affiliate of Seller (other than the Stations) (each, an "Other Seller Station") is party to, or has rights or obligations thereunder (any such Contract, a "Multi-Station Contract"). The rights and obligations under the Multi-Station Contracts to be allocated to the Acquired Companies shall include only those rights and obligations under such Multi-Station Contracts that are applicable to the Stations. The rights of each Other Seller Station with respect to such Multi-Station Contracts and the obligations of each Other Seller Station to such Multi-Station Contracts shall not be allocated to the Acquired Companies. For purposes of determining the scope of the rights and obligations of the Multi-Station Contracts, the rights and obligations under each Multi-Station Contract shall be equitably allocated among (1) the Stations, on the one hand, and (2) the Other Seller Stations, on the other hand, in accordance with the following equitable allocation principles:
 - (a) any allocation set forth in the Multi-Station Contract shall control; or
- **(b)** if there is no allocation as described in clause (a) hereof, then reasonable accommodation (to be determined by mutual good faith agreement of Seller and Buyer) shall control.

Subject to any applicable third-party consents, such allocation and assignment with respect to any Multi-Station Contract shall be effectuated, as mutually agreed by Seller and Buyer, by termination of such Multi-Station Contract in its entirety with respect to the Stations and the execution of new contracts with respect to the Stations or by an assignment to and assumption by the applicable Acquired Company of the related rights and obligations under such Multi-Station Contract. The parties shall use commercially reasonable efforts to obtain any such new contracts or assignments to, and assumptions by, the Acquired Companies in accordance with this Section 6.13; provided, that, completion of documentation of any such allocation under this Section 6.13 is not a condition to Closing.

- **6.14 Non-Solicitation**. For a period of two (2) years from the Closing Date, none of Seller nor its Affiliates (excluding the Acquired Companies) shall, directly or indirectly, solicit for employment or hire (whether as an employee, consultant or otherwise) any employee of Buyer or its Affiliates or any Acquired Company immediately prior to Closing; provided that Seller and its Affiliates shall not be precluded from soliciting or hiring, or taking any other action with respect to any such person who responds to general or public solicitation not targeted at employees of Buyer or any of its Affiliates, including the Acquired Companies.
- **6.15 Mail and Other Communications**. After the Closing, Seller shall, and shall cause its Affiliates to, promptly remit to Buyer any checks, cash, payments, material mail or other material communications directed to any Acquired Company, the Business or any properties or assets of any Acquired Company that are received by Seller or any Affiliate thereof after the Closing Date.
- **6.16 Joint Privilege**. Seller and Buyer acknowledge and agree that the attorney-client privilege, attorney work product and expectation of client confidence involving general business matters of the Acquired Companies and arising prior to the Closing for the benefit of both Seller and any Acquired Company shall be subject to a joint privilege between Seller, on the one hand, and the Acquired Company, on the other hand, and Seller and the Acquired Company shall have equal right to assert all such joint privilege and protection and no such joint privilege may be waived by (a) Seller without the prior written consent of Buyer on behalf of the Acquired Company or (b) the Acquired Company without the prior written consent of Seller.

6.17 Environmental Assessments; Phase I Investigations.

- (a) Within thirty (30) days from the date of this Agreement (the "Phase I Time Period"), Buyer shall have the right, at its sole cost and expense, to engage an environmental consulting firm to conduct a Phase I Environmental Assessment and Compliance Review, as such terms are commonly understood ("Phase I Environmental Assessment") with respect to any Owned Real Property provided such environmental assessment, investigation or review shall be conducted only (i) during regular business hours, (ii) with no less than two (2) Business Days prior written notice to Seller, and (iii) in a manner which will not unduly interfere with the operation of the Business or the use of access to or egress from the Owned Real Property. Any damage to the Owned Real Property caused by Buyer and its consultants in conducting any such environmental assessment, investigation or review shall be repaired by Buyer at its sole cost and expense.
- (b) If any Phase I Environmental Assessment details a Recognized Environmental Condition (as such term is defined in the ASTM Standard for Phase I Environmental Assessments) (a "Recognized Environmental Condition") in respect of any Owned Real Property, the environmental consultant recommends in the Phase I Environmental Assessment a Phase II Environmental Assessment ("Phase II Environmental Assessment"), and Buyer makes a request in writing to Seller within five Business Days following expiration of the Phase I Time Period to conduct such Phase II Environmental Assessment (a "Phase II Request"), then Seller shall have the right, by written notice to Buyer within 10 days after receipt of the Phase II Request (the "Phase II Election Notice"), to: (i) propose an alternative method for handling the risk of the Recognized Environmental Condition; or (ii) agree in principle to the Phase II Request (a "Phase II Consent"). In the event Seller elects option (i) of this Section 6.17(b), the parties shall use their respective good faith efforts to reach an agreement in writing with respect to such alternative method, and such agreement, if any, shall be binding upon the parties. If the parties cannot

reach such an agreement within 10 days after Buyer's receipt of the Phase II Election Notice, Buyer may terminate this Agreement by written notice to Seller within five Business Days after the expiration of such 10-day period unless Seller, by written notice to Buyer within such five-Business Day period, agrees in principle to the Phase II Request (which shall also be deemed a "Phase II Consent" hereunder). In the event Seller provides Buyer with a Phase II Consent pursuant to this Section. Buyer shall deliver to Seller. as promptly as practicable and in any event no later than 10 days after Buyer's receipt of the Phase II Consent, a copy of the proposed work plan, which shall be commercially reasonable, for the Phase II Environmental Assessment, and the parties shall use their respective good faith efforts to reach an agreement in writing with respect to the scope of such Phase II Environmental Assessment. If the parties cannot reach such an agreement within 10 days after Seller's receipt of Buyer's proposed work plan, then Buyer may terminate this Agreement by written notice to Seller within five Business Days after the expiration of such 10-day period unless Seller, by written notice to Buyer prior to the expiration of such five-Business Day period, agrees to the work plan proposed by Buyer. If the parties reach an agreement pursuant to this Section with respect to the scope of the Phase II Environmental Assessment (such agreed upon scope of the Phase II Environmental Assessment, the "Phase II Inspection"), Buyer shall have the right, within 30 days following the date upon which the parties reach such agreement (the "Phase II Time Period"), to conduct, at Buyer's sole cost and expense, the Phase II Inspection; provided such Phase II Inspection shall be conducted only (A) during regular business hours, (B) with no less than two (2) Business Days prior written notice to Seller, and (C) in a manner which will not unduly interfere with the operation of the Business or the use of access to or egress from the Owned Real Property. Any damage to the Owned Real Property caused by Buyer and its consultants in conducting any Phase II Inspection shall be repaired by Buyer at its sole cost and expense. The Phase I Environmental Assessment and Phase II Environmental Assessment are hereinafter referred to as the "Assessments".

- Promptly upon its receipt thereof, Buyer shall deliver to Seller a copy of the environmental consultant's report generated in connection with the Phase II Inspection (the "Phase II **Report**), which shall include, if applicable, an estimate of the cost and expense of further investigation, clean up, removal, remedial, corrective or responsive action necessary to address the Recognized Environmental Condition (the "Environmental Work"); provided, however, the Environmental Work shall be designed to meet the least stringent standards or requirements so as to cause the Recognized Environmental Condition not to be a violation under applicable Environmental Law. In the event the Phase II Report recommends Environmental Work, then (i) Seller, by written notice to Buyer within ten (10) Business Days after the expiration of the Phase II Time Period, may (A) elect to cause such Environmental Work to be performed prior to the Closing at Seller's sole cost and expense, (B) provide Buyer with a purchase price reduction for the estimated cost of such Environmental Work or (C) if the estimated cost of such Environmental Work exceeds Seven Million Five Hundred Thousand Dollars (\$7,500,000) terminate this Agreement (provided that if Seller terminates this Agreement pursuant to this clause (C), Seller shall pay to Buyer an amount equal to the costs and expenses incurred by Buyer in connection with this Agreement, including the costs of conducting the Assessments), or (ii) Buyer, if Seller does not make a timely election pursuant to the preceding clause (i), may (A) elect to terminate this Agreement by written notice to Seller within five (5) Business Days after the expiration of such ten (10) Business Day period or (B) proceed to close the transaction with Seller indemnifying Buyer against any Losses related to the Environmental Work.
- Assessment and any Phase II Environmental Assessment regardless of whether this Agreement is terminated or closed. Seller and Buyer agree that the results of the Assessments carried out pursuant to this Section 6.17 shall not be disclosed to any third party, unless such disclosure is required by Law or is made in connection with any Action between the parties relating to this Section 6.17; provided, however, that each party may disclose such information to such party's officers, directors, employees, lenders, advisors, attorneys, Debt Financing Sources and accountants who need to know such information in connection with the consummation of the transactions contemplated by this Agreement and who are informed by such party of the confidential nature of such information.
- (e) If the Closing is not consummated hereunder, neither party shall have any obligation to remediate any potential liability with respect to the results of such Assessments. Any

damage to the Real Property caused by Buyer and its consultants in conducting any Assessment shall be repaired by Buyer at its sole cost and expense. Any such Environmental Work required under this Section 6.17 shall entail a reasonably cost-effective method necessary so as to cause the Recognized Environmental Condition not to be a violation under applicable Environmental Law. In the event that Seller elects to withhold consent for any Phase II Environmental Assessment, Buyer, at its option, may (1) terminate this Agreement by written notice to the other party without further liability or (2) proceed to close the transaction with Seller indemnifying Buyer against any Losses directly related to facts that would have been discovered in such Phase II Environmental Assessment, including any Environmental Work.

- **6.18 Satellite Matter**. Buyer and Seller hereby agree to the provisions of Schedule 6.18, which is hereby incorporated by reference
- **6.19 Tower Matter**. Buyer and Seller hereby agree to the provisions of Schedule 6.19, which is hereby incorporated by reference.
- **6.20 Certain Additional Matters**. Buyer and Seller hereby agree to the provisions of Schedule 6.20, which is hereby incorporated by reference.

ARTICLE 7

SELLER CLOSING CONDITIONS

The obligations of Seller to consummate the Closing hereunder shall be subject to the satisfaction, at or prior to the Closing, of each of the following conditions (unless waived in writing by Seller):

7.1 Representations and Covenants.

- (a) All representations and warranties of Buyer contained in this Agreement shall be true and correct in all material respects as of the date of this Agreement and at and as of the Closing (other than any representation or warranty that is expressly made as of a specified date, which need be true and correct in all material respects as of such specified date only); except where such failure has not resulted, and would not reasonably be expected to result, in a material adverse effect on the ability of Buyer to perform its obligations under this Agreement or any Buyer Ancillary Agreement, *provided, however,* that for purposes of this Section 7.1(a), all materiality or similar qualifiers within such representations and warranties shall be disregarded.
- **(b)** The covenants and agreements that by their terms are to be complied with and performed by Buyer at or prior to the Closing shall have been complied with or performed by Buyer in all material respects.
- (c) Seller shall have received a certificate dated as of the Closing Date from Buyer executed by an authorized officer of Buyer to the effect that the conditions set forth in Sections 7.1(a) and (b) have been satisfied.
- **7.2 Proceedings**. Neither Seller nor Buyer shall be subject to any Governmental Order, which remains in effect, prohibiting or making illegal the consummation of the transactions contemplated hereby. No Action shall have been commenced by any Governmental Entity against Seller, any Acquired Company or Buyer seeking to restrain the transactions contemplated by this Agreement that would render it unlawful to consummate such transactions.
- **7.3 FCC Authorization**. The FCC Consent shall have been granted and shall be in full force and effect.
 - 7.4 Hart Scott Rodino. The HSR Clearance shall have been obtained.

7.5 Deliveries. Buyer shall have complied with each of its obligations set forth in Section 9.2

ARTICLE 8

BUYER CLOSING CONDITIONS

The obligations of Buyer to consummate the Closing hereunder shall be subject to satisfaction, at or prior to the Closing, of each of the following conditions (unless waived in writing by Buyer):

8.1 Representations and Covenants.

- (a) All representations and warranties of Seller contained in this Agreement shall be true and correct as of the date of this Agreement and at and as of the Closing (other than any representation or warranty that is expressly made as of a specified date, which need be true and correct as of such specified date only), except to the extent that the failure of the representations and warranties of Seller contained in this Agreement to be so true and correct at and as of the Closing (or in respect of any representation or warranty that is expressly made as of a specified date, as of such date only) has not had and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect; provided, however, that for purposes of this Section 8.1(a), all materiality, "Material Adverse Effect" or similar qualifiers within such representations and warranties shall be disregarded.
- **(b)** The covenants and agreements that by their terms are to be complied with and performed by Seller at or prior to the Closing shall have been complied with or performed by Seller in all material respects.
- **(c)** Buyer shall have received a certificate dated as of the Closing Date from Seller executed by an authorized officer or manager of Seller to the effect that the conditions set forth in Sections 8.1(a) and (b) have been satisfied.
- **8.2 Proceedings**. Neither Seller nor Buyer nor any of their respective Affiliates shall be subject to any Governmental Order, which remains in effect, prohibiting or making illegal the consummation of the transactions contemplated hereby. No Action shall have been commenced by any Governmental Entity against Seller, any Acquired Company or Buyer seeking to restrain the transactions contemplated by this Agreement that would render it unlawful to consummate such transactions.
- **8.3 FCC Authorization**. The FCC Consent shall have been granted and shall be in full force and effect.
 - **8.4 Hart Scott Rodino**. The HSR Clearance shall have been obtained.
- **8.5 Deliveries**. Seller shall have complied with each of its obligations set forth in Sections 2.2(a), 2.2(b), 2.3(a) and 9.1.
 - **8.6 Consents**. The Required Consents shall have been obtained and delivered to Buyer.
- **8.7 No MAE.** From the date of this Agreement, there shall not have occurred any Material Adverse Effect.

ARTICLE 9

CLOSING DELIVERIES

9.1 Seller Documents. At the Closing, Seller shall deliver or cause to be delivered to Buyer:

- (a) a good standing certificate issued by the Secretary of State of Seller's and each Acquired Company's jurisdiction of formation;
- **(b)** certified copies of all limited liability company resolutions necessary to authorize Seller's execution, delivery and performance of this Agreement, including the consummation of the transactions contemplated hereby;
 - (c) the certificate described in Section 8.1(c);
- **(d)** certificates representing the Equity Interests, in each case duly endorsed or accompanied by duly executed membership interest powers for transfer to Buyer;
- **(e)** a certificate of non-foreign status conforming to the requirements of Treasury Regulations Section 1.1445-2(b)(2), dated as of the Closing Date and stating that Seller is not a "foreign person" as defined in Section 1445 of the Code, substantially the form attached hereto as **EXHIBIT C**;
- (f) the transition services agreement in substantially the form attached hereto as **EXHIBIT D** (the "*Transition Services Agreement*"), duly executed by Seller;
 - (g) the Indemnity Escrow Agreement, duly executed by Seller and the Escrow Agent;
- **(h)** the Organizational Documents for Seller and each of the Acquired Companies, certified as of a recent date by the Secretary of State of the applicable jurisdiction of organization;
- (i) written resignations of the managers, directors and officers of the Acquired Companies, to be effective as of the Closing, duly executed by such managers, directors and officers;
- (j) mortgage discharges and UCC termination statements or other appropriate releases, in form and substance reasonably satisfactory to Buyer, which when filed will release and satisfy any and all Liens relating to (i) any of the Acquired Company's assets or properties (other than Permitted Liens) or (ii) the Equity Interests, together with proper authority to file such termination statements or other releases at and following the Closing;
- (k) an assignment and assumption agreement, duly executed by Cordillera Interactive LLC (d/b/a Cordillera Digital), pursuant to which Cordillera Digital assigns all of its rights under the CMS MSA solely with respect to the Business to Buyer (including all of Cordillera Digital's rights, licenses, title and/or interest, as applicable, in and to the Deliverables as defined in the CMS MSA) and Buyer assumes all of Cordillera's liabilities and obligations thereunder with respect to the Business accruing after the Closing Date, in each case effective as of the Closing (the "CMS Assignment and Assumption"); and
- (I) such other documents or instruments as Buyer reasonably requests and are reasonably necessary to consummate the transactions contemplated by this Agreement.
- **9.2 Buyer Documents**. At the Closing, Buyer shall deliver or cause to be delivered to Seller (or, in the case of the applicable portion of the Purchase Price, the applicable third parties as contemplated under Section 2.2):
 - (a) the Purchase Price in accordance with Section 2.2;
 - **(b)** the Transition Services Agreement, duly executed by Buyer;
 - (c) the Indemnity Escrow Agreement, duly executed by Buyer;

- (d) a good standing certificate issued by the Secretary of State of Buyer's jurisdiction of formation:
- **(e)** certified copies of all corporate or other resolutions necessary to authorize Buyer's execution, delivery and performance of this Agreement, including the consummation of the transactions contemplated hereby;
 - (f) the CMS Assignment and Assumption, duly executed by Buyer; and
 - (g) the certificate described in Section 7.1(c).

ARTICLE 10

SURVIVAL; INDEMNIFICATION

Survival. The representations and warranties of Seller and Buyer and the covenants and agreements of Seller and Buyer to be performed in full prior to Closing shall survive the Closing until the 15 month anniversary of the Closing Date, other than (i) the representations and warranties contained in Section 3.1, Section 3.2, and Section 3.18, which shall survive until 60 days following the expiration of the applicable statute of limitations (including any extension thereof, whether automatic or permissive), and (ii) the representations and warranties contained in Section 3.5, which shall survive until 180 days following the expiration of the applicable statute of limitations (including any extension thereof, whether automatic or permissive). The covenants and agreements of Seller and Buyer to be performed in whole or in part after Closing shall survive the Closing until performed in full in accordance with their respective terms. Claims for indemnification under Section 10.2(a)(iii) or Section 10.2(a)(iv) shall survive the Closing until 90 days following the expiration of the applicable statute of limitations (including any extension thereof, whether automatic or permissive). Seller and Buyer shall not have any liability under Sections 10.2 and 10.3, respectively, unless a claim for Losses for which indemnification is sought thereunder is asserted by Buyer or Seller, as the case may be, within the applicable survival period set forth above; provided, however, that the good faith written assertion of any such claim by a party against the other party hereunder prior to the applicable survival period set forth above shall extend the survival period with respect to such claim through the date such claim is conclusively resolved.

10.2 Indemnification by Seller.

- (a) Subject to the limitations, conditions and provisions set forth in this ARTICLE 10, after the Closing, Seller shall indemnify and hold Buyer and its Affiliates (including, following the Closing, the Acquired Companies) and their respective partners, shareholders, members, officers, directors, managers, employees and other representatives (each a "Buyer Indemnified Party" and, collectively, the "Buyer Indemnified Parties"), harmless from and against and with respect to, and shall pay and reimburse each of them for, any and all Losses incurred or suffered by, or imposed upon, any such Buyer Indemnified Party resulting from, arising out of, or relating to (i) any inaccuracy in or breach of any representation or warranty of Seller contained in this Agreement or in any certificate or instrument delivered by or on behalf of Seller pursuant to this Agreement, (ii) any failure by Seller to perform any of its covenants or agreements set forth in this Agreement, (iii) any Company Transaction Costs or Indebtedness of any Acquired Company outstanding as of the Closing to the extent not deducted from the Purchase Price pursuant to Section 2.2, and (iv) that matter listed on Schedule 10.2(a)(iv).
- **(b)** Seller's obligation to indemnify the Buyer Indemnified Parties pursuant to Section 10.2(a) shall be subject to the following limitations and provisions, as applicable:
- (i) Seller's obligations to indemnify Buyer and each Acquired Company with respect to Seller Taxes as set forth in Section 11.1(a) are solely as set forth in Section 11.1, which shall be separate and distinct from (but not duplicative of any Losses indemnified pursuant to) the provisions of this **ARTICLE 10** and such indemnification obligations for Seller Taxes shall not be subject to Section

10.2(b)(ii) or Section 10.2(b)(iii)(A), but shall be subject to Section 10.2(b)(iii)(B), Section 10.2(b)(iii)(C) and Section 10.2(c).

(ii) No indemnification shall be required to be made by Seller under Section 10.2(a)(i) with respect to any claims for Losses thereunder until the aggregate amount of all of the Buyer Indemnified Parties' Losses under Section 10.2(a)(i) exceeds Two Million Six Hundred Five Thousand Dollars (\$2,605,000) (the "Deductible Amount"), in which case Seller shall then be liable thereunder to the Buyer Indemnified Parties only for such Losses in excess of the Deductible Amount; provided, however, that the limitation set forth in this Section 10.2(b)(ii) shall not apply to Losses resulting from, arising out of, or relating to any inaccuracy in or breach of any representation or warranty in Section 3.1, Section 3.2, Section 3.5 or Section 3.18 or claims for fraud.

(iii) After the Closing:

- (A) the Buyer Indemnified Parties' sole and exclusive remedy for any Losses arising under Section 10.2(a)(i) shall be a claim for indemnification against the Indemnity Escrow Funds for which the Buyer Indemnified Parties are entitled to indemnification from Seller pursuant to Section 10.2, which shall constitute a cap on the maximum total liability of Seller arising under Section 10.2(a)(i); provided, however, that the limitation set forth in this Section 10.2(b)(iii)(A) shall not apply to Losses resulting from, arising out of, or relating to any inaccuracy in or breach of any representation or warranty in Section 3.1, Section 3.2, Section 3.3, Section 3.5, or Section 3.18 or claims for fraud;
- (B) prior to the date that is the fifteen (15) month anniversary of the Closing Date and so long as any Indemnity Escrow Funds remain available, the Buyer Indemnified Parties shall first make claims for indemnification against the Indemnity Escrow Funds for any Losses (including Losses arising under Section 10.2(a)(i) relating to any inaccuracy in or breach of any representation or warranty in Section 3.1, Section 3.2, Section 3.3, Section 3.5, or Section 3.18, arising under Sections 10.2(a)(ii), (a)(iii) or (a)(iv), or arising under Section 11.1(a)); and
- (C) the aggregate amount of all Losses for which Seller shall be liable arising under Section 10.2(a)(i) relating to any inaccuracy in or breach of any representation or warranty in Section 3.1, Section 3.2, Section 3.3, Section 3.5, or Section 3.18, arising under Sections 10.2(a)(ii), (a)(iii) and (a)(iv) and collectively for all Losses under Section 10.2(a), and arising under Section 11.1(a), together with all other Losses of Buyer under this Agreement, shall be an amount equal to the Purchase Price, which shall in all cases constitute a cap on the total maximum liability of Seller under this Agreement.
- (iv) For purposes of this ARTICLE 10, solely for the purpose of calculating the amount of any Losses arising out of or resulting from any inaccuracy or breach of any representation or warranty of Seller contained in this Agreement (and not for determining the existence of any inaccuracy or breach of any representation or warranty of Seller contained in this Agreement), the representations and warranties of Seller contained in this Agreement shall be deemed to have been made without any qualifications as to materiality or Material Adverse Effect.
- (c) Seller, Buyer and the Escrow Agent shall execute and deliver, at or prior to the Closing, an Indemnity Escrow Agreement substantially in the form attached hereto as **Exhibit E**, with such changes as Seller and Buyer may reasonably agree that are consistent with this Agreement (the "Indemnity Escrow Agreement"), in accordance with which, on the Closing Date, pursuant to Section 2.2, Buyer shall deposit with the Escrow Agent an amount in cash equal to Twenty-Six Million Fifty Thousand Dollars (\$26,050,000) (the "Indemnity Escrow Deposit") (all amounts held from time to time by the Escrow Agent pursuant to the Indemnity Escrow Agreement in respect of such deposit, including any interest or other earnings in respect of such deposit, the "Indemnity Escrow Funds"). The Indemnity Escrow Funds shall be held and disbursed in accordance with the terms of this Agreement and the Indemnity Escrow Agreement and the following provision: no later than the second Business Day after the fifteen (15) month anniversary of the Closing Date, the portion of the Indemnity Escrow Funds not then subject to outstanding indemnification claims by any Buyer Indemnified Party pursuant to this

Agreement shall be released by the Escrow Agent to Seller. In connection with the release of any portion of the Indemnity Escrow Funds pursuant to this Section 10.2(c), Seller and Buyer shall promptly execute and deliver to the Escrow Agent in accordance with the Indemnity Escrow Agreement written instructions instructing the Escrow Agent to make the payments set forth in this Section 10.2(c).

10.3 Indemnification by Buyer.

- (a) Subject to the limitations, conditions and provisions set forth in this ARTICLE 10, after Closing, Buyer hereby agrees to indemnify and hold Seller and its partners, shareholders, members, officers, directors and employees, as the case may be (each a "Seller Indemnified Party" and, collectively, the "Seller Indemnified Parties"), harmless from and against and with respect to, and shall reimburse any of them for, any and all Losses suffered by such Seller Indemnified Party resulting from, arising out of, or relating to any breach of any representation or warranty of Buyer made, or any failure by Buyer to perform any of its covenants or agreements set forth, in this Agreement.
- **(b)** Buyer's obligation to indemnify the Seller Indemnified Parties pursuant to Section 10.3(a) shall be subject to all of the following limitations:
- (i) Buyer's obligation to indemnify Seller with respect to any Losses set forth in Section 11.1(b) are solely as set forth in Section 11.1(b), which shall be separate and distinct from the provisions of this **ARTICLE 10**, and not be subject to this Section 10.3.
- (ii) After the Closing, the Seller Indemnified Parties' sole and exclusive remedy for any Losses (whether such Losses result from a claim framed in tort, contract or otherwise) arising out of or in connection with this Agreement, the other Ancillary Documents and the transactions contemplated hereby and thereby shall be a claim for indemnification from Buyer pursuant to this Section 10.3.
- **10.4 Procedure for Indemnification**. The procedure for indemnification pursuant to this **ARTICLE 10** and **SECTION 11.1** (except to the extent of a conflict with Section 11.8, in which case Section 11.8 shall govern) shall be as follows:
- (a) It is understood and agreed that a party may be entitled to indemnification, in accordance with the provisions hereof, whether based on a claim directly between the parties or based on a claim brought by a third party. The party claiming indemnification (the "Claimant") shall promptly give notice to the party from which indemnification is claimed (the "Indemnifying Party") and, in the case of a claim by a Buyer Indemnified Party against the Indemnity Escrow Funds, the Escrow Agent, of any claim, whether between the parties or brought by a third party, specifying in reasonable detail the factual basis for the claim and the amount thereof, if reasonably practicable and estimated in good faith, and containing a reference to the provision(s) of this Agreement under which such indemnification claim is made. If the claim relates to an Action by a third party against Claimant (a "Third Party Action"), such notice shall be given by Claimant within a reasonable period of time after Claimant receives notice of such Third Party Action, which in any event shall be no later than 20 Business Days after Claimant receives written notice of such Third Party Action. The failure to give such prompt written notice shall not, however, relieve the Indemnifying Party of its indemnification obligations, except and only to the extent that the Indemnifying Party forfeits rights or defenses by reason of such failure.
- (b) With respect to claims solely between the parties, following receipt of notice from the Claimant of a claim, the Indemnifying Party shall have up to 20 days to make such investigation of the claim as the Indemnifying Party deems necessary or desirable. For the purposes of such investigation, the Claimant agrees to make available to the Indemnifying Party and its authorized representatives such documents and other information relating to the claim as the Indemnifying Party may reasonably request. If the Claimant and the Indemnifying Party agree at or prior to the expiration of such 20-day period (or any mutually agreed upon extension thereof) to the validity and amount of such claim, the Indemnifying Party shall immediately pay to the Claimant the full amount agreed to, subject to the terms hereof (including Sections 10.2(b) and 10.3(b)), including where Seller is the Indemnifying Party and the claim is to be

satisfied by Indemnity Escrow Funds, by executing and delivering written instructions to the Escrow Agent to pay the full amount so agreed from the Indemnity Escrow Funds. If the Claimant and the Indemnifying Party do not agree within such 20-day period (or any mutually agreed upon extension thereof), the Claimant may seek appropriate remedies at law or equity, as applicable, to enforce its claim for indemnification hereunder, subject to the applicable limitations of Sections 10.2(b) and 10.3(b).

- The Claimant will be entitled, at the sole expense and liability of the Indemnifying Party and subject to the other terms and conditions herein, to exercise full control of the defense, compromise and settlement of any Third Party Action, unless (i) the Indemnifying Party, promptly after the giving of notice of the Third Party Action by the Claimant to the Indemnifying Party (but in no event later than 15 days thereafter), notifies the Claimant in writing of the Indemnifying Party's intention to assume such defense, acknowledges its obligation to indemnify the Claimant for Losses related to such Third Party Action, and promptly thereafter, retains legal counsel reasonably satisfactory to the Claimant to conduct the defense of such Third Party Action, (ii) the Third Party Action involves only money damages and does not seek an injunction or other equitable relief against the Claimant, (iii) the Third Party Action does not relate to or otherwise arise in connection with any criminal or regulatory enforcement action, and (iv) the Indemnifying Party conducts the defense of the Third Party Action diligently. In the event that the Indemnifying Party does not, within fifteen (15) days of its receipt of notice of a Third Party Action pursuant to subclause (i) above of this Section 10.4(c), elect to undertake such defense, Claimant may undertake the defense, opposition, compromise or settlement of such Third Party Action with counsel selected by it at the Indemnifying Party's cost. If the Claimant defends any Third Party Action pursuant to the preceding sentence or pursuant to subclauses (ii) - (iv) above of this Section 10.4(c), then the Indemnifying Party shall promptly reimburse the Claimant for the reasonable costs and expenses of defending such Third Party Action upon submission of periodic bills.
- (d) Each party will cooperate with the other party in connection with the defense, compromise or settlement of any Third Party Action in accordance with this Agreement in any manner that reasonably may be requested. If the Indemnifying Party so assumes and prosecutes the defense of any such Third Party Action in accordance with Section 10.4(c), the Claimant will have the right to employ separate counsel and to participate in (but not control) the defense, compromise or settlement of the Third Party Action, but the fees and expenses of such counsel will be at the expense of the Claimant, unless (i) the Indemnifying Party has agreed to pay such fees and expenses or (ii) the Claimant has been advised by its legal counsel that there may be one or more defenses available to it which are different from or additional to those available to the Indemnifying Party or that an actual or potential conflict of interest would make the representation of both the Indemnifying Party and the Claimant by the same counsel inadvisable, and in any such case the reasonable fees and expenses of such separate counsel shall be paid by the Indemnifying Party.
- (e) No Claimant will settle or compromise any such Third Party Action for which it is entitled to indemnification under this Agreement without the prior written consent of the Indemnifying Party, such consent not to be unreasonably withheld, delayed or conditioned. Except as otherwise provided in this Section 10.4(e), an Indemnifying Party shall not settle or compromise any Third Party Action without the prior written consent of the Claimant, such consent not to be unreasonably withheld, delayed or conditioned. An Indemnifying Party may settle or compromise any Third Party Action without such consent so long as the Indemnifying Party (i) receives a firm offer to settle such Third Party Action without leading to liability or the creation of a financial or other obligation on the part of any Claimant, (ii) pays the entire amount of such settled or compromised Third Party Action, (iii) such settlement or compromise does not contain an admission of wrongdoing on the part of any Claimant or contain any non-monetary terms, (iv) such settlement or compromise provides for the full and unconditional release of the Claimant in customary form from all liability in connection with such Third Party Action and (v) the Indemnifying Party gives prior written notice to the Claimant of such proposed settlement or compromise and the Claimant fails to consent to such proposed settlement or compromise within 10 days after its receipt of such notice.
- **10.5 No Circular Recovery**. Seller, on behalf of itself and its Affiliates, hereby agrees that it will not make any claim for indemnification against any Buyer Indemnified Party by reason of the fact that

Seller or any of its Affiliates was a controlling person, director, manager, employee or representative of any of the Acquired Companies or was serving as such for another Person at the request of any of the Acquired Companies (whether such claim is for Losses of any kind or otherwise) with respect to any claim brought by a Buyer Indemnified Party against Seller under this Agreement or otherwise relating to this Agreement or otherwise relating to this Agreement or otherwise relating to this Agreement, Seller expressly waives any right of subrogation, contribution, advancement, indemnification or other claim against the Acquired Companies with respect to any amounts owed by Seller pursuant to this **ARTICLE 10** or otherwise.

ARTICLE 11

TAX MATTERS

11.1 Tax Indemnity.

- (a) Subject to Sections 10.2(b)(i) and 10.4, Seller hereby agrees to indemnify Buyer and each Acquired Company and hold them harmless from and against (i) all Taxes (other than Transfer Taxes which, for the avoidance of doubt, are addressed in Section 11.1(a)(iii) and Section 11.6) of Seller with respect to the transactions contemplated by this Agreement and otherwise, (ii) any Losses attributable to all Taxes of the Acquired Companies with respect to any Pre-Closing Tax Period, and (iii) Seller's allocable portion of all Transfer Taxes pursuant to Section 11.6 (collectively, "Seller Taxes"); provided, however, that Seller Taxes shall not include any Taxes solely to the extent directly resulting from a breach by Buyer of Sections 11.3(a), 11.3(c), and 11.6.
- **(b)** Buyer hereby agrees to indemnify Seller and its Affiliates and hold them harmless from and against any Losses attributable to all Taxes of the Acquired Companies arising with respect to a Tax period, or portion thereof, beginning after the Closing Date (a "*Post-Closing Tax Period*").
- **(c)** Notwithstanding anything to the contrary in this Agreement, the provisions of this Section 11.1 shall continue in effect until 180 days following the expiration of the applicable Tax-related statute of limitations (including any extension thereof, whether automatic or permissive), at which time such obligations shall expire and be of no further force or effect.
- 11.2 Straddle Period Allocation. For purposes of Section 11.1, the amount of Taxes allocable to the portion of a Straddle Period ending on the Closing Date shall be: (a) in the case of Taxes imposed on a periodic basis (such as real or personal property Taxes and similar ad valorem obligations), the amount of such Taxes for the entire period multiplied by a fraction, the numerator of which is the number of calendar days in the Straddle Period ending on and including the Closing Date and the denominator of which is the number of calendar days in the entire relevant Straddle Period, and (b) in the case of Taxes not described in clause (a) above (such as payroll Taxes, Taxes imposed upon or measured by income, Taxes based upon occupancy or imposed in connection with any sale or other transfer or assignment of property (real or personal, tangible or intangible)), the amount of any such Taxes shall be determined as if such taxable period ended as of the close of business on the Closing Date (and in the case of any Taxes attributable to the ownership of any equity interest in any partnership or other "flowthrough" entity, as if the taxable period of such partnership or other "flowthrough" entity ended as of the end of the Closing Date).

11.3 Tax Returns.

(a) Buyer, Seller and their respective Affiliates acknowledge and agree that the sale and purchase of the Equity Interests shall be treated for U.S. federal income Tax purposes, and any analogous state or local Tax purposes, as the sale and purchase of the assets of the Acquired Companies. Buyer, Seller and their respective Affiliates will file all Tax Returns in a manner consistent with such treatment, and will take no position inconsistent with such characterization for federal, state or

local income Tax purposes, including in any audit, judicial or administrative proceeding, unless otherwise required by applicable Tax Law.

- **(b)** Except as otherwise provided in Section 11.6, Seller shall prepare and file (or cause to be prepared and filed) (i) all Flow-Through Tax Returns, and (ii) all Tax Returns (other than Flow-Through Tax Returns) for the Acquired Companies, for taxable periods ending on or before the Closing Date (each Tax Return described in clause (ii), a "*Pre-Closing Tax Return*"). Each such Pre-Closing Tax Return shall be prepared in a manner consistent with past custom and practice, except as otherwise required by applicable Law.
- (c) Buyer shall prepare and file (or cause to be prepared and filed), all Tax Returns of the Acquired Companies with respect to a Straddle Period (each, a "Straddle Tax Return"), and each such Straddle Tax Return shall be prepared in a manner consistent with past custom and practice except as otherwise required by applicable Law. Buyer shall provide a copy of each such Straddle Tax Return, together with all supporting documentation and workpapers, to Seller for Seller's review and comment at least thirty (30) days prior to the due date (taking into account all valid extensions) for filing such Straddle Tax Return, and Buyer shall include any reasonable comments provided in writing by Seller to Buyer at least fourteen (14) days prior to the due date (taking into account all valid extensions) for filing such Straddle Tax Return.
- Refunds. The amount or economic benefit of any Tax refund (whether in cash or as a credit against or offset to any Tax) in respect of any Tax of the Acquired Companies attributable to any Pre-Closing Tax Period received by Buyer, the Acquired Companies or any of their respective Affiliates shall be for the account of Seller, to the extent such refund exceeds the amount of the refund of such Taxes reflected as an asset on the books of the Acquired Companies, and the recipient thereof shall pay such amount (including any interest received thereon, but less any Taxes, reasonable costs or expenses incurred by Buyer, the Acquired Companies or any of their respective Affiliates in connection with obtaining such Tax refund) over to Seller within ten (10) days after any such refund is received, credited or applied as an offset, as the case may be. Buyer shall elect not to carry any loss, credit or other Tax benefit item from a Post-Closing Tax Period (a "Post-Closing Tax Refund") back to a Pre-Closing Tax Period, except in any case where it is required by Law to carry back such Post-Closing Tax Refund, in which case any refund due that is attributable to such Post-Closing Tax Refund will be payable to Buyer. Notwithstanding anything in this Agreement to the contrary, in the event that any such Tax refund is subsequently determined by any Governmental Entity to be less than the amount paid by the recipient thereof to Seller pursuant to this Section 11.4. Seller shall return any such disallowed amounts (plus any interest and penalties in respect of such disallowed Tax refunds owed to a Governmental Entity) to such payor within ten (10) Business Days after receipt of written notice from such payor requesting the same.
- Cooperation. Each of Buyer and Seller, and each of their respective Affiliates, shall cooperate fully, as and to the extent reasonably requested by the other party, in connection with the filing of Tax Returns of or with respect to the Acquired Companies and/or during the course of any audit, litigation or other proceeding with respect to Taxes of or attributable to the Acquired Companies. Such cooperation shall include the retention and (upon the other party's request) the provision of records and information that are reasonably relevant to any such audit, litigation or other proceeding and making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder. Each of Seller and Buyer agrees, and following the Closing, Buyer agrees to cause the Acquired Companies, (i) to retain all books and records with respect to Tax matters pertaining to the Acquired Companies until the expiration of the applicable statute of limitations, and (ii) to give the other party (and following the Closing, Buyer agrees to cause the Acquired Companies to give Seller) reasonable written notice prior to transferring, destroying or discarding any such books and records and, on receipt of such notice, if the other party so requests, Seller or Buyer shall, as applicable, allow the other party to take possession of such books and records. Buyer and Seller agree that Seller shall be entitled to retain copies of the books and records of the Acquired Companies which they believe are appropriate for the filing of Tax Returns as described in this Article 11. Buyer and Seller further agree, upon request, to use commercially reasonable efforts to obtain any certificate or other document from any Governmental Entity or any other Person as may be necessary to mitigate, reduce or eliminate

any Tax that could be imposed (including with respect to the transactions contemplated hereby). Any information or documents provided or retained under this Section 11.5 shall be kept confidential by the party receiving or retaining the information or documents, except as may otherwise be necessary in connection with the filing of Tax Returns or in connection with any administrative or judicial proceedings relating to Taxes.

- 11.6 Transfer Taxes. Buyer and Seller shall each be responsible for the payment of fifty percent (50%) of all governmental Taxes, fees and charges applicable to the transfer of the Equity Interests under this Agreement (including sales, use and real property transfer taxes, stamp and stock transfer taxes and the costs of recording or filing all applicable conveyance instruments) (collectively, "Transfer Taxes") and for the reasonable out-of-pocket costs and expenses incurred in preparing and filing any Tax Returns with respect to such Transfer Taxes. The parties hereto will cooperate, to the extent reasonably requested and as permitted by applicable Law, in minimizing any such Transfer Taxes. The party required by applicable Law to file a Tax Return with respect to such Transfer Taxes will (a) do so within the time period prescribed by applicable Law, pay the full amount of such Transfer Tax to the appropriate Governmental Entity in accordance with applicable Law, (b) notify the other party in writing of the amount paid, and (c) attach to such notification a copy of any Tax Returns filed in connection with such Transfer Tax. The other party shall pay the party that paid the Transfer Tax an amount equal to fifty percent (50%) of such Transfer Tax by check or wire transfer of immediately available funds within five (5) Business Days after receiving such notice.
- Purchase Price Allocations. Within ninety (90) days following the determination of the Final Adjustment Amount pursuant to Section 2.3(i), Buyer shall deliver to Seller a draft allocation of the Purchase Price (and all other consideration payable pursuant to this Agreement, as determined for U.S. federal income tax purposes) among the assets of the Acquired Companies in accordance with Section 1060 of the Code and the Treasury regulations promulgated thereunder (the "Draft Allocation"). If Seller does not object to the Draft Allocation within thirty (30) days of receipt thereof, then the Draft Allocation shall become final and binding on the parties. Seller shall make any objection to the Draft Allocation in writing to Buyer and shall set forth the basis for such objection. If Seller timely objects to the Draft Allocation, then Buyer and Seller agree to negotiate in good faith to resolve any disputes over the Draft Allocation and to attempt to resolve any differences between the parties. Seller and Buyer and their Affiliates shall report, act and file Tax Returns (including, but not limited to IRS Form 8594) in all respects and for all purposes consistent with the allocation as agreed to (the "Final Allocation") and shall not take any position (whether in audits or otherwise) that is inconsistent with the Final Allocation; provided, however, that if Buyer and Seller fail to agree on a Final Allocation, the disputed items shall be referred to the Independent Accountant for resolution in accordance with the procedures set forth in Section 2.3(f) and the fees and expenses of such Independent Accountant shall be borne by the parties in accordance with the provisions of Section 2.3(f). Any payments to be made pursuant to Section 2.3(i) and any amounts paid by an Indemnifying Party under Article 10 and Section 11.1 shall be treated as an adjustment to the Purchase Price (including by Buyer and Seller on their respective Tax Returns) for Tax purposes and allocated as provided by Treasury Regulation § 1.1060-1(c). The parties will cooperate in good faith to promptly update the Final Allocation for any adjustments to the Purchase Price and will prepare and file amended Tax Returns to reflect such updated allocation, and not take any position (whether in audits or otherwise) that is inconsistent with such updated allocation.

11.8 Tax Claims.

- (a) Upon receipt by Buyer or any of its Affiliates of any written notice from a Taxing Authority of an audit, claim or proceeding with respect to Taxes for which any of the Buyer Indemnified Parties may be entitled to indemnification under Section 10.2 or Section 11.1 or that could otherwise increase Seller's liability for Taxes (a "*Tax Claim*"), Buyer shall promptly notify Seller in writing. Such notice shall include a copy of the relevant portion of any correspondence received from the relevant Taxing Authority and describe in reasonable detail the nature of such Tax Claim.
- **(b)** Seller shall have the right to control in good faith any Tax Claim that relates solely to one or more taxable periods that end on or prior to the Closing Date or to a Flow-Through Tax

Return (including the right to settle or compromise such Tax Claim); provided, however that, other than with respect to any Tax Claim that relates to a Flow-Through Tax Return: (i) Seller will keep Buyer informed concerning the progress, developments and events of the portion of such Tax Claim; (ii) Seller will provide to Buyer copies of all correspondence and other documents relevant to the portion of such Tax Claim; and (iii) Seller will not settle the portion of such Tax Claim, if any, related to the Acquired Companies, the Business, or the assets of the Acquired Companies, without the prior written consent of Buyer, which consent shall not to be unreasonably withheld, conditioned or delayed. Buyer will have the right to participate in the defense of any such Tax Claim (other than a claim relating to a Flow-Through Tax Return) (which will include participation in meetings with Taxing Authorities and review and comment on written submissions to Taxing Authorities) and to employ counsel, at its own expense, separate from the counsel employed by Seller.

- (c) With respect to any Tax Claim that Seller does not have the right to control pursuant to Section 11.8(b), Buyer will control in good faith such Tax Claim (including the right to settle or compromise such Tax Claim); provided, however, that Buyer (x) will keep Seller reasonably informed concerning the progress, developments and events of such Tax Claim, (y) provide Seller copies of all correspondence and other documents relevant to such Tax Claim, and (z) will not settle such Tax Claim without the prior written consent of Seller, which consent will not be unreasonably withheld, conditioned or delayed. In addition, Seller will have the right to participate in the defense of any such Tax Claim (which will include participation in meetings with Taxing Authorities and review and comment on written submissions to Taxing Authorities) and to employ counsel, at Seller's expense, separate from the counsel employed by Buyer.
- (d) Notwithstanding any provision herein to the contrary, to the extent that a provision of this Section 11.8 directly conflicts with Section 10.4, this Section 11.8 shall control.
- 11.9 Tax Treatment of Indemnity Payments. Each of Buyer and Seller agree that any indemnification payments made under this Agreement shall be treated as Purchase Price adjustments for U.S. federal income tax purposes (and state, local, and foreign Tax purposes where applicable) to the extent permitted by applicable Law.
- 11.10 Wage Reporting. To the extent applicable, Seller and the Buyer shall follow the "standard procedures" for preparing and filing Internal Revenue Service Forms W-2 (Wage and Tax Statements), as described in Revenue Procedure 2004-53 for Continuing Employees. Under this procedure, (i) Seller shall provide all required Forms W-2 to (x) all Continuing Employees reflecting wages paid and taxes withheld by Seller or any Acquired Company prior to the Employment Commencement Date, and (y) all other employees and former employees of Seller or any Acquired Company who are not Continuing Employees reflecting all wages paid and taxes withheld by Seller or any Acquired Company, and (ii) Buyer (or one of its Affiliates) shall provide all required Forms W-2 to all Continuing Employees reflecting all wages paid and taxes withheld by Buyer (or one of its Affiliates) on and after the Employment Commencement Date.
- 11.11 Tax Sharing, Indemnity and Allocation Agreements. Any and all existing Tax sharing, indemnity, allocation or similar agreements between Seller or any Affiliate of Seller (excluding the Acquired Companies), on the one hand, and any of the Acquired Companies, on the other hand, shall be terminated and all payables and receivables arising thereunder shall be settled, in each case, at or prior to Closing. After Closing, the Acquired Companies shall not have any further rights or liabilities thereunder or under any payables or receivables arising thereunder.

ARTICLE 12

Termination And Remedies

- **12.1 Termination**. This Agreement may be terminated prior to Closing as follows:
 - (a) by mutual written agreement of Buyer and Seller;

- **(b)** by written notice from Buyer to Seller if (i) Buyer is not in material breach of its obligations under this Agreement, (ii) Seller breaches its representations or warranties, or defaults in the performance of its covenants, contained in this Agreement and (iii) all such breaches and defaults of Seller that are not cured within the Cure Period would prevent any of the conditions to the obligations of Buyer set forth in Section 8.1 from being satisfied;
- (c) by written notice from Seller to Buyer if (i) Seller is not in material breach of its obligations under this Agreement, (ii) Buyer breaches its representations or warranties, or defaults in the performance of its covenants, contained in this Agreement and (iii) all such breaches and defaults of Buyer that are not cured within the Cure Period would prevent any of the conditions to the obligations of Seller set forth in Section 7.1 from being satisfied; *provided, however,* that no Cure Period shall apply to Buyer's obligation to pay the Purchase Price at the Closing; or
- (d) by written notice from Buyer to Seller, or from Seller to Buyer, if (i) the Closing does not occur by the nine (9) month anniversary of the date hereof (such date, the "*Outside Date*"), unless the Closing has not occurred by such date as a result of a material breach of this Agreement by the party providing such notice of termination, (ii) there shall be any Law that makes consummation of the transactions contemplated by this Agreement illegal or otherwise prohibited or (iii) any Governmental Entity shall have issued a Governmental Order restraining or enjoining the transactions contemplated by this Agreement, and such Governmental Order shall have become final and non-appealable.
- **12.2 Cure Period**. Prior to the Closing, each party shall give the other party prompt written notice upon learning of any breach or default by the other party under this Agreement, and such notice shall specify the breach. The term "*Cure Period*" as used herein means a period commencing on the date Buyer or Seller receives from the other written notice of breach or default hereunder and continuing until the earlier of (i) twenty (20) Business Days after receipt of such notice or (ii) five (5) Business Days after the day otherwise scheduled for the Closing.
- **12.3 Effect of Termination**. In the event that this Agreement is terminated pursuant to Section 12.1, this Agreement shall become void and of no effect and all rights and obligations of the parties hereunder shall terminate without liability on the part of any party hereunder; *provided, however* that the termination of this Agreement shall not relieve either party of any liability for breach or fraud under this Agreement that occurred prior to the date of termination, and *provided further* that notwithstanding anything contained herein to the contrary, **ARTICLE 1** (Definitions; Interpretation), Section 6.1 (Confidentiality), this Section 12.3, and **ARTICLE 13** (Miscellaneous) shall survive any termination of this Agreement.
- Specific Performance. The parties hereto acknowledge and agree that the parties hereto would be irreparably damaged if any of the provisions of this Agreement are not performed in accordance with their specific terms or are otherwise breached and that any non-performance or breach of this Agreement by any party hereto could not be adequately compensated by monetary damages alone and that the parties hereto would not have any adequate remedy at law. Accordingly, in addition to any other right or remedy to which any party hereto may be entitled, at law or in equity (including monetary damages), such party shall be entitled to enforce any provision of this Agreement by a decree of specific performance and to temporary, preliminary and permanent injunctive relief, subject to obtaining any required Governmental Consents, to prevent breaches or threatened breaches of any of the provisions of this Agreement without posting any bond or other undertaking. Without limiting the generality of the foregoing, the parties hereto agree that the party seeking specific performance shall be entitled to enforce specifically (a) a party's obligations under Section 2.5 and (b) a party's obligation to consummate the transactions contemplated by this Agreement (including the obligation to consummate the Closing and pay the Purchase Price), if the conditions set forth in ARTICLE 7 or ARTICLE 8, as applicable, have been satisfied (other than those conditions that by their nature are to be satisfied at the Closing) or waived, without the requirement of the prevailing party to post a bond. In addition to the foregoing, the prevailing party enforcing its rights under this Agreement or any Ancillary Document shall be entitled to prompt payment on demand from the other party of the reasonable attorneys' fees and costs incurred by the prevailing party in enforcing such rights.

ARTICLE 13

MISCELLANEOUS

- responsible for all costs and expenses incurred by it in connection with the negotiation, preparation and performance of and compliance with the terms of this Agreement. For the avoidance of doubt, all expenses incurred by the Acquired Companies on or prior to the Closing Date in connection with the negotiation, preparation and performance of and compliance with the terms of this Agreement and the Ancillary Documents shall be paid by Seller. All governmental fees and charges applicable to any requests for Governmental Consents shall be paid one-half by Seller and one-half by Buyer. Each party is responsible for any commission, brokerage fee, advisory fee or other similar payment that arises as a result of any agreement or action of it or any party acting on its behalf in connection with this Agreement or the transactions contemplated hereby.
- **13.2 Further Assurances**. After the Closing, each party shall from time to time, at the request of and without further cost or expense to the other, execute and deliver such other instruments of conveyance and assumption and take such other actions as may reasonably be necessary in order to more effectively consummate the transactions contemplated hereby.
- Assignment. Neither party may assign this Agreement without the prior written consent of the other party hereto, provided, however, that Buyer may assign its rights hereunder to an Affiliate of Buyer upon written notice to Seller and Buyer may designate Scripps Broadcasting Holdings, LLC, a Nevada limited liability company and wholly-owned subsidiary of Buyer ("SBH LLC"), as the acquirer of the FCC Licenses under the FCC Applications or otherwise effect an assignment of the FCC Licenses to Scripps Broadcasting Holdings, LLC effective as of the Closing or thereafter (and Seller agrees to cooperate with Buyer in pursuing the FCC's consent to this assignment of the FCC Licenses to SBH LLC so that such assignment may be completed at the Closing), provided that (i) any such assignment or designation does not delay the processing of the FCC Applications, the grant of the FCC Consent, the HSR Clearance or the Closing, (ii) any such assignee of Buyer's rights under this Agreement shall deliver to Seller a written instrument of assumption with respect to this Agreement pursuant to which such assignee shall (x) make to Seller the representations and warranties contained in ARTICLE 4 with respect to such assignee and (y) covenant to Seller to observe, satisfy, discharge and perform the covenants of Buyer set forth in this Agreement and (iii) Buyer shall remain liable for all of its obligations hereunder (including those assigned to such assignee), including the payment of the Purchase Price pursuant to Section 2.2 hereof. Notwithstanding anything to the contrary herein, Buyer shall be permitted to collaterally assign its rights and remedies under this Agreement and the other Ancillary Documents to any Debt Financing Source in respect of the Debt Financing, but no such assignment shall relieve any such assignor of its obligations under this Agreement or any of the Ancillary Documents. The terms of this Agreement shall bind and inure to the benefit of the parties' respective successors and any permitted assigns, and no assignment shall relieve any party of any obligation or liability under this Agreement.
- **13.4 Notices**. Any notice pursuant to this Agreement shall be in writing and shall be deemed properly delivered, given and received (a) when delivered by hand, by registered mail, by courier or express delivery service or (b) upon confirmation of receipt (other than an automatically-generated confirmation) when sent by electronic mail to the address or email address, as applicable, set forth beneath the name of such party below (or to such other address or email address, as applicable, as such party shall have specified in a written notice in accordance with this Section 13.4 given to the other parties hereto):

if to Seller:

Cordillera Communications, LLC c/o Evening Post Industries, Inc. 134 Columbus Street Charleston, South Carolina 29403 Attention: Joe Waring Email: jwaring@eveningpostindustries.com

with a copy (which shall not constitute notice) to:

Cooley LLP

1299 Pennsylvania Ave., NW

Suite 700

Washington, DC 20004 Attention: Michael D. Basile Email: mdbasille@cooley.com

if to Buyer: Scripps Media, Inc.

312 Walnut Street, 28th Floor

Cincinnati, OH 45202

Attention: (i) William Appleton, Executive Vice President and General Counsel and (ii) Robin Davis, Vice President/Strategy and Corporate

Development

Email: (i) appleton@scripps.com and (ii)

robin.davis@scripps.com

with a copy (which shall not constitute notice) to:

Baker & Hostetler LLP

312 Walnut Street, Suite 3200

Cincinnati, OH

Attention: Rob Morwood

Email: rmorwood@bakerlaw.com

- 13.5 Amendments; Waivers. No amendment or waiver of compliance with any provision hereof or consent pursuant to this Agreement shall be effective unless evidenced by an instrument in writing signed, in the case of an amendment, by Buyer and Seller, or in the case of a waiver or consent, by the party against whom the waiver or consent is to be effective. No waiver by any party of any breach or violation of, default under or inaccuracy in any representation, warranty or covenant hereunder, whether intentional or not, will be deemed to extend to any prior or subsequent breach or violation of, default under, or inaccuracy in, any such representation, warranty or covenant hereunder or affect in any way any rights arising by virtue of any prior or subsequent such occurrence. No delay or omission on the part of any party in exercising any right, power or remedy under this Agreement will operate as a waiver thereof. Notwithstanding anything to the contrary herein, none of the Debt Financing Provisions may be amended, modified or waived in a manner adverse to the Debt Financing Sources without the prior consent of the Debt Financing Sources.
- 13.6 Severability. If any Governmental Entity holds any provision in this Agreement invalid, illegal or unenforceable as applied to any party or to any circumstance under any applicable Law, such invalidity, illegality or unenforceability shall not affect any other provision or part of a provision of this Agreement, but this Agreement shall be reformed and construed as if such invalid or illegal or unenforceable provision or part of a provision had never been contained herein and such provision or part shall be reformed so that it would be valid, legal and enforceable to the maximum extent permitted by applicable Law; provided that any such reform or construction does not affect the economic or legal substance of this Agreement and the transactions contemplated hereby in a manner adverse to either party and, if any such reform or construction does affect the economic or legal substance of this Agreement and the transactions contemplated hereby in a manner adverse to either party, the parties shall negotiate in good faith a replacement provision for such invalid, illegal or unenforceable provision which shall accomplish the original intention of the parties with respect to such provision to the greatest extent practicable.
- **13.7 No Beneficiaries**. Except as provided in Section 6.11, **ARTICLE 10** and **ARTICLE 11**, nothing in this Agreement expressed or implied is intended or shall be construed to give any rights to any Person other than the parties hereto and their successors and permitted assigns. Notwithstanding the

foregoing, the Debt Financing Sources shall be third-party beneficiaries of each of the Debt Financing Provisions.

13.8 Governing Law; Consent to Jurisdiction; Waiver of Jury Trial.

- (a) This Agreement and the negotiation, execution, performance or nonperformance, interpretation, termination, construction and all matters based upon, arising out of or related to this Agreement, whether arising at law or in equity (collectively, the "Covered Matters"), and all claims or causes of action (whether in contract or tort) that may be based upon, arise out of or relate to the Covered Matters, except for documents, agreements and instruments that specify otherwise, shall be governed by the laws of the State of Delaware without giving effect to the choice of law provisions thereof.
- (b) All Actions arising out of or relating to this Agreement shall be heard and determined exclusively in the Chancery Court of the State of Delaware or federal courts of the United States of America for the District of Delaware, to the extent the Chancery Court of the State of Delaware does not have jurisdiction over any such Action, and the parties hereto hereby irrevocably submit to the exclusive jurisdiction of such courts (and, in the case of appeals, appropriate appellate courts therefrom) in any such Action and irrevocably waive the defense of an inconvenient forum to the maintenance of any such Action. The parties hereto agree that a final judgment in any such Action shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by applicable law. The consents to jurisdiction set forth in this Section 13.8 shall not constitute general consents to service of process in the State of Delaware, shall have no effect for any purpose except as provided in this Section 13.8 and shall not be deemed to confer rights on any third party.
- (c) Notwithstanding the foregoing Section 13.8(a) or Section 13.8(b), each of the parties agrees that it will not bring or support any Action of any kind or description, whether in law or in equity, whether in contract or in tort or otherwise, against the Debt Financing Sources in any way relating to this Agreement or the Ancillary Documents or any of the transactions contemplated hereby or thereby, including but not limited to any dispute arising out of or relating in any way to the Debt Financing or the performance of the transactions related thereto, in any forum other than the Supreme Court of the State of New York, County of New York, or, if under applicable law exclusive jurisdiction is vested in the Federal courts, the United States District Court for the Southern District of New York located in the County of New York (and appellate courts thereof), and makes the agreements, waivers and consents set forth in Section 13.8(b) *mutatis mutandis* but with respect to the courts specified in this Section 13.8(c).
- (d) BUYER AND SELLER HEREBY IRREVOCABLY WAIVE ALL RIGHT TO TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM (WHETHER BASED ON CONTRACT, TORT OR OTHERWISE) ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE ACTIONS OF BUYER OR SELLER IN THE NEGOTIATION, ADMINISTRATION, PERFORMANCE AND ENFORCEMENT HEREOF, INCLUDING ANY ACTION AGAINST THE DEBT FINANCING SOURCES ARISING OUT OF, OR RELATING TO, THE TRANSACTIONS CONTEMPLATED HEREBY OR THE DEBT FINANCING.
- **13.9 Neutral Construction**. The parties hereto agree that this Agreement was negotiated at arms-length and that the final terms hereof are the product of the parties' negotiations. This Agreement shall be deemed to have been jointly and equally drafted by all such parties, and the provisions hereof should not be construed against a party on the grounds that the party drafted or was more responsible for drafting the provision.
- 13.10 Non-Recourse Against Debt Financing Sources; Waiver of Certain Claims. Notwithstanding anything to the contrary contained in this Agreement, the parties hereto hereby agree that no Debt Financing Source shall have any liability to Seller or any of its Affiliates or any other Person relating to or arising out of this Agreement, any of the Ancillary Documents or the Debt Financing (including any claim for any Losses suffered as a result of any breach of this Agreement, any of the Ancillary Documents or the terms of the Debt Financing), or any document related thereto (including any

willful breach thereof), or the failure of the transactions contemplated hereby or thereby to be consummated), whether at law or equity, in contract or in tort or otherwise, and neither Seller nor any of its Affiliates shall have any rights or claims whatsoever against any of the Debt Financing Sources under this Agreement, any of the Ancillary Documents or in connection with the Debt Financing, whether at law or equity, in contract or in tort, or otherwise. Seller hereby agrees, on behalf of itself and its Affiliates, that none of the Debt Financing Sources shall have any liability or obligations to Seller or any of its Affiliates relating to this Agreement, any of the Ancillary Documents or any of the transactions contemplated herein or therein (including with respect to the Debt Financing). Seller and its Affiliates hereby waive any and all claims and causes of action (whether at law, in equity, in contract, in tort or otherwise) against the Debt Financing Sources that may be based upon, arise out of or relate to this Agreement, any of the Ancillary Documents, any financing commitment or the transactions contemplated hereby or thereby (including the Debt Financing). This Section 13.10 is intended to benefit and may be enforced by the Debt Financing Sources and shall be binding on all successors and assigns of Seller.

13.11 Counterparts; Delivery by Facsimile/Email. This Agreement may be executed in separate counterparts, each of which will be deemed an original and all of which together will constitute one and the same agreement. This Agreement, the Ancillary Documents, and each other agreement or instrument entered into in connection herewith or therewith or contemplated hereby or thereby, and any amendments hereto or thereto, to the extent signed and delivered by facsimile transmission or electronic mail in pdf form, shall be treated in all manner and respects as an original agreement or instrument and shall be considered to have the same binding legal effect as if it were the original signed version thereof delivered in person. At the request of any party hereto or any party to any such agreement or instrument, each other party hereto or thereto shall re-execute original forms thereof and deliver them to all other parties. No party hereto or to any such agreement or instrument shall raise the use of a facsimile machine or electronic mail to deliver a signature or the fact that any signature or agreement or instrument was transmitted or communicated through the use of a facsimile machine or electronic mail as a defense to the formation or enforceability of a contract, and each such party forever waives any such defense.

13.12 Entire Agreement. The Schedules and Exhibits hereto are hereby incorporated into this Agreement. This Agreement, the Schedules and Exhibits, and the Ancillary Documents constitute the entire agreement and understanding among the parties hereto with respect to the subject matter hereof, and supersede all prior agreements and understandings with respect to the subject matter hereof, except the NDA, which, subject to Section 6.1, shall remain in full force and effect in accordance with its terms. No party makes any representation or warranty with respect to the transactions contemplated by this Agreement except as expressly set forth in this Agreement (or in any of the Ancillary Documents, or any other agreement executed on the date hereof or thereof in connection herewith or therewith).

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

In WITNESS WHEREOF, the parties have executed this Purchase Agreement as of the date set forth above.

CORDILLERA COMMUNICATIONS, LLC

Name:

Terrance F. Hurley

Title: President

SCRIPPS MEDIA, INC.

Name:

William Appleton

Title:

Executive Vice President and General Counsel

Section 6.11 of this Purchase Agreement is hereby acknowledged and agreed to by EPI Group, LLC.

EPI GROUP, LLC

Name:

John P. Barnwell

Title:

President

IN WITNESS WHEREOF, the parties have executed this Purchase Agreement as of the date set forth above.

CORDILLERA COMMUNICATIONS, LLC

Name:

Terrance F. Hurley

Title:

President

SCRIPPS MEDIA, INC.

Name:

William Appleton

Title:

Executive Vice President and General Counsel

Section 6.11 of this Purchase Agreement is hereby acknowledged and agreed to by EPI Group, LLC.

EPI GROUP, LLC

Name:

John P. Barnwell

Title:

President

IN WITNESS WHEREOF, the parties have executed this Purchase Agreement as of the date set forth above.

CORDILLERA COMMUNICATIONS, LLC

Name:

Terrance F. Hurley

Title:

President

SCRIPPS MEDIA, INC.

Name:

William Appleton

Title:

Executive Vice President and General Counsel

Section 6.11 of this Purchase Agreement is hereby acknowledged and agreed to by EPI Group, LLC.

EPI GROUP, LLC

Name: John P. Barnw

Title: Rresident

Schedule 3.6 Tangible Personal Property

(a)

WLEX Communications, LLC:

2ND INSTALLMENT - TRANSMITTER BUILDING

25,000 SQ. FT. METAL SIDED STUDIO BUILDING ON POURED CONCRETE

PRODUCTION SWITCHER

NEXIO SERVER FOR PLAYBACK

Live truck equipment

Centralized PSIP System

Generator Electrical wiring and pad

Carpet for building

Audio Board

CHANNEL 36 BLDG ADDITION FOR TRANSMITTER

(14) PANASONIC FIELD CAMERAS FOR NEWS

EQUIPMENT IN NISSAN LIVE TRUCK

GENERATOR AT WLEX FACITILY

TRICASTER SWITCHER FOR STUDIO B

NEW SET / LIGHTING DESIGN

2017 CHEVY TRAVERSE SATELLITE VEHICLE

NEW GRAPHICS PKG 2017

NEWS SET, MONITORS & VIDEO KIT

CELL PACK

NEWS SET DESIGN

Grass Valley Stratus

KSBY Communications, LLC

ADC100 Master Control Automation System (Hardware/Software)

Platinum Routing System (Frame, I/O Cards, IconMaster Switchers, Control Panels, Software)

Selenio Primary Encoder/Multiplexer

STL/TSL Redundant Microwave System

Nexio Master Control Record/Play Servers & Storage Array

Vision 3ME Production Switcher System

OverDrive News Automation System (Hardware/Software)

XPression Character Generator System (Hardware/Software)

News Record/Play Servers & Storage Array

News Velocity Edit System

ENPS Newsroom Control System (Hardware/Software)

Studio Camera System (camera, lens, CCU, prompter, pan/tilt head, pedestal)

WSI Weather Computer System (Hardware/Software)

Studio News Set

Studio Lighting Fixtures

Telephone PBX & Voicemail System

SigmaCD Primary Transmitter/Mask Filter

Primary antenna/Feedline

EMCEE Standby Transmitter/Mask Filter

Primary 400-foot tower at Cuesta Peak

Microwave tower at San Antonio Peak

ENG Mobile Ford E-350 Microwave Live Van & Equipment 2013

DXL068T1SA /DXL068DR1SA Translator Microwave Upgrade

(2) OnCall Satellite Units (mounted on news vehicles)

(3) Dejero Engo Transmitters (live remote backpack equipment)

KTVQ Communications, LLC:

NEWS SET CONSTRUCTION

NEXIO AMP 4 CH SD/HD PLAYOUT SERVER

OVERDRIVE NEWS AUTOMATION

CARBONITE EXTREME 64X64

NEWS SET LIGHTING PACKAGE

News Cameras

Xpression Graphics System

News Playout System

Billings Transmitter, KTVQ

Ross Inception

KBZK Communications, LLC

NEWS SET

CARBONITE, REDUNDANT POWER, & COMMISSIONING

HD XPRESSION CG

HD STUDIO CAMERAS

NEXIO NEWS SERVER UPGRADE

NEWS TICKER

HEARTLAND MICROWAVE

KPAX Communications, LLC:

CARPORT

KALISPELL DIGITAL MICROWAVE

KALISPELL DIGITAL TRANSMITTER

SELENIO ENCODER

NEWS SET

NEXIO AMP SERVERS FOR MASTER CONTROL (2)

Ross HD switcher hardware

Generator-TV Mtn

TV Mountain Transmitter

Satellite Truck

Production Trailer

CBS Satellite Dish

News Non-Linear Editing

Ross Inception (Note: KPAX Communications, LLC pays \$1,600 per month for four years. At the end of four years KPAX Communications, LLC will own the equipment)

KXLF Communications, LLC:

MRC DXL5000 TRANSMITTER & RECEIVER FOR MICROWAVE TO HIGH FLAT

CW SELENIO ENCODER

NEXIO FARAD CORE & 24-PORT SWITCHES (2)

ON CALL SAT SYSTEM

XL HEIGHTS GENERATOR

ROSS INCEPTION

Sangre De Cristo Communications, LLC:

Barons 60' Weather Tower, shed/building and 2000 gallon underground gas tank

HD MICROWAVE, 1 transmitter, 3 translators

NEWS ARCHIVE SYSTEM

TELEPHONE SYSTEM - PUEBLO

News Live Truck Equipment

Colorado Springs and Pueblo news sets

Video server

News set structure

News set lighting fixtures

Satellite dishes

Ross video board & automation

WSI HW Distribution amplifier/repeater

SPOT AND PROGRAM DISC ARRAY

Mastercontrol ADC-Automation equipment

satellite truck, 3 live trucks, 17x news/eng vehicles

IP Satellite Dish w/ Uplink system (SUV)

4 generators

Air conditioning

Elevator

Mask filter

Set lighting

KRIS Communications, LLC:

NEWS SET

CUBICLES

ELEVATOR

LAMINATE PLASTIC WORK STATION

KZTV REMODEL FURNITURE

MULTICHANNEL SERVER PLUS MASTER CONTROL

NEW TRANSFER SWITCH & MOVE GENERATOR

HD CONVERSIONS

KDF DIGITAL

Microwaves

OverDrive hardware

Digital STL & TSL microwave

2 Live trucks conversion & equipment

Kingston 8GB Kit DIMM Dual Rank x4 Memory

KRTV Communications, LLC:

KRTV Master Control HD NEXIO SERVER

KRTV HD ROUTER

KRTV NEWS SET

Inception Newsroom Computer system (both stations)

KTGF MASTERCNTRL & ENCODR SYST

KTVH Transmitter

KTVH Master Control Grass Valley System

KRTV TRANSMITTER

Weather System

KTVH NEWS SET

KRTV Nexio News System & Software

KTVH HD News- Carbonite

KTVH JVC STUDIO Cameras

KTVH I-T & PHONE SYSTEM

EXPRESSION NEWS GFX SYSTEM (KRTV & KTVH)

KRTV AIR CONDITIONING UNIT

KRTV HD NEWS CARBONITE

KXLH TRANSMITTER

KTGF TRANSMITTER

KRTV PANASONIC STUDIO CAMERAS

KATC Communications, LLC:

CG REPLACEMENT

3 HD STUDIO CAMERAS

HD SWITCHER UPGRADE

CMC Remodel Construction & Design

Administrative remodel office furniture

Station renovation design services & construction

JVC GYHM890U Pro HD camcorders (6)

On Call Sat - Ku 1.0m Sat Equipment

Computer Upgrade 62 ea (HP: 50 Dell: 11 Samsung: 1)

CHYRON & CAMEO REPLACEMENT

Cordillera Interactive, LLC:

KRIS D3 Equipment

KATC D3 Equipment

WLEX D3 Equipment

KVOA D3 Equipment

KOAA D3 Equipment

KSBY D3 receiver

KPAX D3 receiver

KTVQ D3 receiver

KXLF D3 receiver

KRTV D3 receiver

Video equipment

Video equipment

The towers, buildings, dishes, microwaves, dopplers and transmitters referenced in Schedule 3.7(a) are incorporated by reference herein.

(b)

None.

(c)

Sangre De Cristo Communications, LLC is replacing its tower at Cheyenne Mountain, but is awaiting FAA approval.

The Kaplan Site was damaged due to a plane crash on Friday August 31, 2018. The tower fell on the transmitter building and all equipment was destroyed as a result.

KATZ Communications, LLC has a broken doppler, which Seller is in process of repairing.

KVOA had a signal analyzer go out.

The KSBY Tower Issue described in Schedule 6.19 is incorporated by reference herein.

(d)

KPAX Communications, LLC has a tower at Kalispell, MT, previously used for KAJJ's analog operations, that the Stations no longer use. The internet provider for KPAX Communications, LLC, Montana Digital, took over the lease of such tower from the landowner. There have been very preliminary discussions about transferring ownership of the tower to Montana Digital.

Site Lease Agreement, dated as of January 23, 2003, by and between WLEX Communications, LLC and Lexington-Fayette Urban County Government

Lease Agreement, dated November 16, 2015, by and between WLEX Communications, LLC and North Nicholasville Road Property Owners Association

Lease Agreement, dated February 16, 2010, by and between WLEX Communications, LLC and Christian Broadcasting System, Ltd.

Sparks Refrigeration rents tower space from WLEX Communications, LLC month-to-month, but WLEX Communications, LLC is in the process of negotiating a new lease with them.

Tower Lease Agreement, dated April 8, 1985, by and between KATC Communications, LLC and KAJN Agape Broadcaster, as amended by the Tower Lease Agreement dated November 16, 2015.

Schedule 3.7 Real Property; Real Property Leases

(a)

Owner: WLEX Communications, LLC

1065 Russell Cave Road Lexington, KY

WLEX Communications, LLC has easements with Lexmark for the anchors of guy wires that are on Lexmark's property.

Owner: KSBY Communications, LLC

Main Studio 1772 Calle Joaquin San Luis Obispo, CA 93405

Transmitter Site Building and Tower TV Tower Road San Luis Obispo, CA 93401

Building and Tower Black Ranch Road Casmalia, CA 93429

Owner: KTVQ Communications, LLC

Studio Site and Tower 3203 3rd Ave N Billings, MT

Billings Transmitter Site 1700 Colburn Road Billings, MT

Columbus Microwave Building and Tower Lone Tree Road Columbus, MT

Bozeman Pass Microwave Site Lightning Strike Road Livingston, MT

Scott Cattle Co. Microwave Site Old Hardin Road

Billings, MT

High Flat Microwave Site High Flat Road Bozeman, MT

Sheridan Translator Building and Tower Forest Road 318 Sheridan, WY

Miles City Translator Site Unnamed Road Miles City, MT

Hardin Translator Site Dry Creek Road Hardin, MT

Columbus Translator Building and Tower Unnamed Road, Columbus, MT

Sarpy Translator Building and Tower Unnamed Road Colstrip, MT

Owner: KCTZ Communications, LLC

KBZK Studio 90 Television Way Bozeman, MT

KBZK Transmitter Building Green Mtn Bozeman, MT

KBZK Microwave Site High Flat Bozeman, MT

Owner: KPAX Communications, LLC

Office/Sales Building 1049 W. Central Ave. Missoula, MT 59801

Missoula Studio

2204 Regent St. Missoula, MT 59801

Missoula Transmitter Site & Tower 4 miles east of Evaro Missoula, MT 59808

Microwave Site Georgetown Mountain Georgetown, MT 59711

Tower Georgetown Mountain Georgetown, MT 59711

Kalispell Transmitter Blacktail Mt Lakeside, MT 59922

Owner: KXLF Communications, LLC

KXLF Studio 1003 South Montana Butte, MT

KXLF Transmitter Site and Building XL Heights Butte, MT

KXLF Microwave Building Jack Mtn Basin, MT

KXLF Microwave Building SawTell Peak Island Park. ID

KXLF Road to Transmission Site
Township 3 North, Range 7 West, P.M.M.
All of the Blue Copper, Survey No. 7938
All of Copper Mountain, Survey No. 6338
An undivided 17/32 interest in and to the Davenport, survey No. 6339
All of the Green Copper, Survey No. 7937
An undivided 1/4 interest in and to the Gypsy, Survey No. 8386
An undivided 1/3 interest in and to the Maybelle, Survey No. 6337
An undivided 3/5 interest in and to the McCloskey, Survey No. 6926

An undivided 3/5 interest in and to the Ontario, Survey No. 6927

An undivided 1/2 interest in and to the Spokane, Survey No. 5801

All of the Walk Over, Survey No. 7935

All of the Spokane, Survey No. 5801

All of the McCloskey, Survey No. 6926

All of the Davenport, Survey No. 6339

All of the Ontario, Survey No. 6927

All of the Maybelle, Survey No. 6337

All of the Copper Mountain, Survey No. 6338

All of the Stem winder Survey No. 7523

All of the Black Horse, Survey No. 6663

All of the Maggie Survey No. 2239

All of the Cold Spring, Survey No. 2987

All of the Fairview, Survey No. 2980

All of the King Soloman, Survey No. 1685

All of the Divide, Survey No. 7378

All of the Anona, Survey No. 7907

All of the Ada, Survey No. 7908

All of the Usov, Survey No. 7722

All of the Limit, Survey No. 7725

All of the Siz, Survey No. 8200

All of the Monte Christo, Survey No. 6545

All of the Mary Ellen, Survey No. 2819

All of the Elizabeth, Survey No. 6473

All of the Summit, Survey No. 1570

All of the Fanney Scott, Survey No. 5748

All of the Jupiter, Survey No. 7180

All of the Chief Justice, Survey No. 6928

All of the Kopper Moone No. 2, Survey No. 7140

All of the Arcede, Survey No. 6027

All of the Columbus, Survey No. 6026

Owner: Sangre De Cristo Communications, LLC

News Bureau 2200 7th Avenue Pueblo, CO 81003

Broadcasting Facility 5520 Tech Center Drive Colorado Springs, CO 80919

Land 5530 Tech Center Drive Colorado Springs, CO 80919 Microwave Receivers and Transmitter Site Building and Tower Cheyenne Mt. (c.30-6130) and (dtv-6236) Colorado Springs, CO 80909

Barons Radar Tower, Dish, Transmitter 3200 Baculite Mesa Pueblo, CO 81008

Owner: KRIS Communications, LLC

KRIS Studio/ Tower and Antenna 301 Artesian St Corpus Christi, TX 78401

KRIS Transmitter Building and Tower 3273 CR61 Robstown, TX 78380

Transmitter Building and Tower HWY 222 18 Miles East of Beeville Blanconia, TX 78389

Transmitter Building and Tower HWY 70 south of FM70 Bishop, TX 78364

Corpus Transmitter Building and Tower 4518 SPID Corpus Christi, TX 78412

Owner: KRTV Communications, LLC

KRTV Studio 3300 Old Havre Hwy Black Eagle, MT 59414

Sullivan Hill Microwave Mission Road Cascade MT

Sweet Grass Hills Translator Sweet Grass Hills, MT

Jack Mountain Building Mountain Top Basin, MT Cement Hill Translator Mountain Top Haver, MT

Wirth Ranch Microwave Mountain Top Wolf Creek, MT

Wirth Ranch Building and Tower Wolf Creek, MT

Peter Kuhr Ranch, Building and Tower Havre, MT

KTVH Studio 100 West Lyndale Ave Helena, MT 59601

Hogback Mountain Building and Tower Hogback Mountain Helena MT

Owner: KATC Communications, LLC.

KATC Studio and Doppler 1103 Eraste Landry Rd. Lafayette, LA 70506

Kaplan Transmitter Tower and Equipment Site (the "*Kaplan Site*") 7740 Rosewood Drive Kaplan, LA 70548

DTV Tower and Transmitter Acadia Parish (Section 37, Township 8 South, Range 2 East 101.93 acres) Branch, LA 70525

Vacant Land Pendelton Harbor, Lot #957, Unit 9 Hemphill, TX

(b)

WLEX Communications, LLC:

Generator Access and Use Agreement, by and between WTVQ-TV, LLC and WLEX Communications, LLC, as supplemented by the Tower and Building Space License Agreement,

dated as of June 6, 2005 by and between Media General Operations, Inc. and WLEX Communications, LLC (WTVQ Tower and Building)

KSBY Communications, LLC:

Communications Use Lease, dated as of February 3, 1997, by and between KSBY Communications, LLC and the United States of America (Cuesta Peak Communications Site) (the "Cuesta Peak Permit"), which is currently expired, but KSBY Communications, LLC is under negotiation with the U.S. Forest Service for a new agreement with respect to the Cuesta Peak Permit.

KSBY Communications, LLC receives a bill each year from the U.S. Forest Service for a use permit for a roadway.

Santa Ynez Peak Site Use Agreement, dated as of July 1, 1991, by and among KSBY Communications, LLC, Comsite Pacific Inc. and Bud-North Shore Communications (Santa Ynez Peak, Santa Barbara, CA 93105)

Short Form Tower License Agreement, dated as of December 8, 2009, by and between KSBY Communications, LLC and Crown Castle MU LLC (Broadcast Peak, Santa Barbara, CA 93105).

Lease Agreement, dated November 1, 2009, by and between Ruthanne Tompkins, Trustee of the Tompkins Survivor's Trust, and KSBY Communications, Inc. (Land at Black Ranch Road, Santa Barbara, CA 93429)

Bud-North Shore Sub Lease, dated March 1, 1990, between Bud-North Short Communications and KSBY-TV and letter dated August 18, 2000 from Community Radio, Inc. and signed by John P. Franklin to NVG-Santa Barbara, LLC. (Gibraltar Peak, Santa Barbara, CA 93105)

Trade Agreement, dated January 1, 2016, by and between Atrium Ventures and KSBY. (731 South Lincoln Street, Santa Barbara, CA 93458)

KTVQ Communications, LLC:

Lease Agreement, dated as of 2004, by and between Yellowstone Bighorn Research Association, Box 630, Red Lodge, MT and KTVQ Communications, Inc. (Red Lodge Translator Site at Howell Gultch Road, Red Lodge, MT)

Option to Lease and Lease Agreement, dated as of February 11, 1981, by and between Mary Lou Lorash and Charles W. Lorash (as transferred to Duncan Campbell) and Garryowen Corporation (Land at Columbus Microwave Site at Lone Tree Road at Columbus, MT)

Communications Use Lease, dated September 29, 2016, between KTVQ Communications, Inc. (as Lessee) and United States of America (Land at Sheridan Translator Site at Forest Road 318 in Sheridan. WY)

Lease Agreement between KTVQ Communications, Inc. (as Lessee) and Lonnie Peterson and Helen Peterson, dated 2004 (Land at Columbus Translator Site at the unnnamed road, 1 mile S of Columbus, MT)

KTVQ Communications, LLC, as Lessee, has a no cost agreement with respect to the Sarpy Translator Site at the unnamed road, 35 miles ENE of Hardin, Colstrip, MT with Northwestern Energy

KTVQ Communications, LLC, as Lessee, has an oral, no cost agreement with respect to the Cedar Mountain Translator Site at Spirit Mountain Road in Cody, WY with Cedar Mountain Properties, LLC subject to a Roadway Use Agreement, dated as of June 20, 2012, by and between KTVQ Communications, Inc. and Elbert O. Sowerwine.

KTVQ Communications, LLC, as Lessee, has an oral, no cost agreement with respect to the Spring Creek Translator Site at the unnamed road, 3.9 miles NNE of Meeteetse, WY and the McCullough Peak Translator Site at McCullough Peaks Road in Cody, WY with Park County, Wyoming.

Commercial Lease Agreement, dated June 25, 1986, between KCTZ Communications, Inc. and North Pass, LLC

KPAX Communications, LLC:

Lease, dated as of May 10, 2017, by and between the United States of America and KPAX Communications, LLC (Land for Missoula Transmitter at 4 Miles East of Evaro, Missoula, MT 59808)

Tower Lease Agreement, dated as of March 27, 2014, by and between Bresnan Communications, LLC and KPAX Communications, Inc. (Microwave Site at Blacktail Mt, Lakeside, MT 59922)

Special Use Permit, dated as of February 18, 2011, by and between U.S. Department of Agriculture, Forest Service, and KPAX Communications, Inc. and Lease, dated May 1, 2010, by and between KPAX Communications, Inc. and the United States of America, Department of Energy (Land for Kalispell Transmitter Site at Blacktail Mt, Lakeside, MT 59922)

Lease, dated as of October 23, 2012, by and between Angel Point Capital LLC and KPAX Communications, Inc. (Kalispell Studio at 301 First Ave E, Kalispell, MT 59901)

KXLF Communications, LLC

Commercial Lease Agreement, dated June 25, 1996, by and between Green Mountain Grazing Association and Bee Broadcasting Associates (Transmitter Site Land at Green Mtn 8 miles East of Bozeman, Bozeman, MT)

Lease between KXLF Communications Inc. and the United States of America, acting through the Forest Services, Department of Agriculture, dated January 29, 2010 (SawTell Peak, Island Park, ID)

U.S. Government Lease for Real Property, dated as of May 1, 1979, by and between the United States of America and KXLF

KXLF Communications, Inc. has a month-to-month lease with the Montana Economic Revitalization and Development Institute with respect to the Thornton Building in Butte, MT.

Sangre de Cristo Communications, LLC

Lease, dated February 1, 2018, by and between Barbara G. Griffin and Sangre de Cristo Communications, LLC (6151 Lake Gulch Rd, Castle Rock, CO 80104)

Agreement for Easement, dated January 1, 1995, between the University of Southern Colorado Foundation and Sangre de Cristo Communications, Inc. (3200 Baculite Mesa, Pueblo, CO 81008)

Lease, dated as of October 12, 2005, between Cheyenne Propagation Co. and Sangre De Cristo Communications, Inc. (Cheyenne Mt.(dtv-6236), El Paso, CO 80909, 6236 Transmitter Lane Colorado Springs CO 80909)

Lease, dated as of September 1, 2010, between Cheyenne Propagation Co. and Sangre De Cristo Communications, Inc (Cheyenne Mt.(c.30-6130), El Paso, CO 80909, 6130 Transmitter Lane Colorado Springs CO 80909)

License Agreement, dated September 30, 2013, by and between American Towers LLC and Sangre de Cristo Communications, Inc. (5955 Trout Creek Rd., Woodland Park, CO 80863)

Sangre de Cristo Communications, LLC has an oral agreement with Tegna, Inc. for use of land at KUSA 500 Speer Blvd., Denver, CO where its Stations' microwave transmitter and dish are located and Lookout Mountain in Golden, CO where its Stations' microwave receiver and transmitter and two dishes are located, in each case for no charge.

Lease Agreement between Educational Communications of Colorado Springs, Inc. d/b/a KTLF-FM and KOAA-TV, dated September 8, 2000 (TV Hill Evans Dr. South, Fremont, CO 81290)

KRIS Communications, LLCKRIS Transmitter 3273 CR61
Partition Agreement, dated as of December 30, 2015, by and between John Morgan O'Brien III, Bridget N. Gayle, James R. O'Brien, Michael F. O'Brien and Mary K. Ball (Blanconia Transmitter Site- HWY 222 18 Miles East of Beeville)

Contract of Lease, dated February 13, 1980, by and between Cecil R. Patrick and Cismek Corporation (Bishop Transmitter Site- HWY 70 south of FM70 near Bishop TX)

Lease Agreement, dated as of September 15, 1992, by and between Texas Gun Shop Inc. and KDF-TV (Corpus Transmitter Site- 4518 SPID)

Antenna Site License Agreement, dated May 10, 1999, by and between TGFW, L.P., Shoreline Plaza, Inc. and KRIS Communications (One Shoreline Plaza- 800 N. Shoreline)

Real Estate Lease, dated October 1, 1998, by and between T. Frank Smith, Jr. and KRIS-TV (Harrison Building- 511 S. Staples)

KRTV Communications, LLC:

Lewistown Tower Site Lease Agreement between Hi-Line Radio Fellowship, Inc. and KRTV Communications, Inc., dated July 1, 2007 (North West of Lewistown, Lewistown, MT)

Ground Lease and Easement Agreement between Phillip R. Wirth, Robert N. Wirth, Josephine M. Wirth and KRTV Communications, Inc., dated October 1, 2010 (Mountain Top, Wolf Creek, MT)

KRTV Communications, LLC has an oral, no cost lease for the translator in a community-owned building in Chinook, MT.

KRTV Communications, LLC has a reciprocal oral lease with Jim Stiles, who has equipment in their Sweetgrass Hills and Sullivan Hills sites, for the translator in West Knees, MT.

Lease Agreement between Peter Kuhr and KRTV Communications, Inc., dated October 10, 2013 (Land at Mountain Top, Haver, MT)

Lease between KRTV Communications Inc. and the United States of America, acting through the Bureau of Land Management, Department of the Interior, dated August 26, 2010 (Sweet Grass Hills, Joplin, MT)

Lease Agreement between CCR-Helena III, LLC and KRTV Communications, LLC, dated January 1, 2016 (Microwave Building and Tower at Mountain Top, Mary's Ville, MT)

Lease between Carroll College and Beartooth Communications Company, dated September 23, 1997 (Land for KTVH Studio at 100 West Lyndale Ave, Helena, MT 59601)

Special Use Permit OME No. 0596-0113 from the United States Department of Agriculture (Mountain Top, Helena, MT for Land for KTVH Transmitter)

Lease Agreement between Montana Department of Administration, General Services and KTVQ-TV, dated October 1, 1990 (KRTV Capital Bureau in Helena, MT)

KRTV Communications, LLC has an oral, no cost lease with Kelly Norskog for Translator K11 at Stanford, MT

Lease, dated May 17, 2018, between Douglas W. Demarest amd KRTV Communications, LLC (Demerest Ranch Road Easement)

KATC Communications, LLC:

Lease, dated as of June 15, 1977, by and between Kenneth Clark (as assigned by Helen Latiolais) and Acadian Television Corporation. (7740 Rosewood Drive Vermilion Parish, Kaplan LA 70548)

Lease, dated as of September 5, 2000, by and between Warren L. Guidry and Elizabeth Guidry, L.P. and KATC Communications, Inc. (Acadia Parish DTV Tower, Section 37, Township 8 South, Range 2 East, Church Point, LA 70525)

KATY Communications, LLC has an oral lease with Rfwave who has 1x microwave dish on the

KXLF transmission site building at XL heights for \$500 per month.
(c)
None.
(d)
None.
(e)
None.
(f)
Lexington is replacing the sewer line that runs through the 1065 Russell Cave Road property, bu such replacement will not adversely affect the use of such property. The project is slated for late

ıt 2019 or early 2020 for completion.

(g)

KATC Communications, LLC:

Tower Lease Agreement, dated April 8, 1985, by and between KATC Communications, LLC and KAJN Agape Broadcaster, as amended by the Tower Lease Agreement dated November 16, 2015.

Sublease Agreement, dated November 29, 2000, by and between KATC Communications, Inc. and Louisiana Educational Television Authority

KXLF Communications, LLC:

Lease Agreement, dated, by and between KXLF Communciations, Inc. and GCM Services

Lease Agreement, dated, by and between KXLF Communciations, Inc. and D.J. Minerals

Term of Right-of-Way and Easement Agreement, dated October 5, 1994, by and between KXLF Communication Inc. and Verizon Wireless

Reciprocal Electronic Use Agreement, dated as of December 12, 1986, by and between Eagle Communications, Inc. and KPAX-TV, Inc. and KXLF-TV, Inc.

Tower Site Lease Agreement, dated February 1, 2008, by and between KXLF Communications, Inc. and Hi-Line Radio Fellowship, Inc.

Lease, dated as of April 1, 1999, by and between Educational Media Foundation and KXLF Communications, Inc.

Lease Agreement, dated as of November 1, 2006, by and between KXLF Communications, Inc. and the Educational Media Foundation

Co-Location Agreement, dated February 28, 2007, by and between KXLF Communications, Inc. and DirectTV, Inc.

Lease Agreement, dated January 1, 2016, by and between KXLF Communications, LLC and CCR-Butte III, LLC

U.S. Government Lease for Real Property No. L12CIO0147 between the United States of America and KXLF Communications, Inc.

Sublease, dated April 8, 2004, by and between KCTZ Communications, Inc. and Capstar Radio Operating Company

Lease for Real Property, dated August 28, 1984, by and between the United States of America and KXLF-TV, as amended by Attachment to Lease No. 85-ABF-00004

Sublease, dated August 1, 2009, by and between KCTZ Communications, Inc. and Max Media of Montana, LLC

U.S. Government Lease for Real Proeprty, dated as of May 1, 1979, by and between the United States of America and KXLF

KTVQ Communications, LLC

Ownership and Use Agreement between KTVQ Communications, Inc. and KULR-TVia of Montana, dated 2005

License and Easement Agreement, dated December 27, 2017, by and between KTVQ Communications, LLC and T-Mobile West LLC

Lease Agreement, dated August 18, 1995, by and between KTVQ Communications, Inc. and Calvary Satellite Network

Site Lease, dated April 20, 2004, by and between KTVQ Communications, Inc. and Montana Uplink Corporation, as amended

Tower Lease ,dated August 26, 2005, by and between KTVQ Communications, Inc. and Orion Media LLC

Lease Agreement, dated August 30, 1991, by and between KTVQ Communications, Inc. and Eastern Montana College

Lease Agreement, dated August 18, 1998, by and between KTVQ Communications, Inc. and CSN International

Lease Agreement dated 2006, by and between KTVQ Communications, Inc. and CSN International, Inc.

Lease Agreement, dated September 15, 1998, by and between KTVQ Communications, LLC and AT&T Inc.

Site Lease with Option Agreement, dated August 28, 1998, by and between KTVQ Communications, Inc. and Peach Acquisitions, LLC

Lease Agreement, dated June 30, 2012, by and between Gallatin Wireless, LLC and KTVQ Communications, Inc.

Lease Agreement, dated May 1, 2009, by and between KTVQ Communications, Inc. and Strom & Associates, P.C.

Lease Agreement, dated August 21, 2003 , by and between KTVQ Communications, Inc. and Spectrum Group Architects, PC

Lease Agreement, dated June 1, 2017, by and between KTVQ Communications, Inc. and Big Brothers Big Sisters of Yellowstone Country

Lease Agreement, dated August 1, 2017, by and between KTVQ Communications, Inc. and Netzer Law Office, PC, as amended by the Lease Addendum No. 1 dated November 1, 2018

Lease Agreement, dated May 17, 2016, by and between KTVQ Communications, Inc. and Mehling Tax and Accounting

Lease Agreement, dated September 5, 2017, by and between KTVQ Communications, Inc. and R-Calf USA

Lease Agreement, dated June 21, 2017, by and between KTVQ Communications, Inc. and Gabriel Espino and DLH Marketing

Lease Agreement, dated July 26, 2017, by and between KTVQ Communications, Inc. and Janet Schroeder Counseling

Lease Agreement, dated February 22, 2005, by and between KTVQ Communications, Inc. and Joni Seeley and Carol McCracken

Lease Agreement, dated May 1, 2015, by and between KTVQ Communications, Inc. and MusEco Media & Education

Lease Agreement, dated November 24, 2008, between KTVQ Communications, Inc. and GAP Broadcasting II, LLC

Lease Agreement, dated November 11, 1997, between KTVQ Communications, Inc. and Montana State University-Bilings

Site Lease, dated March 17, 2000, between KTVQ Communications, Inc. and Quorom Broadcasting of Montana, LLC

KXLF Communications, LLC

Lease, dated November 1, 2001, by and between KXLF Communications, Inc. and River Radio Ranch

Tower Site Lease Agreement, dated October 15, 2009, by and between Cordillera Communications, Inc. and KUSM / Montana State University

KCTZ Communications, LLC.

U.S. Government Lease for Real Property, dated June 4, 2013, between The Government of the United States of America and KCTZ Communications, Inc.

Lease Agreement, dated October 2, 2002, by and between KCTZ Communications, Inc. and Eagle Communications Inc.

KRTV Communications, LLC

Sublease Agreement, dated January 1, 2016, by and between KRTV Communications LLC and Lewis & Clark Rural Fire Council

Sub-Lease Agreement, dated January 1, 2016, by and between KRTV Communications LLC and Montana Department of Transportation, Communications Bureau

Sub-Lease Agreement, dated January 1, 2016, by and between KRTV Communications LLC and Montana State University

KRTV Communications, LLC has a reciprocal oral lease with Jim Stiles, who has equipment in their Sweetgrass Hills and Sullivan Hills sites, for the building and tower in West Knees, MT.

WLEX Communications, LLC:

Lease Agreement, dated February 16, 2010, by and between WLEX Communications, LLC and Christian Broadcasting System, Ltd.

Lease Agreement, dated November 16, 2015, by and between WLEX Communications, LLC and North Nicholasville Road Property Owners Association

KSBY Communications, LLC:

Following the expiration of the Lease Agreement, dated as of June 1, 2003, by and between NVG Santa Barbara, LLC and TCSN Net, Inc., KSBY Communications, LLC has an oral, month-to-month lease with TSCN Net, Inc. until a new lease is negotiated.

Short Form Tower License Agreement between Crown Castle MU LLC and KSBY Communications, Inc. dated December 8th, 2009

Santa Ynez Peak Site Use Agreement between Comsite Pacific Inc., Bud-North Shore Communications and KSBY-TV Inc., dated July 1, 1991

Antenna Space License, dated January 13, 2016, by and between KSBY Communications, LLC and Strasbaugh Development Corp.

Antenna Space License, dated February 28, 2018, by and between KSBY Communications, LLC and the County of Santa Barbara

Antenna Space License, dated February 8, 2016,, by and between KSBY Communications, LLC and Radio Bilingue Inc.

Antenna Space License, dated August 20, 2016, by and between KSBY Communications, LLC and KLFF(FM)

Antenna Space License, dated January 1, 2009, by and between KSBY Communications, LLC and the Educational Media Foundation, Santa Monica Community College District and Gold Coast Radio, L.L.C. (as assigned by El Dorado Broadcasters LLC)

Antenna Space License, dated 2008, by and between KSBY Communications, LLC and KXBC, Inc.

Amended and Restated Ground Lease, dated January 26, 2010, between Chumash Hill Properties and KSBY Communications, Inc.

Sublease Agreement, dated as of July 1, 2010, by and between KSBY Communications, Inc. and the County of San Luis Obispo

KRIS Communications, LLC

Antenna Space License, dated July 18, 2017, by and between KRIS Communications, LLC and City of Corpus Christi

FM Facility and Tower Lease Agreement, dated May 4, 1998, between KRIS Communications, LLC and Diocesan Telecommunications Corporation

KPAX Communications, LLC

Reciprocal Sub-Lease Agreement, dated as of March 6, 2015, by and between Bonten Media Group, LLC and KPAX Communications, LLC, and KCTZ Communications, LLC.

(h)

None.

TRANSITION SERVICES AGREEMENT

THIS TRANSITION SERVICES AGREEMENT (this "<u>Agreement</u>") is made and entered into as of [MONTH] [DAY], 2019 ("<u>Closing Date</u>"), between Cordillera Communications, LLC, a South Carolina corporation ("<u>Provider</u>"), and Scripps Media, Inc., a Delaware corporation ("<u>Recipient</u>"). Provider and Recipient are each referred to herein as a "<u>Party</u>" and collectively as the "<u>Parties</u>". Capitalized terms not otherwise defined herein shall have the meanings set forth in the Purchase Agreement (as defined below).

WHEREAS, Provider and Recipient are parties to that certain Purchase Agreement, dated as of October 27, 2018 (the "<u>Purchase Agreement</u>"), pursuant to which, among other things, Provider agreed to sell to Recipient, and Recipient agreed to purchase, the Equity Interests;

WHEREAS, Provider has agreed to provide to Recipient, from and after the Closing Date, certain temporary transition services on the terms and conditions set forth herein; and

WHEREAS, this Agreement constitutes the Transition Services Agreement referred to in the Purchase Agreement.

NOW, THEREFORE, in consideration of the foregoing and the mutual covenants, conditions and provisions contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, each of Provider and Recipient hereby agrees as follows:

ARTICLE I TRANSITION SERVICES TO BE PROVIDED BY PROVIDER

- 1.1 <u>Transition Services</u>. During the Term, Provider shall provide to Recipient, on the terms set forth herein, the services specifically listed on Schedule A attached hereto (the "<u>Services Schedule</u>"). The services described in the Services Schedule are, collectively, the "<u>Transition Services</u>". Upon the expiration of the Term (or earlier pursuant to <u>Article V</u>) the obligations of Provider with respect to the provision of the Transition Services (or applicable Transition Service) shall automatically and immediately terminate.
- 1.2 <u>Standard of Performance</u>. Provider shall use commercially reasonable effort to provide (or cause to be provided) to Recipient the Transition Services at substantially the same level of service, and with substantially the same degree of care, as Provider used in its operation of the Stations prior to Closing.
- 1.3 <u>Usage</u>. The Transition Services may not be used by Recipient for any other purpose or in any other manner (including as to volume, amount, level or frequency, as applicable) other than as the Transition Services were used in connection with the Business as of the Closing Date. Without limiting the generality of the foregoing, the Transition Services shall be used only by Recipient and its Affiliates and solely in connection with the operation of the Business and Recipient shall not resell, license the use of, or otherwise permit the use by others of any such Transition Services other than Recipient's Affiliates.

Right to Change Practices. Nothing herein shall prohibit Provider from making such changes as it deems necessary in its sole discretion (including upgrading or changing technology, software or information systems used by it in connection with this Agreement and replacing third-party service providers) and applying such changes, to the extent applicable, to the Transition Services; provided that the Provider shall, to the extent reasonably practicable, provide at least twenty (20) days' prior notice of any change that could reasonably be expected to materially and adversely affect the Transition Services and the Parties shall discuss in good faith whether such change can be implemented so as not to so affect the Transition Services provided hereunder. Provider shall have no obligation to provide, or cause to be provided, Transition Services to the extent that any changes are made to the business of Recipient or its Affiliates that materially increase or materially and adversely change Provider's burden or cost with respect to the provision of such Transition Services, or that make commercially impracticable the provision of such Transition Services. Provider shall have no obligation to provide, or to cause to be provided, Transition Services to any Person other than Recipient and its Affiliates. Without limiting the generality of the foregoing, the Transition Services are not intended to (and shall not) include assistance for expansion, acquisition, employee additions or reductions not in the ordinary course of business, new applications, or site redesign by Recipient or any of its Affiliates or their respective businesses.

1.5 <u>Cooperation; Access.</u>

- (a) Recipient agrees to fully cooperate (at Recipient's sole cost and expense) with Provider in connection with any matters related to the provision or receipt of the Transition Services. To the extent required for the performance of any Transition Service, Recipient shall, at its own expense, provide to Provider reasonable access, on an as needed basis, to Recipient's and its Affiliates' personnel, equipment, office space, telecommunications and computer systems and any other assets and operations reasonably required for delivery of all or any part of the Transition Services. Provider and its Affiliates and Third Party Service Providers shall be entitled to rely, without inquiry, upon the genuineness, validity or truthfulness of any document, instrument or other writing presented by Recipient or its Affiliates in connection with this Agreement.
- (b) Provider agrees that all of its and its Affiliates' employees and any Third Party Service Providers, when on the property of Recipient or any of its Affiliates or when given access to any equipment, office space, telecommunications, computer systems or any other assets owned or controlled by Recipient or any of its Affiliates, shall conform to the policies and procedures of Recipient and its Affiliates concerning health, safety and security that are made known to Provider in advance in writing.
- (c) Provider and its Affiliates and any Third Party Service Providers shall not be liable for any impairment of any Transition Service directly caused by their not receiving the information, materials or access required by this <u>Section 1.5</u>. If any Transition Service provided hereunder is dependent on, or will require the involvement or performance of, Recipient or any of its Affiliates (including Recipient or its Affiliate maintaining sufficient and appropriate levels of staffing, including, as applicable, with respect to level of experience and skill), Provider and its Affiliates and Third Party Service Providers shall not be deemed in breach of this Agreement, including any Services Schedule hereto, or otherwise be liable for any failure to provide any

Transition Services, or to otherwise fully perform any of its obligations hereunder, including any Services Schedule, to the extent arising from any failure of Recipient or any of its Affiliates to fully perform its obligations hereunder.

- (d) As required by the Communications Act of 1934, as amended, at all times Recipient shall be responsible for maintaining the supervision and control of the Station to which the Transition Services are being provided, including control of such Station's personnel, programming and finances.
- 1.6 <u>Transitional Nature of Services</u>. Recipient acknowledges and agrees that Provider is providing (or causing to be provided) the Transition Services on a transition basis only in order to allow Recipient a period of time to provide or obtain similar services for itself, and that Provider is not a commercial provider of such Transition Services. Accordingly, as promptly as practicable following the execution of this Agreement and in no event later than the end of the Term, Recipient agrees to use commercially reasonable efforts to promptly transition each Transition Service to its own internal organization or obtain alternate third-party sources to provide the Transition Services and end its use of each and every Transition Service (the cessation of a Transition Service, a "Service Migration"). In no event shall Provider be responsible for providing any services in connection with, or bearing any costs of, any Service Migration.

1.7 <u>Compliance with Law; Exclusions.</u>

- (a) Without limiting Provider's obligations under Sections 1.1 and 1.2 hereof, neither Provider nor any of its Affiliates nor any Third Party Service Provider shall be obligated to perform, or to cause to be performed, any Transition Service in a manner that is not consistent with their respective policies, procedures and practices, and Provider shall determine, in its sole discretion, the means and resources used to provide the Transition Services pursuant to this Agreement. Provider's obligation to provide the Transition Services hereunder is subject, in all respects, to the current policies, procedures and practices of Provider and its Affiliates, including as such policies, procedures and practices may be amended by Provider or its Affiliates from time to time in the ordinary course of business, as applicable. Provider shall not have any obligation to (x) replace or cause the replacement of any Person who is employed or otherwise engaged by Provider or any of its Affiliates to perform Transition Services upon the termination of such Person's employment or other engagement with Provider or any of its Affiliates, as the case may be, for whatever reason (provided that Provider shall still provide the Transition Services using suitably qualified, skilled and experienced personnel), or (y) perform or cause to be performed any Transition Services performed by any Person who is employed or otherwise engaged by Provider or any of its Affiliates if such Person becomes employed or otherwise engaged by Recipient or any of its Affiliates during the Term.
- (b) Each Party shall perform its obligations and exercise its rights hereunder in material compliance with all applicable Laws. Notwithstanding anything to the contrary herein, including in any Services Schedule, Provider, its Affiliates and any Third Party Service Providers shall not be required hereunder to take any action (including by providing any Transition Services) that would constitute (a) a violation of applicable Law, (b) a breach of any

of Provider's contractual obligations to a third party, or (c) any other violation of a third party's rights.

- Notwithstanding any other provision of this Agreement, the Transition Services shall in no event include, and Provider shall not be required to provide or cause to be provided to Recipient, services that involve or relate to (i) the custody of cash; (ii) the commingling of funds; (iii) the provision of legal, compliance, regulatory or tax advice; (iv) the filing of tax returns; (v) the provision of data or records except as otherwise provided in the Purchase Agreement or herein; (vi) the funding of any "employee benefit plan" as such term is defined in Section 3(3) of ERISA, or other retirement, deferred compensation, incentive, equity based, severance, employment, change in control or fringe benefit plan, program, policy or arrangement, whether or not subject to ERISA, adopted or maintained by Recipient or its Affiliates (the "Recipient Employee Benefit Plans"); (vii) any benefit program compliance services with respect to any Recipient Employee Benefit Plans; (viii) the funding of any payroll, payroll taxes or workers' compensation or (ix) the reimbursement of any travel expenses. Recipient further acknowledges and agrees that Provider will not become a fiduciary (including for purposes of ERISA) with respect to any Recipient Employee Benefit Plans by reason of providing any of the Transition Services under this Agreement, and to the extent the provision of any such Transition Service would cause Provider to become a fiduciary (including for purposes of ERISA, the Code, or other applicable Laws) with respect to any Recipient Employee Benefit Plan, Provider shall not be required to provide, or cause to be provided, such Transition Services to any of Recipient or its Affiliates.
- 1.8 <u>Delegation of Transition Services</u>. Recipient acknowledges that Provider may provide the applicable Transition Services itself, through any of its Affiliates or through one or more third parties engaged by Provider to provide Transition Services in accordance with the terms of this Agreement (each such third party, a "<u>Third Party Service Provider</u>"). Provider acknowledges that notwithstanding any delegation of its responsibilities under this Agreement, it shall remain responsible for the provision of the Transition Services and its Affiliates' and Third Party Service Provider's compliance with the standard of performance set forth herein.
- 1.9 <u>Intellectual Property Rights</u>. Each Party shall retain all right, title and interest in and to its Intellectual Property. All Intellectual Property created or developed by or on behalf of Provider in connection with this Agreement shall belong to Provider and all Intellectual Property created or developed by or on behalf of Recipient or any of its Affiliates in connection with this Agreement shall belong to Recipient or its applicable Affiliate. Except as set forth in <u>Section 1.10</u>, no license, express or implied, is being granted by Provider or Recipient under this Agreement.
- 1.10 <u>License to Use Intangible Property</u>. Recipient, on behalf of itself and its Affiliates, hereby grants to Provider, its Affiliates and any Third Party Service Provider a limited, royalty-free, non-exclusive, worldwide right and license to use the Intangible Property, and any other Intellectual Property controlled by Recipient or any of its Affiliates, in each case, as may be reasonably necessary to perform the Transition Services in connection with, and solely during the Term of, and solely for the purposes of performing the Transition Services under, this Agreement; provided, however, that no such license of any Licensed IP or other Intellectual Property shall be granted hereunder if such license would violate any IP Agreement or the rights

of any third party. The license set forth in this Section 1.10 shall automatically and immediately terminate upon the termination of the Term.

ARTICLE II PRICING AND PAYMENT FOR TRANSITION SERVICES

- 2.1 <u>Service Fees</u>. The fees ("<u>Fees</u>") payable to Provider for each Transition Service during the Initial Term are set forth in the attached Services Schedule¹. Commencing upon the expiration of the Initial Term and for the remainder of the Term, if applicable, the Fees for those Transition Services to be provided for the remainder of the Term will be as mutually agreed upon by the Parties.
- 2.2 <u>Third Party Costs</u>. Recipient shall reimburse Provider for all reasonable and documented costs, fees and charges paid by Provider to Third Party Service Providers in connection with performing the applicable Transition Services (collectively, the "<u>Third Party Pass Through Cost</u>").
- 2.3 <u>Invoices</u>. Provider shall invoice Recipient monthly for the Transition Services, including all Third Party Pass Through Costs, and Recipient shall pay Provider within thirty (30) days after the date of Recipient's receipt of each such invoice. Each invoice shall set forth in reasonable detail the applicable Transition Services provided during such period and the corresponding amounts owed for each of the Transition Services (with such supporting documentation as Recipient may reasonably request with respect to Third Party Pass Through Costs). Recipient shall not offset any amounts payable by it hereunder against any other amounts owed to it by Provider. If Recipient fails to pay the full amount of any undisputed invoice within such thirty (30) day period, such failure will be considered a material breach of this Agreement, for which Provider may, without liability and in its sole discretion, (i) suspend its obligations hereunder to provide any and all Transition Services to Recipient until such time as all invoices have been paid in full or (ii) terminate this Agreement. Notwithstanding the foregoing, Provider may terminate this Agreement at any time following such twenty (20) day period even if all invoices have been paid in full after such twenty (20) day period.
- 2.4 <u>Interest Payable on Amounts Past Due</u>. All payments required to be made pursuant to this Agreement shall bear interest from and including the date such payment is due until but excluding the date of payment, at a monthly rate equal to the lesser of (i) one and one quarter of one percent (1.25%) per month or (ii) the maximum rate permitted by applicable Law. In addition, Recipient shall indemnify Provider for its costs, including reasonable attorneys' fees and disbursements, incurred to collect any unpaid amount that is past due and not disputed in writing by Recipient.
- 2.5 <u>Taxes</u>. The fees described herein with respect to the Transition Services do not include any Taxes. In addition to the amounts required to be paid as set forth herein, Recipient shall pay and be responsible for and shall promptly reimburse Provider for any sales and use Taxes imposed with respect to the provision of Transition Services hereunder.

_

¹ NTD: There will need to be a fee listed for each Transition Service so that the fee can be determined on partial termination.

2.6 No Right to Setoff. There shall be no right of setoff with respect to any claim, debt or obligation against payments to Provider under this Agreement.

ARTICLE III REPRESENTATIVES

3.1 Provider and Recipient shall each designate in writing a Coordinators. representative to act as Provider's and Recipient's respective primary contact persons to coordinate the provision of all of the Transition Services (collectively, the "Primary Coordinators"). Each Primary Coordinator may designate one or more service coordinators for each specific Transition Service (the "Service Coordinators"). Each Party may treat an act of a Primary Coordinator of another Party as being authorized by such other Party without inquiring behind such act or ascertaining whether such Primary Coordinator had authority to so act, and each Party may treat an act of a Service Coordinator as being authorized by such other Party only to the extent such act is directly related to the Transition Service for which such Service Coordinator has been designated; provided, however, that no such Primary Coordinator or Service Coordinator has authority to amend this Agreement. Each Party shall promptly (and in any event within ten (10) business days) advise the other in writing of any change in its Primary Coordinator or any Service Coordinator for a particular Transition Service. Provider and Recipient agree that all communications relating to the provision of the Transition Services shall be by electronic mail and shall be directed to the Service Coordinators for such Transition Service with copies to the Primary Coordinators.

ARTICLE IV CONTRACT RIGHTS

4.1 <u>Third Party Consents</u>. Provider's obligation to deliver any Transition Services described in this Agreement is conditional upon Provider obtaining the consent, where necessary, of any relevant third party, and Provider will use commercially reasonable efforts to obtain such third-party consents; <u>provided</u>, <u>however</u>, that if such consents cannot be obtained, the Parties shall work together in good faith and use their respective commercially reasonable efforts to arrange for alternative methods of delivering such Transition Service. Provider and Recipient shall each negotiate in good faith to determine the responsibility for paying the costs, fees and expenses of obtaining any third party consents or arranging alternative methods of delivering Transition Services.

ARTICLE V TERM; TERMINATION

5.1 <u>Term.</u> This Agreement shall become effective upon the Closing Date of this Agreement and shall remain in effect for ninety (90) days (the "<u>Initial Term</u>"), unless earlier terminated pursuant to <u>Sections 5.2</u> or <u>5.3</u>. Each Transition Service will terminate on the earlier of: (i) the expiration date specified in the Services Schedule, if any, for such Transition Service; (ii) the termination of such Transition Service pursuant to <u>Section 5.2</u>; and (iii) the expiration or termination of this Agreement.

- 5.2 <u>Partial Termination</u>. Recipient may terminate or reduce the scope of the provision of any Transition Service (in whole or in part) upon thirty (30) days' prior written notice to Provider. Any election to so terminate or reduce any Transition Service or a portion thereof shall not relieve Provider of its continuing duty to provide those Transition Services or portions thereof that have not been terminated or reduced. Recipient will reimburse Provider for the reasonable out-of-pocket costs incurred by Provider resulting from Recipient's early termination or reduction of such Transition Service being performed, including commitments made to, or in respect of, personnel or Third Party Service Providers, prepaid expenses related to the terminated or reduced Transition Service and costs related to terminating such commitments. Such reimbursement shall be made by Recipient to Provider in accordance with <u>Article II</u>.
- 5.3 <u>Complete Termination</u>. The Parties may terminate this Agreement, in whole, solely in accordance with this <u>Section 5.3</u>.
- (a) <u>By Recipient</u>. Recipient may terminate this Agreement for any reason or no reason, at any time, upon not less than thirty (30) days' prior written notice to Provider. Recipient will reimburse Provider for the reasonable out-of-pocket costs incurred by Provider resulting from Recipient's early termination, including commitments made to, or in respect of, personnel or Third Party Service Providers, prepaid expenses related to the terminated Transition Services and costs related to terminating such commitments. Such reimbursement shall be made by Recipient to Provider in accordance with <u>Article II</u>. All accrued and unpaid charges for Transition Services shall be due and payable upon termination of this Agreement and shall be paid by Recipient to Provider in accordance with <u>Article II</u>.
- (b) <u>By Provider</u>. Except with respect to a termination by Provider for Recipient's non-payment of invoices which is governed by <u>Section 2.3</u> above, Provider may terminate this Agreement in the event of a material breach of this Agreement by Recipient (i) if such material breach is not curable, upon written notice to Recipient, or (ii) if such material breach is curable, if such material breach is not cured within ten (10) days after Provider notifies Recipient of such material breach.
- (c) <u>By Either Party</u>. Either party may terminate this Agreement any time upon giving written notice to the other party if the other party is adjudicated as bankrupt, becomes insolvent, suffers permanent or temporary court-appointed receivership of substantially all of its property, makes a general assignment for the benefit of creditors, or suffers the filing of a voluntary or involuntary bankruptcy petition that is not dismissed within sixty (60) days after filing.
- (d) <u>By Both Parties</u>. This Agreement will terminate upon the mutual written agreement of the Parties.
- 5.4 <u>Extension</u>. The Initial Term may be extended only by mutual written agreement of the Parties either in whole or with respect to one or more of the Transition Services. The Initial Term and any extension thereof are collectively the "Term."
- 5.5 <u>Effect of Termination or Expiration</u>. Except as otherwise agreed by the Parties, promptly following the expiration or termination of this Agreement (but in no event later than

- thirty (30) days thereafter), each Party shall return to the other Party in accordance with such other Party's instructions all of the other Party's Confidential Information in its possession or control that was disclosed under this Agreement. Each Party acknowledges and agrees that the termination or expiration of a Transition Service or this Agreement for any reason shall not release a Party from any liability or obligation that already has accrued as of the effective date of such termination or expiration, as applicable, and shall not constitute a waiver or release of, or otherwise be deemed to adversely affect, any rights, remedies or claims which a Party may have hereunder at Law, in equity or otherwise or which may arise out of or in connection with such termination or expiration.
- 5.6 <u>Survival</u>. <u>Section 1.9</u> (Intellectual Property Rights), <u>Article II</u> (Pricing and Payment for Transition Services), <u>Section 5.5</u> (Effect of Termination or Expiration), this <u>Section 5.6</u> (Survival), <u>Section 6.2</u> (DISCLAIMER OF WARRANTIES), <u>Article VII</u> (LIMITATION OF LIABILITY), <u>Article VIII</u> (Confidentiality) and <u>Article IX</u> (Miscellaneous) shall survive any termination or expiration of this Agreement.

ARTICLE VI REPRESENTATIONS AND WARRANTIES

- 6.1 <u>Representations and Warranties</u>. Each Party represents and warrants to the other that:
- (a) it is a legal entity validly existing and in good standing (to the extent such concept is legally recognized) under the Laws of the jurisdiction of its organization;
- (b) it has all necessary power and authority, and has taken all action necessary, to authorize, execute, deliver and perform this Agreement, in accordance with the terms of this Agreement; and
- (c) this Agreement has been duly and validly executed and delivered by such Party, and, assuming the due authorization, execution and delivery by the other Party, constitutes a valid, legal and binding agreement of such Party, enforceable against such Party in accordance with its terms, subject to the effect of any applicable Laws relating to bankruptcy, reorganization, insolvency, moratorium, fraudulent conveyance or preferential transfers, or similar Laws relating to or affecting creditors' rights generally and subject, as to enforceability, to the effect of general principles of equity (regardless of whether such enforceability is considered in a proceeding at equity or at Law).
- (d) The Parties acknowledge and agree that, except as expressly set forth herein, the Recipient assumes all risks and liabilities arising from or relating to its use of and reliance upon the Transition Services, and Provider makes no representation or warranty with respect thereto.
- 6.2 <u>DISCLAIMER OF WARRANTIES</u>. THIS IS A SERVICE AGREEMENT. EXCEPT AS EXPRESSLY STATED ELSEWHERE IN THIS AGREEMENT, (I) PROVIDER DOES NOT GUARANTEE, REPRESENT OR WARRANT THE TRANSITION SERVICES TO BE PROVIDED HEREUNDER, (II) THE TRANSITION SERVICES SHALL BE PROVIDED ON AN "AS IS" AND "WITH ALL FAULTS" BASIS, AND (III) THERE ARE

NO OTHER, AND PROVIDER DOES NOT MAKE AND HEREBY DISCLAIMS ALL OTHER, REPRESENTATIONS, WARRANTIES OR GUARANTEES OF ANY KIND WITH RESPECT TO THIS AGREEMENT OR THE TRANSITION SERVICES, EXPRESS OR IMPLIED, INCLUDING ANY WARRANTY OF MERCHANTABILITY, TITLE/NONINFRINGEMENT OR FITNESS FOR A PARTICULAR PURPOSE, WHICH ARE SPECIFICALLY DISCLAIMED. PROVIDER DOES NOT GUARANTEE, REPRESENT OR WARRANT THE CORRECTNESS, COMPLETENESS, CURRENTNESS, MERCHANTABILITY, OR FITNESS FOR A PARTICULAR PURPOSE OF ANY DATA PROVIDED TO RECIPIENT.

ARTICLE VII LIMITATION OF LIABILITY

7.1 Indemnification.

- (a) Recipient agrees to indemnify, defend and hold Provider, its Affiliates, the Third Party Service Providers, and each of Provider's, its Affiliates' and the Third Party Service Providers' respective employees, agents, officers and directors harmless from and against any losses, costs, interest, charges, expenses (including reasonable attorneys' fees), obligations, liabilities, settlement payments, awards, judgments, fines, penalties, damages, assessments or deficiencies ("Losses") arising out of (i) any third-party claim in connection with or by reason of this Agreement, (ii) any Transition Services provided by or on behalf of Provider hereunder, or (iii) any use of such Transition Service by Recipient, except in each case to the extent such Losses arise out of the willful misconduct or gross negligence of Provider, any of its Affiliates or any of the Third Party Service Providers or any breach of this Agreement by Provider.
- (b) Subject to the limitations set forth in Section 7.2, Provider agrees to indemnify, defend and hold Recipient, its Affiliates, and each of Recipient's and its Affiliates' respective employees, agents, officers and directors harmless from and against any Losses arising out of (i) the willful misconduct or gross negligence of Provider, any of its Affiliates or any of the Third Party Service Providers in connection with the Transition Services provided hereunder or (ii) any material breach of this Agreement by Provider.
- 1.2 LIMITATION ON LIABILITY. IN NO EVENT SHALL PROVIDER BE LIABLE FOR ANY SPECIAL, PUNITIVE, CONSEQUENTIAL, INDIRECT, INCIDENTAL OR EXEMPLARY DAMAGES (INCLUDING LOSS OF DATA, PROFITS, INTEREST, OR REVENUE OR ANY INTERRUPTION OF BUSINESS) IN CONNECTION WITH, ARISING FROM OR IN RELATION TO THIS AGREEMENT (INCLUDING ANY OF THE TRANSITION SERVICES TO BE PROVIDED HEREUNDER OR THE PERFORMANCE OF OR FAILURE TO PERFORM PROVIDER'S OBLIGATIONS UNDER THIS AGREEMENT), WHETHER BASED ON WARRANTY, CONTRACT, TORT (INCLUDING NEGLIGENCE OR STRICT LIABILITY) OR ANY OTHER LEGAL OR EQUITABLE GROUNDS, AND REGARDLESS OF WHETHER SUCH DAMAGES ARE FORESEEABLE OR WHETHER PROVIDER IS ADVISED OF THE POSSIBILITY OR LIKELIHOOD OF SUCH DAMAGES. IN NO EVENT SHALL PROVIDER BE LIABLE FOR ANY LOSSES IN CONNECTION WITH, ARISING FROM OR IN RELATION TO THIS AGREEMENT (INCLUDING ANY OF THE TRANSITION SERVICES TO BE PROVIDED HEREUNDER OR THE

PERFORMANCE OF OR FAILURE TO PERFORM PROVIDER'S OBLIGATIONS UNDER THIS AGREEMENT) FOR ANY AMOUNT IN EXCESS OF THE AGGREGATE FEES RECEIVED (AT THE TIME OF THE DISPUTE) BY PROVIDER UNDER THIS AGREEMENT. THE PARTIES INTEND THAT NO DOUBLE REMEDIES OR RECOVERIES ARE INTENDED OR PERMITTED UNDER THIS AGREEMENT AND THAT CLAIMS ASSERTED UNDER ONE SECTION OR SUBSECTION OF THIS AGREEMENT MAY NOT ALSO BE ASSERTED UNDER ANOTHER SECTION OR SUBSECTION OF THIS AGREEMENT OR UNDER THE PURCHASE AGREEMENT OR ANY OTHER ANCILLARY DOCUMENT IF SUCH ASSERTION WOULD RESULT IN DOUBLE RECOVERY.

ARTICLE VIII CONFIDENTIALITY

- 8.1 <u>Duty of Confidentiality</u>. Each Party (a "<u>Receiving Party</u>") agrees to safeguard the other Party's (the "<u>Disclosing Party</u>") Confidential Information received under this Agreement with the same degree of care that the Receiving Party uses to protect its own similar Confidential Information, but in no event less than a reasonable degree of care. Such Confidential Information of the Disclosing Party received hereunder may be used by the Receiving Party in connection with the Transition Services only and may only be copied or reproduced to the extent reasonably necessary for the Receiving Party to perform its obligations or enforce its rights under this Agreement. The Disclosing Party shall not disclose any such Confidential Information of the Receiving Party to any Person; <u>provided</u>, that, Provider may disclose Recipient's Confidential Information to any of its Affiliates or Third Party Service Providers as may be reasonably necessary to perform the Transition Services in connection with, and solely during the Term of, and solely for the purposes of performing the Transition Services under, this Agreement.
- 8.2 <u>Permitted Disclosure</u>. Notwithstanding anything contained herein to the contrary, Section 8.1 shall not restrict the Receiving Party from disclosing Confidential Information to the extent reasonably necessary in connection with the enforcement of this Agreement or as required by applicable Law or legal or regulatory process (including to the extent requested by any Governmental Entity in connection with any such Law or legal or regulatory process), including any Tax audit or litigation. In the event that the Receiving Party or its permitted service providers become legally required by deposition, interrogatory, request for documents, subpoena, civil investigative demand, formal regulatory request or similar judicial or administrative process to disclose any Confidential Information of the Disclosing Party, the Receiving Party shall, to the extent permitted by Law, provide the Disclosing Party with prompt prior written notice of such requirement so that the Disclosing Party may seek a protective order or other similar remedy to cause such Confidential Information not to be disclosed. The Receiving Party shall reasonably cooperate with the Disclosing Party, at the Disclosing Party's expense, in connection with the Disclosing Party's seeking of such protective order or similar remedy. In the event that such protective order or other similar remedy is not obtained or the Disclosing Party waives compliance with the provisions of this Section 8.2, the Receiving Party or its permitted service provider shall furnish only that portion of the Confidential Information that has been legally required, and shall exercise commercially reasonable efforts to obtain assurance that confidential treatment shall be accorded such disclosed Confidential Information.

Confidential Information. The following shall be considered "Confidential Information" of a Party under this Agreement: all proprietary and confidential information, provided or received in connection with the provision of the Transition Services hereunder, concerning the business, business relationships (including prospective customers and business partners) and financial affairs of such Party or its Affiliates, whether or not in writing and whether or not labeled or identified as confidential or proprietary, including inventions, trade secrets, technical information, know-how, product and pricing information and plans, research and development activities, marketing plans and activities, customer, supplier and prospect information, employee and financial information, and information disclosed by third parties of a proprietary or confidential nature or under an obligation of confidence; provided that Confidential Information does not include, and there shall be no obligation hereunder, with respect to information that (i) is or becomes generally available to the public, other than as a result of a disclosure in breach of this Agreement by the Receiving Party, (ii) the Receiving Party can demonstrate was or became available to the Receiving Party from a third party not bound by any confidentiality obligation to the Disclosing Party, (iii) was already in the Receiving Party's possession (without an obligation of confidentiality) prior to direct or indirect disclosure pursuant to this Agreement (or any predecessor agreement between the Parties governing the confidentiality of such information) and was not generated in the course of, or in connection with, the Transition Services, or (iv) is developed independently by the Receiving Party without reference to the Disclosing Party's Confidential Information.

ARTICLE IX MISCELLANEOUS

- 9.1 <u>Amendment and Waiver</u>. This Agreement may not be modified or amended except by an instrument or instruments in writing signed by each Party and, to the extent required, with the approval of the DOJ. Either Party to this Agreement may, only by an instrument in writing, waive compliance by the other Party to this Agreement with any term or provision of this Agreement on the part of such other Party to this Agreement to be performed or complied with. The waiver by any Party to this Agreement of a breach of any term or provision of this Agreement shall not be construed as a waiver of any subsequent breach.
- 9.2 <u>Notices</u>. All notices and other communications to be given to any Party hereunder shall be sufficiently given for all purposes hereunder if in writing and upon delivery if delivered by email or by hand, one (1) Business Day after being sent by courier or overnight delivery service, three (3) Business Days after being mailed by certified or registered mail, return receipt requested, with appropriate postage prepaid, or when sent in the form of a facsimile and receipt confirmation is received, and shall be directed to the address or facsimile number set forth below (or at such other address or facsimile number as such Party shall designate by like notice):

If to Provider:

Cordillera Communications, LLC c/o Evening Post Industries, Inc. 134 Columbus Street Charleston, South Carolina 29403 Attention: ______ Email: _____

with a copy (which shall not constitute notice) to:

Cooley LLP 1299 Pennsylvania Avenue, NW Suite 700 Washington, DC 20004 Attn: Michael D. Basile Email: mbasile@cooley.com

If to Recipient:

Scripps Media, Inc. 312 Walnut Street, 28th Floor Cincinnati, Ohio 45202

Attention: Robin A. Davis, Vice President/Strategy and Corporate

Development

Email: robin.davis@scripps.com

with a copy (which shall not constitute notice) to:

Scripps Media, Inc. 312 Walnut Street, 28th Floor Cincinnati, Ohio 45202

Attention: William Appleton, Executive Vice President and General Counsel

Email: appleton@scripps.com

- 9.3 <u>Successors and Assigns</u>. This Agreement shall be binding upon and inure to the benefit of the Parties to this Agreement and their respective successors and assigns; <u>provided</u>, <u>however</u>, that no Party to this Agreement may assign this Agreement without the express prior written consent of the other Party to this Agreement.
- 9.4 <u>Severability</u>. If any term, provision, covenant or restriction of this Agreement is held by a court of competent jurisdiction or other authority to be invalid, void or unenforceable, the remainder of the terms, provisions, covenants and restrictions of this Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner materially adverse to any Party. Upon such a determination, the Parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the Parties as closely as possible in a mutually acceptable manner in order that the transactions contemplated hereby be consummated as originally contemplated to the fullest extent possible.

- 9.5 <u>Third-Party Beneficiaries</u>. Except with respect to any indemnitees under <u>Section 7.1</u>, this Agreement is not intended to confer on or on behalf of any Person not a party to this Agreement (and their successors and assigns) any rights, benefits, causes of action or remedies with respect to the subject matter or any provision hereof.
- 9.6 Entire Agreement. This Agreement (including the Services Schedule), together with Purchase Agreement and other agreements contemplated hereby and thereby, constitutes the entire agreement between the Parties with respect to the subject matter of this Agreement and supersedes any prior discussion, correspondence, negotiation, proposed term sheet, agreement, understanding or arrangement. Each Schedule to this Agreement is hereby incorporated into this Agreement by reference.
- 9.7 <u>Counterparts; Electronic Delivery.</u> This Agreement may be executed in one or more counterparts, and by either of the Parties in separate counterparts, each of which when executed shall be deemed to be an original but all of which taken together shall constitute one and the same agreement. Delivery of an executed counterpart of a signature page to this Agreement by facsimile shall be as effective as delivery of a manually executed counterpart of this Agreement.
- 9.8 <u>Headings; Definitions</u>. The section and article headings contained in this Agreement are inserted for convenience of reference only and will not affect the meaning or interpretation of this Agreement.

9.9 <u>Governing Law</u>.

- (a) This Agreement shall be governed by and construed in accordance with the Laws of the State of Delaware applicable to contracts executed and to be performed wholly within such State and without reference to the choice-of-law principles that would result in the application of the Laws of a different jurisdiction.
- (b) Each Party irrevocably submits to the exclusive jurisdiction of the federal courts located in the State of Delaware any Action arising out of or relating to this Agreement, and hereby irrevocably agrees that all claims in respect of such Action may be heard and determined in such court. Each Party hereby irrevocably waives, to the fullest extent that it may effectively do so, the defense of an inconvenient forum to the maintenance of such Action. The Parties further agree, (i) to the extent permitted by Law, that a final and unappealable judgment against any of them in any Action contemplated above shall be conclusive and may be enforced in any other jurisdiction within or outside the United States by suit on the judgment, a certified copy of which shall be conclusive evidence of the fact and amount of such judgment and (ii) that service of process upon such Party in any such Action shall be effective if notice is given in accordance with Section 9.2.
- (c) Each Party to this Agreement waives trial by jury in any Action brought by any of them against the other arising out of or in any way connected with this Agreement or any other agreements executed in connection herewith, or the administration thereof, or any of the transactions contemplated herein or therein. No Party to this Agreement shall seek a jury trial in any Action based upon, or arising out of, this Agreement or any related instruments or the

relationship between the Parties. No Party will seek to consolidate any such Action in which a jury trial has been waived with any other action in which a jury trial cannot be or has not been waived. Each Party to this Agreement certifies that it has been induced to enter into this Agreement or instrument by, among other things, the mutual waivers and certifications set forth above in this section. No Party has in any way agreed with or represented to any other Party that the provisions of section will not be fully enforced in all instances.

- Recipient and Provider are not employees, partners or joint venturers of the other. Under no circumstances shall any of the employees of Recipient or Provider be deemed to be employees of the other for any purpose. Recipient and Provider shall not have the right to bind the other to any agreement with a third party nor to represent itself or themselves as a partner or joint venturer of the other. For the avoidance of doubt, Recipient and Provider shall be solely responsible for the operation of their respective businesses and the decisions and actions taken in connection therewith, and nothing contained herein shall impose any liability or responsibility on Provider with respect thereto.
- 9.11 <u>Purchase Agreement</u>. Nothing herein is intended to modify, limit or otherwise affect the representations, warranties, covenants, agreements and indemnifications contained in the Purchase Agreement, and such representations, warranties, covenants, agreements and indemnifications shall remain in full force and effect in accordance with the terms of the Purchase Agreement.
- Force Majeure. Neither Party shall be liable to the other for any failure or delay in the performance of its obligations under this Agreement (other than with respect to payment obligations under this Agreement) to the extent such failure or delay is caused by a Force Majeure Event. Upon the occurrence of a Force Majeure Event, (i) the non-performing Party shall promptly notify the other Party of the circumstances hindering its performance and of its plans and efforts to implement a work-around, (ii) the non-performing Party shall be excused from any further performance or observance of the affected obligation(s) for as long as such circumstances prevail, and (iii) the non-performing Party shall continue to attempt to recommence performance or observance to the greatest extent possible without delay. The non-performing Party shall also notify the other Party promptly when the Force Majeure Event has abated. If Provider fails to provide any Transition Services in connection with any Force Majeure Event, the amounts due Provider under this Agreement shall be equitably adjusted in a manner such that Recipient shall not be responsible for the payment of any amounts for such Transition Services that Provider failed to provide. The term "Force Majeure Event" shall mean an event or condition that is caused by or results from any (a) act of war, terrorism, civil riots or rebellions, (b) Law, demand, seizure or requirement of any Governmental Entity, quarantine, embargo or any other similar unusual action of a Governmental Entity, (c) extraordinary element of nature or act of God, fire, flood, or storm, (d) strike, lockout or other labor trouble, delays by suppliers or carriers, shortages of fuel, power, raw materials or components, or (e) any other event or condition outside the reasonable control of the Party subject to such failure or delay. Additionally, Provider shall not be liable under this Agreement for any failure to provide or make available Transition Services as set forth herein to the extent such failure was the direct result of Recipient's operations or systems, or acts or omissions of Recipient.

Interpretation; Absence of Presumption. For the purposes of this Agreement, (i) words in the singular shall be held to include the plural and vice versa, and words of one gender shall be held to include the other gender as the context requires; (ii) references to the terms Article, Section, and Schedule are references to the Articles, Sections, and Schedules to this Agreement unless otherwise specified; (iii) the terms "hereof," "herein," "hereby," "hereto," and derivative or similar words refer to this entire Agreement, including the Schedules hereto; (iv) references to "\$" shall mean U.S. dollars; (v) the word "including" and words of similar import when used in this Agreement shall mean "including without limitation," unless otherwise specified; (vi) the word "or" shall not be exclusive; (vii) the word "extent" in the phrase "to the extent" shall mean the degree to which a subject or other thing extends, and such phrase shall not mean simply "if"; (viii) references to "written" or "in writing" include in electronic form; (ix) provisions shall apply, when appropriate, to successive events and transactions; (x) the headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement; (xi) Provider and Recipient have each participated in the negotiation and drafting of this Agreement and if an ambiguity or question of interpretation should arise, this Agreement shall be construed as if drafted jointly by the parties thereto and no presumption or burden of proof shall arise favoring or burdening either Party by virtue of the authorship of any of the provisions in this Agreement; (xii) a reference to any Person includes such Person's successors and permitted assigns; (xiii) any reference to "days" means calendar days unless Business Days are expressly specified; (xiv) when calculating the period of time before which, within which or following which any act is to be done or step taken pursuant to this Agreement, the date that is the reference date in calculating such period shall be excluded; and (xv) unless otherwise stated in this Agreement, references to any Contract are to that Contract as amended, modified or supplemented from time to time in accordance with the terms thereof.

* * * * *

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed by their duly authorized officers as of the date first set forth above.

PROV	IDER
CORD	ILLERA COMMUNICATIONS,
LLC	
By:	
Name:	
Title:	
RECIP	PIENT
SCRIP	PS MEDIA, INC.
By:	
Name:	
Title:	